Needs of people living with dementia and their informal caregivers for activating interventions

Netta van 't Leven

Needs of people living with dementia and their informal caregivers for activating interventions

Netta van 't Leven

The studies presented in this thesis were conducted at the Research Center Innovations in Care of Rotterdam University of Applied Sciences, in cooperation with the Department of Clinical, Neuro- & Developmental Psychology at the Vrije Universiteit Amsterdam and the Amsterdam Public Health research institute, the Netherlands.



Rotterdam University of Applied Sciences awarded a PhD-grant (promotievoucher) to the author, and financially supported the publication of the articles from chapter 4 and 6, and the printing of this thesis.



Alzheimer Nederland financially supported the printing of this thesis.

COLOFON

ISBN: 9789402820096

Lay out and cover design: Mariël Lam BNO, Den Bosch Printed by: Ipskamp Printing, Enschede

© 2020 Netta van 't Leven

All rights reserved. Chapters 2, 3 and 5 may only be reproduced with permission of the author, or the copyright-owning journals.

VRIJE UNIVERSITEIT

Needs of people living with dementia and their informal caregivers for activating interventions

ACADEMISCH PROEFSCHRIFT

ter verkrijging van de graad Doctor aan de Vrije Universiteit Amsterdam, op gezag van de rector magnificus prof.dr. V. Subramaniam, in het openbaar te verdedigen ten overstaan van de promotiecommissie van de Factulteit der Gedrags- en Bewegingswetenschappen op woensdag 3 juni 2020 om 13.45 uur in de aula van de universiteit, De Boelelaan 1105

> door Maria Antoinette van 't Leven geboren te Den Haag

promotor: prof.dr. A.M. Pot

copromotoren: dr. J. de Lange dr. E.S. van der Ploeg

Contents

CHAPTER 1	General introduction	7
PARTI	The impact of dyadic, activating interventions for people with dementia and their informal caregivers	
CHAPTER 2	Dyadic psychosocial interventions for community-dwelling people with dementia and their family caregivers: a systematic review.	19
CHAPTER 3	How do activating interventions fit the personal needs, characteristics and preferences of people with dementia living in the community and their informal caregivers?	57
CHAPTER 4	Working mechanisms of dyadic, psychosocial, activating interventions for people with dementia and informal caregivers: a qualitative study.	81
PART II	The appropriateness of criteria for referral to dyadic, activating interventions	
CHAPTER 5	Indicators to estimate the appropriateness of activating interventions for people living with dementia and for their informal caregivers.	105
CHAPTER 6	Exploring the usefulness of indicators for referring people with dementia and their informal caregivers to activating interventions: a qualitative analysis of needs assessments.	125
CHAPTER 7	General discussion	143
SUMMARY		159
SAMENVATTI	NG (SUMMARY IN DUTCH)	167

DANKWOORD (ACKNOWLEDGEMENTS IN DUTCH)	175
OVER DE AUTEUR (ABOUT THE AUTHOR IN DUTCH)	179
PUBLICATION LIST	183

CHAPTER 1 -

General introduction

Background

Challenges to maintain activities after a dementia diagnosis

Worldwide, the number of people with dementia is expected to increase exponentially in the coming decades due to an increased life expectancy. In the Netherlands, 270,000 people currently live with dementia, with more than 620,000 people projected for 2050.¹ An even larger increase is expected worldwide, from 46 million in 2015 to 131.5 million in 2050.² Dementia is characterized by a progressive decline of cognitive functions, profoundly limiting the activities of daily living and social functioning.³ As there is no cure for dementia, psychosocial interventions are important to maintain quality of life by coping with the consequences of dementia.^{4,5}

Community-based care and support for people with dementia will continue to be needed in the coming decades.⁶ About 74% of Dutch people with dementia live in their own homes and receive support and care from informal caregivers.⁷ These informal caregivers are most often spouses and adult children, though they may occasionally be neighbors or close friends. About 60% of informal caregivers live with the person with dementia.¹ With the increase of people with dementia and the current trend for them to live at home longer, informal caregivers are projected to play a more substantial future role in dementia care. Being a caregiver, which includes making decisions for the person with dementia, changes the nature of the relationship with the partner, child, friend, or neighbor. Caring is time consuming and the ongoing demand and unpredictability of dementia's progression is burdensome.^{8,9} Interventions which support both people with dementia and the caregiver are seen as most effective in maintaining the guality of life for both.¹⁰⁻¹² Awareness of dementia's effect on the relationship between people with dementia and their informal caregivers has increased in recent years, especially when it comes to communication, shared activities, reciprocity, and happiness.¹³⁻¹⁵ The behaviors of the informal caregiver and his or her sense of competence can greatly influence behavioral changes in people with dementia.^{14,16} Dyadic interventions enable them to discuss relationship issues, such as dealing with emotions and practicing skills for coping with declines in capacities.^{15,17}

Continuing to participate in meaningful activities is important for people with dementia and their caregivers, but not easy. These activities can have a positive effect on the well-being and health of people with dementia, building self-esteem and satisfaction, while also providing life continuity.¹⁸⁻²⁰ Although some people with dementia develop strategies to remain active and continue activities they

previously enjoyed, many frequently report a lack of company and daytime activities, both at home and elsewhere.²¹⁻²⁵ Informal caregivers report difficulties in helping the person with dementia maintain their activities, while also keeping up with their own.^{7,9,26} Informal caregivers have repeatedly expressed a need for professional advice regarding continuing activities.^{23,27}

Person-centered dementia care

The focus of this research, the need for meaningful activities, is one of the domains of person-centered care, along with comfort, attachment, inclusion and identity.⁴ Kitwood includes these five domains of psychological needs to fulfil the central need to be a loved and respected person when living with dementia. They are all essential for maintaining a sense of personhood or 'continuation of self'.^{4,28,29} In a person-centered approach, needs are interpreted from the viewpoint of the person with dementia, the individual course of the dementia process and their individual life history.^{4,29,30} This also includes the person's experiences with the interdependency and reciprocity that are inherent in caring relationships.³¹ Therefore, needs within the domains may vary per individual.

Person-centered care is directed at maximizing the individual's potential by optimally using their individual capacities and creating a physical and social environment that supports the person with dementia in living with meaning and dignity.^{4,30} People with dementia make their own choices where possible, and are not seen as passive recipients of standard care services.^{4,29} Self-evidently, an informal caregiver also has an individual biography, lifestyle, and coping strategy.^{13,14,32} In a person-centered approach, the needs, characteristics and preferences of both the people with dementia and the caregivers are equally important in offering appropriate care and support.

Three dyadic, activating interventions

The primary investigator and co-authors have been involved in three promising activating interventions based on the available evidence at the start of this project, which focus on the activity needs of people with dementia and their informal caregivers, in line with a person-centered approach. These interventions emphasized potential adaptations of activities to fit the remaining capacities of the people with dementia. The interventions include psycho-education, emotional support, the practice of various activities, and communication skills for caregivers. The interventions consist of six to ten home visits. These interventions are: the Pleasant Events Program; the Exercise and Support Intervention; and Community Occupational Therapy in Dementia (COTiD) (Chapter 3, Table 1, p. 60). The Pleasant Events Program focuses on finding enjoyable activities for the dyad, planning them, and when necessary, simplifying these activities for the person with dementia. Examples are: walking around the neighborhood; taking public transport to a historic part of the city; watching a specific television program.³³⁻³⁵ The Exercise and Support Intervention aims to increase the physical health and the mood of both people with dementia and caregivers while decreasing the caregiver's burden. The intervention includes joint exercises and planning pleasant activities. The exercises focus on flexibility, balance, strength and/or endurance.^{36,37} Community Occupational Therapy in Dementia (COTiD) aims to generally improve the performance of daily activities. Activities that are relevant to the dyad are chosen, then practiced, often in combination with environmental adaptations. Examples are: using a remote control with orientation marks for the television, completing a bicycle ride on a safe route, and maintaining daily structure through an individualized agenda.^{38,39} These interventions showed positive effects^a on the mood, daily activities, general health and guality of life of people with dementia, as well as on the mood, competence, and quality of life of their caregivers.^{33,34,37,44,45}

Current usual care to maintain activities

People with dementia and their informal caregivers often consult their general practitioner or a geriatrician for memory complaints when they suspect dementia. In the Netherlands, people with dementia are offered a case manager, although, only 35% of recently diagnosed people with dementia actually have a specialized dementia case manager.^{7,46} The general practitioner, geriatrician, and case manager are the leading professionals for coordinating care and support for people with dementia and their informal caregivers after diagnosis. Despite reported difficulties in maintaining daily activities, not many people with dementia are referred to dyadic, activating interventions.⁴⁷⁻⁵⁰ For a person-centered approach, care services should meet personal activity needs. Currently, criteria for referrals to interventions are lacking. Physicians report a need for more refined criteria that help to refer people with dementia to interventions.^{51,52} In effect studies of psychosocial interventions, there is often a call to find a better intervention to fit the specific needs of the individual with dementia and the informal caregiver who supports them.^{53,54}

a Later studies did not always duplicate the evidence found in these studies ⁴⁰⁻⁴³.

Aim and research questions of this thesis

The aim of this thesis was to study (1) the impact of dyadic, activating interventions for people with dementia and their caregivers, and (2) the appropriateness of criteria for referring dyads to these interventions.

Research questions in Part I

Question 1

What evidence supports psychosocial, dyadic interventions for communitydwelling people with dementia and their caregivers, and what is the relationship with treatment components?

Question 2

How do the three promising dyadic, activating interventions match the needs, characteristics and preferences of the participating dyads?

Question 3

What do people with dementia, their caregivers and coaches perceive as working mechanisms in the three dyadic, activating interventions?

Research questions in Part II

Question 4

Which criteria describing activity needs, characteristics, and preferences of people with dementia and their informal caregivers are recognizable for physicians and case managers in their daily practice?

Question 5

Are these criteria helpful for assessing activity needs, characteristics, and preferences of community-dwelling people with dementia and their caregivers?

Outline and methodology

Part I The impact of dyadic, activating interventions for people with dementia and their informal caregivers

Chapter 2 describes dyadic psychosocial interventions in a systematic literature review. We followed the guidelines from the Cochrane Handbook for Systematic Reviews of Interventions. We applied a quantitative analysis with a qualitative

approach to search for similarities and differences in the treatments of the different interventions. Finally, we compared the characteristics and treatments with significant effects of the interventions on the outcome domains.

Chapter 3 explains the matching of the three dyadic, activating interventions with the needs, characteristics and preferences of the participants and their caregivers. In a qualitative study, we investigated the experiences of the dyads and their coaches in these three dyadic, activating interventions. We used semi-structured interviews to gather in-depth information.

Chapter 4 describes the working mechanisms of the interventions, as perceived by people with dementia, their caregivers, and coaches. We used the same qualitative interviews from Chapter 3.

Part II The appropriateness of criteria for referral to dyadic, activating interventions

The findings of Chapter 3 provided input for the studies of Part II of our research. From the qualitative descriptions in the interviews, the research team operationalized the factors as observable criteria. Each criterion describes a need, characteristic or preference of people with dementia and/or their caregivers. After thoroughly discussing clarity and consistency, we drafted 31 conceptual criteria.

Chapter 5 answers the question which criteria physicians and case managers recognize in their daily practice with people with dementia and their caregivers. Using the 'RAND Appropriateness Method', an expert panel, consisting of geriatricians, general practitioners and dementia case managers, rated the recognizability of these criteria.

Chapter 6 explores the usefulness of the criteria for real-life assessment of activity needs, characteristics and preferences. We performed a secondary analysis of needs assessments-interviews conducted by a case manager with community-dwelling people with dementia and their caregivers. We interpreted text fragments in the context of the whole interview.

Chapter 7 is a general discussion of the thesis and summarizes the main findings, reflects on the findings and the methodology, and formulates implications for clinical practice and future research.

References

- 1. Dutch Alzheimer's Society. Factsheet Feiten-cijfers [factsheet facts-figures], 2019. https:// www.alzheimer-nederland.nl/dementie/feiten-cijfers
- 2. Prince M, Wimo A, Guerchet M, et al. The Global Impact of Dementia, An analysis of prevalence, incidence, cost and trends. London: Alzheimer's Disease International, 2015.
- 3. Livingston G, Sommerlad A, Orgeta V, et al. Dementia prevention, intervention, and care. *Lancet*. 2017;390(10113):2673-2734.
- 4. Kitwood TM. Dementia Reconsidered: The Person Comes First Buckingham UK: Open University Press, 1997.
- Moniz-Cook E, Vernooij-Dassen M, Woods B, Orrell M. Psychosocial interventions in dementia care research: the INTERDEM manifesto. Aging & Mental Health. 2011;15(3):283-290.
- 6. WHO. Dementia: A Public Health Priority Geneva, Switzerland: WHO, 2012.
- 7. Francke A, Heide Ivd, Bruin Sd, et al. Een samenhangend beeld van dementie en dementiezorg [An integrated picture of dementia and dementia care]. Utrecht: NIVEL, 2018.
- Borsje P, Hems MA, Lucassen PL, Bor H, Koopmans RT, PotAM. Psychological distress in informal caregivers of patients with dementia in primary care: course and determinants. *Family Practice*. 2016;33(4):374-381.
- 9. Van Bruggen S, Gussekloo J, Bode C, Touwen DP, Engberts DP, Blom JW. Problems experienced by informal caregivers with older care recipients with and without cognitive impairment. *Home Health Care Services Quartily.* 2016;35(1):11-24.
- 10. Olazaran J, Reisberg B, Clare L, et al. Nonpharmacological therapies in Alzheimer's disease: a systematic review of efficacy. *Dementia Geriatric Cognitive Disorders*. 2010;30(2):161-178.
- Smits CHM, de Lange J, Dröes R-M, Meiland F, Vernooij-Dassen M, Pot AM. Effects of combined intervention programmes for people with dementia living at home and their caregivers: A systematic review. *International Journal of Geriatric Psychiatry*. 2007;22(12):1181-1193.
- 12. Thinnes A, Padilla R. Effect of educational and supportive strategies on the ability of caregivers of people with dementia to maintain participation in that role. *American Journal of Occupational Therapy*. 2011;65(5):541-549.
- 13. Ablitt A, Jones GV, Muers J. Living with dementia: a systematic review of the influence of relationship factors. *Aging & Mental Health*. 2009;13(4):497-511.
- 14. de Vugt ME, Stevens F, Aalten P, et al. Do caregiver management strategies influence patient behaviour in dementia? *International Journal of Geriatric Psychiatry* 2004;19:85-92.
- Rausch A, Caljouw MA, van der Ploeg ES. Keeping the person with dementia and the informal caregiver together: a systematic review of psychosocial interventions. *International Psychogeriatrics*. 2017;29(4):583-593.

- 16. Corcoran MA. Caregiving styles: a cognitive and behavioral typology associated with dementia family caregiving. *Gerontologist*. 2011;51(4):463-472.
- 17. Prick AE. Evaluation of a home-based physical exercise and support intervention for people living with dementia and their informal caregivers: a randomised controlled trial (PhD thesis). Amsterdam: Department of Clinical, Neuro- and Developmental psychology, EMGO+ institute, Faculty of Behavioural and Movement Sciences, Vrije Universiteit, 2016.
- Öhman A, Nygard, L. Meanings and Motives for Engagement in Self-Chosen Daily Life Occupations Among Individuals With Alzheimer's disease. OTJR: Occupation, Participation and Health. 2005;25(3):89-97.
- 19. Stav WB, Hallenen T, Lane J, Arbesman M. Systematic review of occupational engagement and health outcomes among community-dwelling older adults. *American Journal of Occupational Therapy*. 2012;66(3):301-310.
- 20. Vikstrom S, Josephsson S, Stigsdotter-Neely A, Nygard L. Engagement in activities: Experiences of persons with dementia and their caregiving spouses. *Dementia: The International Journal of Social Research and Practice*. 2008;7(2):251-270.
- 21. Clare L, Goater T, Woods B. Illness representations in early-stage dementia: a preliminary investigation. *International Journal of Geriatric Psychiatry*. 2006;21(8):761-767.
- Black BS, Johnston D, Rabins PV, Morrison A, Lyketsos C, Samus QM. Unmet needs of community-residing persons with dementia and their informal caregivers: findings from the maximizing independence at home study. *Journal of the American Geriatrics Society*. 2013;61(12):2087-2095.
- Johnston D, Samus QM, Morrison A, et al. Identification of community-residing individuals with dementia and their unmet needs for care. *International Journal of Geriatric Psychiatry*. 2011;26(3):292-298.
- 24. Miranda-Castillo C, Woods B, Orrell M. The needs of people with dementia living at home from user, caregiver and professional perspectives: a cross-sectional survey. *BMC Health Services Research Journal*. 2013;13:43.
- 25. van der Roest HG, Meiland FJM, Maroccini R, Comijs HC, Jonker C, Dröes RM. Subjective needs of people with dementia: a review of the literature. *International Psychogeriatrics*. 2007;19:559-592.
- 26. Vikström S, Borell L, Stigsdotter-Neely A, Josephsson S. Caregivers' Self-Initiated Support Toward Their Partners With Dementia When Performing an Everyday Occupation Together at Home. OTJR: Occupation, Participation and Health. 2005;25(34):149-159.
- 27. Zwaanswijk M, Peeters JM, van Beek AP, Meerveld JH, Francke AL. Informal caregivers of people with dementia: problems, needs and support in the initial stage and in subsequent stages of dementia: a questionnaire survey. *Open Nursing Journal*. 2013;7:6-13.
- 28. Brooker D, Latham I. Person-Centred Dementia Care, Making Services Better with the VIPS Framework. Second Edition: ed. London: Jessica Kingsley Publishers, 2016.

- 29. Dewing J. Personhood and dementia: revisiting Tom Kitwood's ideas. International Journal of Older People Nursing. 2008;3(1):3-13.
- 30. de Lange J. Omgaan met dementie [Dealing with dementia](PhD thesis). Utrecht: Trimbos-instituut [Netherlands Institute of Mental Health and Addiction], 2004.
- 31. Nolan MR, Ryan T, Enderby P, Reid D. Towards a more inclusive vision of dementia care practice and research. *Dementia: The International Journal of Social Research and Practice*. 2002;1:193-211.
- 32. Peeters JM, Van BeekAPA, Meerveld JHCM, Spreeuwenberg PMM, Francke AL. Informal caregivers of persons with dementia, their use of and needs for specific professional support: a survey of the National Dementia Programme. *BMC Nursing.* 2010;9(9).
- Gitlin LN, Winter L, Burke J, Chemett N, Dennis MP, Hauck WW. Tailored activities to manage neuropsychiatric behaviors in persons with dementia and reduce caregiver burden: A randomized pilot study. American Journal of Geriatric Psychiatry. 2008;16(3):229-239.
- 34. Teri L, Logsdon RG, Uomoto J, McCurry SM. Behavioral treatment of depression in dementia patients: a controlled clinical trial. *Journal Gerontology - Series B: Psychological Sciences and Social Sciences*. 1997;52(4):P159-166.
- 35. Verkaik R, Francke A, Lahr M, Zuidervaart M. Huisbezoeken; Voorkom Somberheid bij Geheugenproblemen, handboek voor de hulpverlener (plezierige activiteitenmethode). [Home visits: Preventing depressive mood in memory problems, manual for the professional: Pleasant activities]. retrieved from www.nivel.nl/publicaties. Utrecht: Nivel, 2005.
- 36. Prick AE, de Lange J, Scherder E, Pot AM. Home-based exercise and support programme for people with dementia and their caregivers: study protocol of a randomised controlled trial. BMC Public Health. 2011;11:894.
- 37. Teri L, Gibbons LE, McCurry SM, et al. Exercise plus behavioral management in patients with Alzheimer disease: a randomized controlled trial. *JAMA*. 2003;290(15):2015-2022.
- 38. Graff M, Melick Mv, Thijssen M, Verstraten P, Zajec J. Ergotherapie bij ouderen met dementie en hun mantelzorgers: Het EDOMAH-programma [Occupational Therapy for Older Persons with Dementia and their informal Caregivers: COTiD-programme]. Houten: Bohn Stafleu van Loghum, 2010.
- 39. Graff MJ, Vernooij-Dassen MJ, Zajec J, Olde Rikkert MG, Hoefnagels WH, Dekker J. How can occupational therapy improve the daily performance and communication of an older patient with dementia and his primary caregiver? A case study. Dementia: The International Journal of Social Research and Practice. 2006;5(4):503-532.
- 40. Döpp CM, Graff MJ, Teerenstra S, Olde Rikkert MG, Nijhuis-van der Sanden MW, Vernooij-Dassen MJ. Effectiveness of a training package for implementing a community-based occupational therapy program in dementia: a cluster randomized controlled trial. *Clinical Rehabilitation*. 2015;29(10):974-986.

- 41. Prick AE, de Lange J, Scherder E, Twisk J, Pot AM. The effects of a multicomponent dyadic intervention on the mood, behavior, and physical health of people with dementia: a randomized controlled trial. *Clinical Interventions in Aging*. 2016;11:383-395.
- 42. Prick AE, de Lange J, Twisk J, Pot AM. The effects of a multi-component dyadic intervention on the psychological distress of family caregivers providing care to people with dementia: a randomized controlled trial. *International Psychogeriatrics*. 2015;27(12):2031-2044.
- 43. Voigt-Radloff S, Graff M, Leonhart R, et al. A multicentre RCT on community occupational therapy in Alzheimer's disease: 10 sessions are not better than one consultation. *British Medical Journal open*. 2011.
- Graff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM.
 Community based occupational therapy for patients with dementia and their care givers:
 Randomised controlled trial. British Medical Journal. 2006;333(7580):1196-1199.
- 45. Graff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM.
 Effects of community occupational therapy on quality of life, mood, and health status in
 dementia patients and their caregivers: A randomized controlled trial. *Journals of Gerontology* Series A Biological Sciences and Medical Sciences. 2007;62(9):1002-1009.
- 46. Pol Hvd, Luijk R, Leegwater E. Zorggebruik van mensen met dementie in beeld [An overview of use of careservices for people with dementia]. Zeist: Vektis, 2019.
- 47. Aminzadeh F, Molnar FJ, Dalziel WB, Ayotte D. A review of barriers and enablers to diagnosis and management of persons with dementia in primary care. *Canadian Geriatrics Journal*: 2012;15(3):85-94.
- 48. Boots LM, Wolfs CA, Verhey FR, Kempen GI, de Vugt ME. Qualitative study on needs and wishes of early-stage dementia caregivers: the paradox between needing and accepting help. International Psychogeriatrics. 2015;27(6):927-936.
- 49. Martin F, Turner A, Wallace LM, Choudhry K, Bradbury N. Perceived barriers to self-management for people with dementia in the early stages. *Dementia: The International Journal of Social Research and Practice. (London).* 2013;12(4):481-493.
- O'Sullivan G, Hocking C. Translating action research into practice; seeking occupational justice for people with dementia. OTJR: Occupation, Participation and Health. 2013;33(3):168-176.
- 51. Döpp CM, Graff MJ, Teerenstra S, et al. A new combined strategy to implement a community occupational therapy intervention: designing a cluster randomized controlled trial. *BMC Geriatrics*. 2011;11:13.
- 52. Van't Leven N, Graff MJ, Kaijen M, de Swart BJ, Olde Rikkert MG, Vernooij-Dassen MJ. Barriers to and facilitators for the use of an evidence-based occupational therapy guideline for older people with dementia and their carers. *International Journal of Geriatric Psychiatry*. 2012;27(7):742-748.

- 53. Brodaty H, Arasaratnam C. Meta-analysis of nonpharmacological interventions for neuropsychiatric symptoms of dementia. *American Journal of Psychiatry*. 2012;169(9):946-953.
- 54. Coon DW, Thompson L, Steffen A, Sorocco K, Gallagher-Thompson D. Anger and depression management: psychoeducational skill training interventions for women caregivers of a relative with dementia. *Gerontologist*. 2003;43(5):678-689.

PART I

The impact of dyadic, activating interventions for people with dementia and their informal caregivers

CHAPTER 2 -

Dyadic interventions for community-dwelling people with dementia and their family caregivers: a systematic review

Published as:

Van't Leven N, Prick AE, Groenewoud JH, Roelofs P, de Lange J, Pot AM. Dyadic interventions for community-dwelling people with dementia and their family caregivers: a systematic review. *International Psychogeriatrics*, 2013; 25(10),1581-603. doi: 10.1017/S1041610213000860. Epub 2013 Jul 24.

Background

In this review, we study the effects of dyadic psychosocial interventions focused on community-dwelling people with dementia and their family caregivers, and the relationship of the effects with intervention components with programs.

Methods

A search from January 2005 to January 2012 led to 613 hits, which we reviewed against our inclusion criteria. We added studies from 1992 to 2005 reviewed by Smits et al.¹² We assessed the methodological quality of 41 programs with the Cochrane criteria and two items of the Oxford Centre of Evidence-based Medicine guidelines.

Results

Studies of moderate to high quality concerning 20 different dyadic psychosocial programs for people with dementia and caregivers were included. Nineteen of these programs show significant effects on the patient with dementia, the caregiver, or both. Due to differences in the programs and the studies, this study does not provide an unequivocal answer about which programs are most effective. Programs with intervention components that actively train one or more specific functional domains for the person with dementia and/or the caregiver seem to have a beneficial impact on that domain, although there are exceptions. Reasons can be found in the program itself, the implementation of the programme, and the study design.

Conclusion

Dyadic psychosocial programs are effective, but the outcomes for the person with dementia and the caregiver vary. More attention is needed for matching the targeted functional domains, intervention components, and delivery characteristics of a program with the needs of the person with dementia and the family caregiver.

Introduction

Most people with dementia live in their own homes in the community. They need support and care in everyday life, and they are dependent on informal care, mainly provided by spouses and adult children, but also by neighbors or friends. Although caregiving is satisfying for most informal caregivers because they care about their loved ones, it is also very burdensome.¹⁻³ People with dementia and their caregivers have to cope with impaired daily functioning and changing roles, often with a negative impact on their health condition.³⁻⁵ Many psychosocial supporting interventions for people with dementia and their caregivers have been developed in the last decades.^{6,7} Evidence for the effectiveness of these interventions, being even more effective than pharmacological therapies, has been published.⁸⁻¹⁰ In recent years, psychosocial interventions have focused on both the person with dementia and the informal caregiver (also referred to as the "dyad" in this paper). Directing the dyad is seen as most effective because of the mutual influence between the person with dementia and the informal caregiver. For instance, behavioral symptoms of dementia may increase the caregiver burden; caregiver management strategies will influence both the behavior of the person with dementia and the feelings of competence and mood of the caregiver.¹¹ The effects of psychosocial intervention programs have been studied in a previous review that included publications up to 2005.¹² The authors found that psychosocial intervention programs may contribute to the quality of life of both members of the dyad, and may decrease caregivers' mental health problems. The effects on most other functional and behavioral domains, however, are moderate or inconsistent. Some interventions led to statistically significant effects in subgroups only. Currently, a wide range of psychosocial programs are offered to people with dementia and their caregivers. Some of these have been evaluated in randomized controlled trials (RCTs). The purpose of our current study was to update Smits et al.'s systematic review, and to provide the current best evidence about psychosocial programs for the dyads that involve face-to-face contact between professional caregivers and both the patient and the caregiver. We describe the program characteristics and the measured effects on both members of the dyads. These outcomes are related to the intervention components of the programs.

Method

Search strategy

We searched the databases Psychinfo, Embase, Medline, and Cinahl for single studies and reviews, and we searched the Cochrane Library for systematic reviews. Since we built on the review of Smits et al. (2007), our search covered publications from January 2005 to January 2012. We used the same search string with the following keywords: (Alzheimer* OR dementia) AND (caregiv* OR family members) AND (support program OR training OR counselling OR intervention) AND (effec* OR effic^{*}), as well as Mesh or Emtree terms to ensure that the search was as complete as possible.¹³⁻¹⁵ Any systematic reviews we found were searched for mention of additional single RCTs involving psychosocial interventions (Figure 1).

Inclusion criteria

We included effect studies evaluating dyadic psychosocial interventions for both older people with dementia living in the community and their caregivers. A broad definition of psychosocial interventions was used. Interventions that encompass other treatment components than psychosocial ones - such as environmental modifications and exercise – were also included. The interventions had to involve face-to-face contact between a care professional and the person with dementia as well as the informal caregiver and the same care professional. In addition, the interventions had to target psychosocial outcomes, improving mental health or well-being. In contrast with Smits et al., we included only RCTs in our current review. We excluded RCTs involving respite interventions, and technological devices, as well as cost-effectiveness studies, studies among nursing home residents and integrated studies where results could not be related to a specific intervention or program (Table 1).

Table 1 Inclusion and exclusion criteria

	Inclusion criteria
Study participants	People with dementia 65 years old or more.
	People with dementia and their informal caregivers living in the community, not a nursing home.
Study design	Effect study: randomized controlled trial.
Psychosocial intervention	Intervention aimed at reducing or preventing the mental health decline of one or both members of the dyad, including the areas of cognition, activities, daily living skills, competence, and interpersonal relationships. Face-to-face contact between care professional and person with demen- tia, and between the same care professional and caregiver.
Language	English, Dutch, German, and French.
	Exclusion criteria
Study aim	Cost-effectiveness.
Pooled data	Combination of intervention studies.

Selection of studies

First, one reviewer (NL) screened the titles against the inclusion criteria and discarded obviously irrelevant publications. Second, two pairs of reviewers (NL/ AEP and NL/JG) independently assessed the abstracts of the remaining publications and the additional studies found in the reviews. Any discrepancies were resolved by consensus of all three reviewers. Finally, NL/AEP and NL/JG examined potentially relevant articles in full text.



Figure 1 Flow chart of identification of studies

Quality assessment

The two pairs of reviewers (NL/AEP and NL/JG) independently assessed all publications (that is, those resulting from the current search and any additional ones included in Smits et al.'s review) for methodological quality by using the Cochrane rating criteria for RCTs.¹⁴ The items "blinding of participants" and "blinding of therapists" were not scored because blinding is not feasible for the type of intervention studied. We added the following two items from the Oxford Centre of Evidence-based Medicine guidelines to the Cochrane criteria (http://www.cebm. net): the specific components of the intervention should be described, and the experimental and control groups must each have a minimum of 30 participants (Table 2).¹⁶ If information was missing, we contacted the corresponding authors of the publication for such information.

Data analysis

We used several strategies for data analysis to do justice to the variety of programs and studies. First, we described the intervention programs by delivery characteristics (e.g., dose, mode of delivery, group vs. individual, adaptability/control), intervention components, and targeted functional domains.¹⁷ The intensity of contact in the program was rated on a scale ranging from 1 to 4, with 1 representing "1-2 sessions" and 4 representing "more than 10 sessions".¹⁰ Second, for all outcomes of interest, we assessed the strength of the body of evidence using the Grades of Recommendation, Assessment, Development and Evaluation Working Group (GRADE) approach, as recommended in the Cochrane Handbook 5.1.¹⁴ The strongest evidence comes from one or more good-guality RCTs. Limitations in the design suggesting bias may warrant downgrading the guality of the evidence of the RCT to moderate or even lower. We assessed the quality of the body of evidence as "low", "moderate" or "high" for each outcome category. If the data warranted it, we quantitatively compared studies for the same targeted psychosocial outcome with the Review Manager (software version 5.1).¹⁴ The standardized mean difference was used to compare effect sizes if the studies used different instruments to measure the outcome of interest. A random effects model analysis was applied for the statistical heterogeneity of the studies. Data obtained after intervention (or at 12 months for the programs that lasted one year or more) were used for this analysis. Pooled estimates were not calculated because of the clinical and statitical heterogeneity between the studies.

Table 2 Quality assessment of studies meeting the inclusion criteria

Criteria	1	2	3	4	5	6	7	I	П		
Columns 1-7: Cochrane criteria and columns I, II: Oxford Centre of Evidence-based Medicine Guideline for assessing methodological quality 2012-2005	Was allocation randomized?	Was randomization procedure adequate and transparent?	Was data collection blinded (independent assessor)?	Were baseline characteristics of study groups comparable?	Were follow-up data available and were drop-outs described? ⁱ	Were data for all respondents analyzed in the group of randomization? Was intention-to-treat data analyzed?	Were experimental and control groups treated the same way except for the intervention?	Was a detailed description of the intervention given?	Were the effects on at least 30 patients assessed?	Inclusion	Program
Jansen et al., 2011 ³¹	+	+	+	+	+	+	+	+	+	+	1
Clare et al., 2011 ³²	+	+	+	+	+	?	+	+	-	-	
Chien & Lee, 2011 ³³	+	?	+	+	+	+	+	+	+	+	2
Carbonneau et al., 2011 ³⁴	+	?	?	+	+	+	+	+	-	-	
Bakker et al., 2011 ³⁵	+	+	-	+	+	+	+	+	+	+	3
McCurry et al., 2011 ^{36*}	+	+	+	+	+	+	+	+	+	+	4
McCurry et al., 2010 ³⁷	+	?	?	?	?	+	+	-	+	-	
McCurry et al., 2005 ³⁸	+	+	+	-	-	+	+	+	-	-	
Logsdon et al., 2010 ^{39*}	+	?	?	+	+	+	+	+	+	+	5
Logsdon et al., 2006 ⁴⁰	+	?	?	+	+	+	+	+	-	-	
Gitlin et al., 2010A ²² #	+	+	+	+	+	+	+	+	+	+	6
Gitlin et al., 2010B ²³	+	+	+	+	+	+	+	+	+	+	6
Neely et al., 2009 ⁴¹	+	-	-	+	+	+	+	+	-	-	
Eloniemi-Sulkava et al., 2009 ⁴²	+	+	-	?	+	+	+	+	+	+	7
Gitlin et al., 2008 ²¹	+	+	+	?	+	+	+	+	+	+	8
Dias et al., 2008 ⁴³	+	+	+	+	+	?	+	+	+	+	9
Onor et al., 200744	+	?	?	?	+	+	+	+	-	-	
Callahan et al., 200645	+	+	+	-	+	+	+	+	+	+	10
Dröes et al., 2006 ⁴⁶	-	-	-	+	-	+	+	+	-	-	
Voigt-Radloff et al., 2011 ²⁰ #	+	+	+	+	+	+	+	+	+	+	11
Graff et al., 2007 ^{29*}	+	+	+	+	+	+	+	+	+	+	11
Graff et al., 2006 ¹⁹	+	+	+	+	+	+	+	+	+	+	11
Onder et al., 200547	+	+	+	?	+	+	+	+	+	+	12
Martin-Cook et al., 200548	+	?	?	+	-	+	+	+	-	-	
Hepburn et al., 2005 ⁴⁹	+	+	-	-	+	+	+	+	+	+	13

>> Table 2 Quality assessment of studies meeting the inclusion criteria

Criteria	1	2	3	4	5	6	7	I	П		
Columns 1-7: Cochrane criteria and Columns I, II: Oxford Centre of Evidence-based Medicine Guideline for assessing methodological quality 2005-1992	Was allocation randomized?	Was randomization procedure adequate and transparent?	Was data collection blinded (independent assessor)?	Were baseline characteristics of study groups comparable?	Were follow-up data available and were drop-outs described? ⁱ	Were data for all respondents analyzed in the group of randomization? Was intention-to-treat data analyzed?	Were experimental and control groups treated the same way except for the intervention?	Was a detailed description of the intervention given?	Were the effects on at least 30 patients assessed?	Inclusion	Program
Berger et al., 2004 ⁵⁰	-	+	?	+	+	+	+	+	-	-	
Dröes et al., 2004a ^{51*} Dröes et al., 2004b ⁵²	-	+	+	-	+	+	+	+	-	-	
Dröes et al., 2000 ⁵³	-	+	+	-	+	+	+	+	-	-	
Gitlin et al., 2003 ⁵⁴ #	+	+	?	+	+	+	+	+	+	+	
Gitlin et al., 2001 ⁵⁵	+	+	?	+	+	+	+	+	+	+	14
Teri et al., 2003 ⁵⁶	+	+	+	+	+	+	+	+	+	+	15
Romero&Wenz, 2002 ⁵⁷	-	-		-		-	-		-	-	
Eloniemi-Sulkava et al., 2001 ⁵⁸	+	+	+	+	+	?	+	+	+	+	16
Quayhagen&Quayhagen, 2001 ⁵⁹	+	?	?	+	-	+	+	+	-	-	
Chu et al., 2000 ⁶⁰	+	?	?	+	+	+	+	+	+	+	17
Aupperle&Coyne, 200061	-	-	?	+	?	+	+	+	-	-	
Ostwald et al., 1999 ⁶²	+	?	?	+	?	+	+	+	+	+	18
Logiuduce et al, 1999 ⁶³	+	+	-	+	+	+	+	+	-	-	
Miller et al., 1999 ^{64‡} Newcomer et al., 1999 ⁶⁵ Yordi et al., 1997 ⁶⁶	+	+	+	+	+	+	+	+	-	+	19
Moniz-Cook et al., 1998 ⁶⁷	-	+	?	+	+	+	+	+	-	-	
Riordan & Bennett, 199868	-	+	-	+	-	+	+	+	-	-	
Teri et al., 1997 ⁶⁹	+	-	?	+	+	+	+	+	-	-	
Brodaty et al., 1997 ^{70*} Brodaty&Gresham, 1989 ⁷¹	+	?	?	?	+	+	+	+	+	+	20
Hinchcliffe et al., 1995 ⁷²	+	-	+	+	+	+	+	+	-	-	
Vernooij-Dassen et al., 1995 ^{73*}	+	?	?	?	?	?	?	-	-	-	
Vernooij-Dassen, 1993 ⁷⁴	+	+	+	+	+	+	+	-	+	-	

Notes: Criterium 5; Were follow-up data for a sufficient proportion of all included patients available and were dropouts described? (loss of 20% for short-term follow-up and 30% for long-term follow-up (>6 months)). +Low risk, - high risk, ~not applicable, ? no information given *Programs are published in more than one publication #Programs are studied again in a new study design and population.

Results

Literature search and quality assessment

For the period 2005–2012, the search strategy led to 608 single studies and five additional studies in the reviews. After the inclusion and exclusion criteria were applied, 25 publications remained. Smits et al. included 25 publications from the period 1992 to 2005.¹² Therefore in total 50 publications were judged on methodological quality (Figure 1).

These 50 publications concerned 41 intervention programs. Table 2 shows the outcomes for the methodological quality criteria per study and the final judgment for inclusion. Finally, 20 dyadic psychosocial programs studied in 23 RCTs were included in this review. Thus, three RCTs were replication studies of intervention programs that were already studied in an earlier RCT.

Program characteristics

Table 3 shows the characteristics of each program (the numbers in square brackets in the text below correspond with the program numbers in Table 3). On the basis of the delivery characteristics, programs can be classified in following three categories:

- Short-period, intensive programs, consisting of six to ten home visits [six programs: 2, 4a, 4b, 5, 6a, 6b, 8, 9] or group sessions [four programs: 1, 3, 7, 10] during a period of five weeks to six months with scheduled topics. All these programs explicitly target both members of the dyad.
- 2. Long-lasting programs, that is, case management up to 2 years, with home visits and telephone contact [six programs: 11, 13, 14, 15, 16, 17] or in combination with a group session [one program: 12]. The intervention components of these programs primarily target the caregiver, and to a lesser extent the person with dementia.
- 3. <u>Other programs</u> with temporary hospitalization [three programs: <u>18,19,20</u>]. The Integrative Reactivation and Rehabilitation (IRR) program involves hospitalization of the person with dementia for at least 13 weeks and limited supervision or training of the caregiver [<u>18</u>]. The supporting program and the training program include residence for both members of the dyad for ten days, with the focus on both [<u>19,20</u>]. (Numbers of short-period programs are written in standard font, long-lasting programs in italics, and other programs are <u>underlined</u>.

Most programs consist of multiple treatment components, including information, training for activities of daily life (ADL), walking or exercise, and environmental adaptations for the person with dementia; and information, psycho-education, skills training, and coping strategies for the caregiver. Targeted functional domains include behavioral problems, cognitive functioning, mood, independence in daily activities, sleep, and quality of life of the person with dementia; and mood, burden, competence, and quality of life of the caregiver. The intervention targets of two programs, the Reality Orientation Program [15] and the Sleep-Supporting Intervention [2], involve one single functional domain. The other programs target two or more functional domains for change. Some programs aim at reducing the time to institutionalization [12,16,17, 19,20].

All 20 programs claim to tailor their interventions to the dyad's needs. Eight of the 20 programs start with a needs assessment for the caregiver, and some programs also assess the needs of the person with dementia, using an interview or structured observation, followed by individual goal setting [1,2,4,6,8,11,12,<u>18</u>]. In contrast, the other 12 programs immediately start with treatment sessions and tailor the content to the clients during the program.

review
the
⊒.
σ
lde
СГ
.⊑
udies
stl
σ
Ē
ъ
ns
a
Б
õ
2
fI
č
ō
÷
ē
Ξ
š
Š
m
<u>e</u>
å
Ĥ,

	istruments for CG	Statistically significant outcomes with [®]	- FCBI*** - WHO QoL - BREF *** - SSQ 6 - FSSI** - NPI**	SDI	RMBPC*
	Measurement ir for PD	Statistically significant outcomes with®	- MMSE - Institutio- nalization ***	 Total sleep/ awake time at night, pr, p2, p3** Number of Number of Time in bed Time in bed aytime sleep or inactivity 	- Qol-AD*** - SF-36 - GDS** - COM- FAM - COM- FAM - PSS - Self-efficacy scale
		MMSE Mean (5D) P-C baseline	P 17.5 (4.7) C 17.3 (3.9)	P 19.2 (7.7) P 17.9 (7.0) P 19.1 (5.8) C 18.7 (6.9)	P 23.2 (4.7) C 24.0 (3.8)
STUDY		Group size P=program C=control Control Condition Interval after baseline	P=46 C=46 Usual care 6, 12, 18 months	P1=32 P2=34 P3=33 C=33 Three sessions, Three sessions, not directed at sleep problems, walking, or light 2, 6 months	P=96 C=46 Educational leaflets from Alzheimer's Association 10 weeks
	onents	Component for CG	 Needs assessment Information Psycho-Education Problem solving, sharing with peers support from health resources Improvement of hom care Finance skills 	- Information - Psycho-education - Daily sleep log	- Information - Psycho-education
	Treatment compc	Components for PD	- Needs assessment - Environmen tal adaptations and memory aids	- Individual sleeping plan with - Walking - Light-exposure	 Information Developing strategies for coping with dementia
	Functional domain	Primary goal: - Functional domain PD - Functional domain CG	PD: none CG: Health status, Quality of life	PD: Sleep-wake activity CC: Distress with nocturnal behaviors	PD: Quality of life Health status Mood CG: Quality of life Mood Stress
		Emphasis on PD and CG aimed at mild or moderate dementia	CG mild to moderate	PD-CG mild	PD-CG mild
	istics	Provided by	Case manager social worker	Professional with master degree level	Professional with master degree level
PROGRAM	Delivery characte	Duration Contacts/home visits (Dosage)	6 months 4 HVs for needs assessment (oces(y), (bi-week(y) (2-3)	8 weeks 6 HVs of 60 mins each (3)	9 weeks 9 GSs of 90 mins each for caregivers and persons with dementia, partly separated during the session (3)
			 Dementia Family Care Program for home-residing persons with dementia DFCP (Chien & Lee, 2011³³⁾ 	2 Night-time Insomnia Treatment and Education in Alzheimer's Disease NITE (MCCurry etal., 2011 ⁵)	Early-Stage Memory Loss Support Groups (Logsdon etal. 2010 ³⁹)

29

	struments for CG	Statistically significant outcomes with°	 Caregiver Upset. 4 months^{**}, 6 months^{**}, 6 months^{**}, confidence in managing care, 4 months^{**}, 6 months^{**} - 2Els (1.2 items), 4 months^{**} - Erceived Change Indax, 5 4 month^{***}, 6 months^{***}, 6 months^{***}, 6 months^{***}, 6 months^{****}, 6 months^{****}, 6 months^{****}, 6 months^{****}, 6 months^{****}, 6 months^{****} 	- Perceived Change Index, 4 months ^{**} - Confidence in using activities, 4 months ^{**}	 - ZBI, - Hours doing for the patient, 4 months^{**} - Hours feel on duty, 4 months^{**} - CES. - Confidence in using activities, 4 months^{**} - TMSI
STUDY	Measurement in: for PD	Statistically significant outcomes with [®]	Behavioral occurrences**	- Qol-AD - Activity Engagement, 4 months° - Aultonal dependence ADL, IADL, 4 months°	- ABID, 4 months** - CSDD - Activity engagement, engagement, engagement, - Pleasure in recreation - QoL scale - Ability to keep busy.4 months*
		MMSE Mean (SD) P-C baseline	Р 13.1 (8.2) С 12.8 (8.1	P 13.1 (8.2) C 13.6 (7.9)	P 11.0 (7.3) C 12.2 (8.8)
		Group size P=program C=control Control Condition Interval after baseline	P=137 C=135 Usual care 4, 6 months	P=102 C=107 - Three telephone calls calls reational materials 4, 9 months	P= 30 C= 30 Waiting list 4 months
	onents	Component for CG	 Interview preferences and goal setting Information, Skill training: Information, Psycho-education, stress manage ment 	see 6a	 Pleasant Event schedule Psycho-education Skill training in communicative techniques
	Treatment compo	Components for PD	 Assessment undiagnosed medical conditions Structured observation strategies: Environmental adaptations Assistive devices Engagement in activities 	see 6 a	- Structured observation - Pleasant Event schedule - Trainingfor three activities - Environmental modifications
	Functional domain	Primary goal: - Functional domain PD - Functional domain CG	PD: Frequency of behavioral occurrences CC: Upset confidence in managing care	PD: Functional dependence CC Confidence in using activities	PD: - Frequency of behavioral occurrences - Activity engagement CC: Burden mastery
		Emphasis on PD and CG aimed at mild or moderate dementia	PD-CG moderate	see 6a	PD-CG moderate
	stics	Provided by	Occupational therapist (OT) Nurse (N)	see 4a	Occupational therapist
PROGRAM	Delivery characte	Duration Contacts/home visits(Dosage)	6 months max 9 HVs of 90 mins each (OT), 1 HVN lin month 1-4; 1-4; months 5-6 (3)	4 months maximum 10 HVs of 90 minutes each (OT)in much 1-4 1 HV N, 1TC N (3)	4 months maximum 6 HVs of 90 minutes each 2 TCs of 15 minutes (2-3)
			4a Advanced Caregiver Training ACT (Citlinet al., 2010A ²²)	 4b Care of Persons with Dementia in their Environ- ments COF (Citlin et al., 20108²³) 	5 Tailored Activity Program TAP (Citlinet al., 2008 ²³)

	Delivery characte	ristics		Functional domain	Treatment compo	nents	AUUY		Measurement for PD	instruments for CC
	Duration Contacts/home visits(Dosage)	Provided by	Emphasis on PD and CC aimed at mild or moderate dementia	Primary goal: - Functional domain PD - Functional domain CG	Components for PD	for CG	Group size P=program C=control Control Condition Interval after baseline	MMSE Mean (SD) P-C baseline	statistically significant outcomes with°	Statistically significant outcomes with [®]
6a Community 6b Occupational Therapy in Dementia COTID	5 weeks 10 HVs of 60 minutes each (3)	Occupational therapist	PD-CG mild, moderate	PD: Daily functioning CG: Competence	 Interview preferences and goal setting Structured observation 	- Interview preferences and goal setting - Information, - Sycho-education	P=68 C=67 Usual care 6, 12 weeks	P 19.0(5.7) C 19.0(4.0)	- AMPS**** - IDDD**** - CSDD**** - Dqol**** - GHQ***	- SCQ*** - CES-D*** - GHQ*** - Dq0!*** - Mastery Scale***
(uraff et al., 2005 ¹⁹ , 2007 ²⁹) (Voigt-Radloff et al., 2011 ²⁰)					- rraining or meaningful activities with compensation strategies - Environmental adaptations	(penavioral management) - Stress management	P=54 C=50 One home visit counseling: leaflet on coping with dementia 6 weeks, 4, 6, 12 months	P 20.4 (3.1) C 19.0 (3.3)	- PRPP - IDDD - CSDD - CSDD - Dqol - SF-12 physical, mental	-SCQ -CE-D -Dqol -SF-12 physical, mental -SDL care (hours per day)
7 Partners in Caregiving: A Psychoeducati- on Program (Plepburn et al., (Hepburn et al., 2005 ⁴⁹)	6 weeks 6 weekly GSs of 120 mins each (3)	Multidiscipli- nary team	mid	PD: - CG: Distress Burden	- Activity groups (occupational or music therapist)	 P1: psycho-education P2: same as P1. but completed with a Decision Making framework paying attention to values and preferences Homework Demonstrations of effective management- effective management- effective and preferences 	P1=79 P2=72 C=64 Usual care 6, 12 months	C 17.12		- Distress measure, 6 months° - BACS, 6 months° - Competence, 12 months°
Ba Environmental Skill-building Program ESP (Citlin etal., 2003 ⁵⁴)	12 months 5 HVs of 90 mins each, 1 TC in months 1- 6; 1 HV, total 3 TCs in months 6-12 (3)	Occupational therapist	CG-PD moderate	PD: - Behavioral occurrences - Dependence ADL - ADL - ADL - ADL - CC: - Stressreduction - Competence	- Environmental adaptations - Assistive devices	 Interview preferences and goal setting Information, Psycho-education Skill training: Skiplifying task and communication 	P=89 C=101 Usual care 6 months	C12.5(7.1) C12.5(7.1)	- RMBPC - FIM ADL - FIM IADL	 RMBPC: upset withmemory- related behaviors, 6 months" RMBPC: upset with disruptive behaviors Houptive behaviors Houptive behaviors Hoars providing care Days receiving help, 6 months" Mastery Index TMSI Perceived Change Index (QoL) affect, 6 months"

rement instruments for CG	sically Statistically cant significant nes outcomes with ^o	PC - Behavior ADL self-efficacy aDL ⁶ - ADL self-efficacy - IADL self-efficacy - IADL - Behavior upset - ADL upset - IADL upset	ssion	C - ZBI, 5 months* - RMBPC
Measu for PD	Statist signific outcor with [®]	RMBF - FIM A - FIM I - FIM I	 SF36* SIP* CSDD CSDD Time admis 	RMBPC
	MMSE Mean (SD) P-C baseline	~	P 17.6 (6.8 C 15.9 (7.4	P 17.81 (7.10) C 19.20
STUDY	Group size P=program C=control Control Condition Interval after baseline	P=93 C=78 Usual care, educational materials, for safety in the house 3 months	P=76 C=77 Usual care 3, 6,12,18, 24 months	P=52 C=31
lents	Component for CG	- Interview preferences and goal setting - Information - Psycho-education - Skill training: simplifying task and communication	 Aerobic, strength, balance, and flexibility training Psycho-education (behavioral management) Pleasant activities 	- Information - Psycho-education
Treatment compo	Components for PD	- Environmental adaptations - Assistive devices	- Aerobic, strength, balance and flexibility training	- Test on cognitive functioning
Functional domain	Primary goal: - Functional domain PD - Functional domain CG	PD: - Behavioral occurrences - Dependence ADL - Dependence IADL CG: Upset Self-efficacy with behavior and ADL/IADL	PD: - Physical function - Affective status/ depression CG:	PD: Behavioral problems
	Emphasis on PD and CG aimed at mild or moderate dementia	cc-PD moderate	PDCG moderate	CG-PD mild to severe
eristics	Provided by	Occupatio- nal therapist	Experienced home health professionals	Multidiscipli- nairy team
PROGRAM Delivery charact	Duration Contacts/home visits (Dosage)	3 months 5HVs 90 mins each, bi-weekly (2)	3 months, total of 12 HVs of 60 mins each (4)	7 weeks 7 weekly GSs 120 mins
		8b Environmental Skill-building Program ESP (Gitlin et al., 2001 ⁵⁵)	 Reducing Disability in Alzheimer Disease RDAD (Teri et al., 2003⁵⁶) 	 Minnesota Family Workshop MFW Psycho-educatio-

	: instruments for CG	Statistically significant outcomes with [°]	- 5CQ - 5F-36 - CES-D - SPPIC	Use and costs of care services, 24 months ⁶	- GHQ* - ZBI* - NPI
	Measurement for PD	Statistically significant outcomes with [®]	DQol	Time to long- term institu- tionalization, 18 months°	- EASI - NPI
		MMSE Mean (SD) P-C baseline	P 22 (4.2) C 22.7 (3.8)	Р 13 4 (6.2) С 14.2 (6.6)	Mild, moderate
STUDY		Group size P=program C=control Control Condition Interval after baseline	P=54 C=45 Usual care 6, 12 months	P= 63 C= 62 - Written info - Referals to community services - Contact with study nurse during assess- ments (0, 6, 12 months) 6,12,24 months	P=41 C= 40 Education on dementia 3, 6 months
	ıents	Component for CG	 Assessment, Information, Information, Planning, organizing, organizing, and monitoring of care; Facultative: Group support program 	- Interview preferences and goal setting - Tailored support plan in collaborati- on with couple GS: - Information - Emotional support - Services services	 Information and ad- vice for regulations Emotional support Psycho-education Practical support Family networking
	Treatment compor	Components for PD	 Assessment Information Informing the primary care physician Refearing to other health care professio- nals 	- Exercise training	- Consult psychiatrist - When useful: medication
	Functional domain	Primary goal: - Functional domain PD - Functional domain CG	PD: Quality of life CC: Competence, Quality of life	PD: Delay of institutionali- zation CG: Use of services	PD: Behavioral problems CG: Burden Health status
	istics	Emphasis on PD and CG aimed at mild or moderate dementia	CG-PD mild	CG mild, moderate	CG mild, moderate
		Provided by	District nurse	Family care coordinator (trained public health nurse)	Home care advisor (being literate, higher sec. school) Psychiatrist
PROGRAM	Delivery characte	Duration Contacts/home visits (Dosage)	12-month minimum: 2 HVs + TC every 3 months; more contact if necessary (1-2)	Max. 2 years flexible HVs (mean 3 xa year) flexible TCs (mean 15x year) 5 GSs for spouse CG 5 GSs for PD (4)	6 months HV minimum, bi-weekly (total 12 HVs or more); TC GS (voluntary) (3-4)
			<pre>11 Case management (Jansen et al., 2011³¹)</pre>	Multicomponent support program (Eloniemi et al., 2009 ⁴³)	 Home Care Program Goa, India (Dias et al., 2008⁴³)

	struments for CG	Statistically significant outcomes with [®]	- NPI, 12 months* - PHQ (9 it), 18 months*	- Caregiver Burden Inventory - HRDS - SF-36	- ZBI, 6 months° - RMBPC, 6 months°
	Measurement in for PD	Statistically significant outcomes with [°]	- NPI, 12, 18 months** - CSDD (by CG) - ADL-ADCS - ADL-ADCS - Telephone version of MMSE - Pharmacologi - Pharmacologi - Health care resource use	- MMSE, 6 months* - ADAS - Cognition, 6 months** 6 months** - BI - Number impaired IADL - NPI	Days to admission
		MMSE Mean (SD) P-C baseline	P17.5 (5.2) C18.6 (5.9)	P 20.2 (3.3) C 19.9 (3.0)	P 22.7 (3.8) 22.8 (4.2)
STUDY		Group size P=program C=control Control Condition Interval after baseline	P=84 C= 69 Augmented usual care: face-to-face counseling communicating and written information 6,12,18 months	P=79 C=77 Donepezil 6 months	P=37 C=38 Usual care + information package 3,6,10,14,18 months
	nents	Component for CG	 - Communicating diagnosis - Information, legal and financial advice - Skill training in communication - Psycho-education - Faculative: 1-8 behavioral - faculative: 1-8 behavioral intervention 	- Education: training ROT (one meeting) - Manual	 Psycho-education (behavioral management) When appropriate; Home services, respite, Occupational therapy Social worker
	Treatment compo	Components for PD	 Cholinesterase inhibitors Exercise guidelines with a guidebook and video Voluntary group sessions for exercise 	- Three reality orientation sessions by caregivers each week (30 mins) week (30 mins) week (30 mins) the day the day	When appropriate; - Home services, respite - Occupational therapy - Social worker
	Functional domain	Primary goal: - Functional domain PD - Functional domain CG	PD: Behavioral problems CC: Stress Health status	PD: Cognitive functioning CG: Well-being	PD: - Long-term planning and use of services - Time to admission CC: - Burden - Mood
		Emphasis on PD and CG aimed at mild or moderate dementia	сс ыід	CG mild, moderate	CG-PD mid
	ristics	Provided by	Primary care physician and geriatric nurse practitioner	Training by physicians, psychologists and therapists	Case manager (nurse) and multidiscipli- nairy team
PROGRAM	Delivery charact	Duration Contacts/home visits (Dosage)	1 year face-to-face consultations in primary care primary care when needed (mean 1 per month) (4)	25 weeks [by caregiver: three times a week(30 minutes)] (1)	18 months minimum 1 TC or HV a month, more when needed (4)
			14 Collaborative care for Older Adults with Alzheimer Disease (Callahan et al., 2006 ⁴⁵)	IS Reality Orientation with cholinesterase inhibitors (Onder et al., 2005 ⁴⁷)	16 Early Home Care Program (Chu et al., 2000 ⁶⁰)

STUDY	Measurement instruments for PD for CG	Statistically significant outcomes with°		- Hours of caregiving - ZBI, 6 months ⁶ - CDS, 18,24 months ⁶	 NPI distress, 3months* CB.3 months* CB.3 months*** CCL, 3 months*** 9 months** 	
		Statistically significant outcomes with°	Time to admission		 NPIsymptoms, Months⁴% months⁴ NPI severity, NPI severity, months⁴, 9 months⁶ MMSE; months⁶ BI, 3 months⁶ SF-20 EQ 5D Length of stay in nursing home 	Time to long-term institutionali- zation, 12 months°
		MMSE Mean (SD) P-C baseline	1 0		P 20.04 (4.50) C 20.58 (3.84	P14.4 (6.2) C15.3 (5.5)
		Group size P=program C=control Control Control Condition Interval after baseline	P 1 3965 P2 4130 C 3944	P1+P2 2731 C 2576 Usual care 6,12,18, 24,30, 36 months	P=81 C=87 Usual care 3, 9 months	P=53 C=47 Usual care 12, 24 months
PROGRAM	Treatment components	Component for CG	Case management Model A sites: 1:100 Model B sites: 1:30 When appropriate; - Home services, respite - Therapies		- Psychological counseling - Family therapy	 - Advocacy - Comprehensive support - Counseling - Assistance social and healthcare services - 24-hour availability
		Components for PD	Case management Model A sites: 1:100 Model B sites: 1:30 When appropriate; - Home services, respite - Therapies - Adaptive and assistive equipment		 Diagnostic Diagnostic assessment assessment plan with counse- plan with counse- ling, cognitive behavioral therapy, behavioral therapy, support, rehabilitation, and psycho-education 	- Advocacy - Comprehensive support - Assistance social and healthcare services
	Functional domain	Primary goal: - Functional domain PD - Functional domain CG	PD: Time to admission CG: - ADL/IADL Assistance - Burden - Mood		PD: Neuro- psychiatric symptoms CG: Burden, Competence	PD: - Time to admission - CG:
	Delivery characteristics	Emphasis on PD and CG aimed at mild or moderate dementia	CC-PD mild to severe		PD mild	CG-PD moderate
		Provided by	Social worker or nurse		Multi- disciplinary team	Family care coordinator (nurse)
		Duration Contacts/home visits (Dosage)	3 years Case management for Model Asites: 11:00 - less financial resources Model B sites: 1:30 + more financial resources (?)		Clinical stay of 13 weeks in a psychiatric nursing home unit (4)	2 years - 10-day clinical training program - Flexible HVs, Men needed - Annual training courses, 5 days (4)
			 Medicare Alzheimer's Disease Demonstra- tion Evaluation MADDE (Miller, Newcomer et al., 1999⁶⁴) (Newcomer, Miller et al., 1999⁶⁵) 		18 Integrative Reactivation and Rehabilitation Program IRR (Bakker et al., 2011 ³⁵)	19 Supporting program (Eloniemi-Sulkava et al., 2001 ^{ss})
		ically ant nes	° sth			
---------	------------------------------------	--	---			
	ments CG	Statist signific outcor with [®]	GHQ. 12 mon			
	Measurement instru for PD for (Statistically significant outcomes with°	Time to admission, 30 months P:65% lived in the community, C1:26% C2? 8 years: P 79% in nursing home°, C190%, C2 83%			
		MMSE Mean (SD) P-C baseline	P 17.1 (6.5)			
STUDY		Group size P=program C=control Control Condition Interval after baseline	P 33 CI 32 C2 31 C2 31 training for PD and respite for CC months/ 8 years			
	Treatment components	Component for CG	- Psycho- Education - Skill training - Family therapy - Recreation and outings			
		Components for PD	- Memory training - Rot - Activities - Reminiscence - Recreation and outings			
	Functional domain	Primary goal: - Functional domain PD - Functional domain CG	PD: Time to admission CC: Mental health Health - Mood			
	Delivery characteristics	Emphasis on PD and CG aimed at mild or moderate dementia	CG-PD mild, moderate			
		Provided by	Program coordinator in psychiatric hospital, multidiscipli- nary team			
PROGRAM		Duration Contacts/home visits (Dosage)	10 days in-hospital training group TCs 2-6 weeks (4)			
			Training program (Brodary, Gresham 1989 ²¹) Prince Henry Hospital program (Brodary et al., 1997 ⁷⁰)			

Notes: Dosage/intensity: 1 = minimal; (1-2 sessions), 2 = moderate (3-5 sessions), 3 = medium high (6-10 sessions), 4 = high/intensive (>10 sessions; Brodaty et al., 2003) * significant p = 0.05; ***p = 0.01, ****p = 0.001. Explanation of abbreviations: ABID = Agitated Behavior In Dementia, Alzheimer Disease Cooperative Study; ADA5 = Alzheimer Disease Assessment Scale; ADL - ADC5 = Activities of Daily Living-Alzheimer Disease Cooperative Study Group; ADL = Activities of Daily Living; AMP5 = Assessment of Motor and Process Skills; BAC5 = Beliefs About Carregiving Scale; BJ = Barthel Index; CB = Carregiver Burden; CCL= Caregiver Competence List; CE5-D = Centre of Epidemic Studies - Depression; CG=Caregiver; CSDD = Cornell Scale Depression in Dementia; COM- FAM Scale is and interpersonal Relationships; subscale of Family Assessment Measure; Dqol = Dementia Quality of Life instrument; EQ5D = Euro-Quality of Life 5D; EASI= Everyday Abilities Scale for Indit; FCBI = Family Caregiving Burden Inventory; FIM = Functional

Independent Measure: FSSI = Family Support Services Index; GDS = Geriatric Depression Scale; GHQ = General Health Questionnaire; GS = group session; HY = home visit; HRDS = Hamilton Rating Depression Scale: IADL = Instrumental Activities of Daily Living: IDDD = Interview for Deterioration in Daily Living Activities in dementia; Performance scale: Intitative scale: MMSF Mini Mental State Examination; NPI = Neuro-Psychiatric Inventory; PD= Person with Dementia; PHQ = Patient Health Questionnaire; PRPP = Perceive Recal IPlan and Perform System of Task Analysis; PSS = Perceived Stress Scale; Qot AD = Quality of Life – Alzheimer's Disease Scale; MBPC = Revised Memory and Behavioral Problems Checklist; SCQ = Sense of Competence Questionnaire; Physical scale, Mental scale; SIP = Sickness Impact Profile; SPPIC = Self Perceived Pressure Informat Care; SSQ = Social Suport Questionnaire; TC = telephone contact; T5MI = Task Management Strategy Index; WHO QoL BREF = World Health Organization Quality of Life Measure-Brief Persion; SBI = Zarit Burden Interview; K

Study characteristics and strength of the body of evidence

The studies varied with regard to measurement instruments, control conditions, and/or time to follow-up (Table 3). "Usual care" and "waiting list" are the most often used control conditions [1, 4a, 5, 6a, 7, 8a, 9, 10, 11, 15, 17, 19, <u>18, 20</u>]. Some studies use information leaflets for the informal caregiver [3, 4b, 8b, 12, 13, 16], or one to three face-to-face contacts, or telephone contacts [2, 6b, 12, 14, <u>20</u>] in the control condition. Following the GRADE approach, four limitations influence the strength of the body of evidence. Two of them, lack of blinding of participants and therapists as well as indirectness of evidence (the control condition is usual care), are realistic for studies in the current field. The other two limitations are apparent in the studies: a short follow-up period or heterogeneity of results (e.g. significant outcomes at different follow-up moments) [12, 14, 19, <u>20</u>]. Although all studies targeted both members of the dyad, two studies had outcome measures for the person with dementia only [9] or for the caregiver only [7].

Effects of dyadic psychosocial programs

Eleven of the 23 studies concerning ten programs showed statistically significant positive effects for both members of the dyad [1, 3, 4a, 4b, 5, 6a, 8b, 14, <u>18</u>, <u>19</u>, <u>20</u>]. Four studies showed statistically significant effects for the person with dementia only [2, 9, 12, 15], and six studies found statistically significant effects for the caregiver only [13, 7, 8a, 10, 16, 17]. This was partly because these studies had no effect on outcomes of interest for the other person of the dyad, and partly because outcomes were not measured for the other person of the dyad (Table 3). The two remaining studies did not show any statistically significant effects [6b, 11].

Effects on the person with dementia

Behavioral Problems

Eight of the 23 studies, concerning eight programs, measured behavioral problems, [4b, 5, 8a, 10, 13, 14, 15, <u>18</u>] (Figure 2).The strength of the body of evidence for this outcome is moderate. Three of the eight studies had positive outcomes for behavioral problems (neuropsychiatric symptoms) [5, 14, <u>18</u>]: one short-period program, one long-lasting program, and one program with hospitalization. The three programs comprised different intervention components for each member of the dyad. The other five studies that did not show statistically significant effects on behavioral problems involved programs with comparable intervention components, both short-period programs [4b, 8a, 10] and long-lasting programs [13, 15]. There was no evident relation between intervention components and the outcome of "behavioral problems". The IRR-program with hospitalization [<u>18</u>] showed positive effects on people with mild dementia but high scores on behavioral problems. For example, the long-lasting Collaborative Care program [14] showed positive effects on people with moderate dementia and regular behavioral problems and the short-period TAP- program [5] showed positive effects on people with moderate dementia and regular behavioral problems.



(1) Ostwald; groupprogram

SD: Standard Deviation; CI: Confidence Interval

Figure 2 Person with dementia outcomes: behavioral problems

Mood

Six of the 23 studies, involving five programs, measured the mood of the person with dementia as a separate outcome [3, 5, 6a, 6b, 9, 14] (Figure 3). The body of evidence for this outcome is strong because of longer follow-up periods and comparison with a control condition other than usual care, although there is heterogeneity in the results of the studies on the Community Occupational Therapy in Dementia (COTiD) program [6a, 6b]. Three of the six studies showed statistically significant positive effects on the mood of the person with dementia [3, 6a, 9]; these were all short-period programs: one group program for early-stage dementia and two individual programs with home visits for mild and moderate dementia. In all three programs the professional involved the person with dementia actively in group sessions [3], activities [6a] or exercise [9]. The study on the long-lasting Collaborative Care program showed a trend toward positive effects on mood, although this was not statistically significant [14]. The two remaining studies, both involving a short-period program, did not show a statistically significant effect [5, 6b], although they comprise intervention components comparable to those of the programs with statistically significant effects.



(1) Logsdon: groupprogram

SD: Standard Deviation; CI: Confidence Interval

Figure 3 Person with dementia outcomes: mood

Daily activities

Ten of the 23 studies, involving eight programs, measured independence and engagement in ADL [4b, 5, 6a, 6b, 8a, 8b, 13, 14, 15, 18] (Figure 4). The strength of the body of evidence for this outcome is moderate. Five of the ten studies showed statistically significant positive effects [4b, 5, 6a, 8b, 18]. Four of these concern short-period programs in which the professional actively involves both the person with dementia as the caregiver in skill training [4b, 5, 6a, 8b]. The intervention components in these programs are daily activity training, choosing meaningful (pleasant or purposeful) activities, and environmental adaptations for the person with dementia; and psycho-education and skills training for the caregiver. Whereas these four studies measured instrumental ADL (IADL), the outcome of the fifth study [18], concerning the IRR program, was personal care, measured with the Barthel Index. One other study, the long-lasting Reality Orientation [15] program, shows a trend toward positive effect on personal care, also measured with the Barthel Index. Of the four remaining studies, two long-lasting programs [13,14], showed no statistically significant effects. The other two were trials of COTiD and the Environmental Skill-Building Program [6b, 8a], and thus they had inconsistent results.



SD: Standard Deviation; CI: Confidence Interval

Figure 4 Person with dementia outcomes: ADL/IADL

Quality of life

Eight of the 23 studies, concerning seven programs, measured the quality of life of the person with dementia [3, 4b, 5, 6a, 6b, 9, 11, <u>18</u>] (Figure 5). The body of evidence for this outcome is moderate to strong. Four of the eight showed a statistically sig-nificant better quality of life of the person with dementia [3, 5, 6a, 9]. Another study showed a trend toward better quality of life [4b]. These are all short-period programs: one group program for people with early dementia [3] and four individual programs with home visits and training [4b, 5, 6a, 9]. These studies also showed positive effects on two other outcomes: mood [3, 6a, 9] and ADL/IADL dependency [4b, 5, 6a]. The three remaining studies showed no statistically significant effect [6b, 11]. The third study of the IRR-program with hospitalization did not show effect on quality of life although, it was effective for behavioral problems [<u>18</u>].



(1) Logsdon groupprogram

SD: Standard Deviation; CI: Confidence Interval

Figure 5 Person with dementia outcomes: quality of life

Institutionalization

We studied "institutionalization" or "time to admission" for seven studies [1, 9, 12, 16, 17, <u>19</u>, <u>20</u>]. The body of evidence for this outcome is moderate to strong. One short-period program [1], one long-lasting program [12], and two programs with hospitalization [<u>19</u>, <u>20</u>] significantly reduced institutionalization or the time to institutionalization. Another program, the long-lasting Early Home Care Program [16], also had significant effects, although for a subgroup with a Mini-Mental State Examination (MMSE) of less than 23 only. The short-period Reducing Disability program [9] showed a trend toward delaying institutionalization. The Medicare Alzheimer's Disease Demonstration program [17] showed no significant effects. The data for this outcome were not suited for quantitative comparison. In more recent studies 'institutionalization' is less often studied. Next it is more often included as an outcome measure in studies of long-lasting programs.

Effects for the caregiver

Mood

Nine studies involving eight programs measured the mood of the caregiver as a secondary outcome [4a, 5, 6a, 6b, 10, 11, 14, 15, 17] (Figure 6). The body of evidence for this outcome is moderate to strong. Two studies of short-period programs showed significant positive effects after the intervention [4a, 6a]; another study, the longlasting Collaborative Care program, showed significant effects at 18 months, but not at earlier intervals [14]. A study of the Tailored Activity program showed a trend toward positive effects, but statistical significance was not reached [5]. The intervention components of these four programs include

information, psycho-education, and communication skills training for the caregiver. The other five studies, involving both short-period and long-lasting programs, did not show statistically significant effects [11, 6b, 10, 15, 17]. Four of these programs lack the communication skills-training component [10, 11, 15, 17].



(1) Ostwald groupprogram

SD: Standard Deviation; CI: Confidence Interval



Perceived burden and competence

Seventeen studies involving 15 programs measured the perception of providing care with burden and/or competence questionnaires: burden [1,_3, 4a, 5, 7, 8a, 8b, 10, 11, 13, 14, 15, 16, 17, <u>18</u>], competence [4a, 4b, 5, 6a, 6b, 7, 8a, 11, <u>18</u>], or both [4a, 5, 7, 8a, 11, <u>18</u>] (Figure 7). The strength of the body of evidence for this outcome is moderate. Thirteen studies, eight short-period programs [1, 3, 4a, 4b, 6a, 7, 8a, 10], four long-lasting programs [13, 14, 16, 17], and one program with hospitalization [<u>18</u>] showed significant positive effects for burden/competence, although not at all moments of follow-up. The programs with statistically significant effects included varying intervention components. The remaining four studies without significant effects involve both short-period programs and long-lasting programs [5, 8b, 11, 15]. It is not clear which intervention components of the 15 programs are effective and which are not. Six of the 17 studies measured burden and perceived competence. In three studies, the scores for the two concepts were in the same direction: both concepts had significant effects [4a, <u>18</u>] or both had non-significant effects [11]. The other three studies had statistically significant effects for one outcome [5, 7, 8a].

Experimental		с	ontrol		Std. Mean Difference	Std. Mean Difference		
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	IV, Random, 95% CI	IV, Random, 95% CI
2.2.1 Intensive, limited								
COTiD Graff 07	-104.6	13.4	67	-88.4	13.7	65	-1.19 [-1.56, -0.82]	
DFCP Chien (1)	56.9	14.8	46	64	13.1	46	-0.50 [-0.92, -0.09]	
ACT Gitlin	19	8.5	117	21	9.3	122	-0.22 [-0.48, 0.03]	-+-
ESP Gitlin 03	0.43	0.5	89	0.56	0.66	101	-0.22 [-0.51, 0.07]	-++
TAP Gitlin	20.3	8.8	27	20.6	10.4	29	-0.03 [-0.55, 0.49]	-+-
ESML Logsdon (2)	1.07	0.7	92	1.04	0.7	44	0.04 [-0.32, 0.40]	+
PIC Hepburn	36.17	12.25	120	34.91	14.53	46	0.10 [-0.24, 0.44]	-
MFW Ostwald (3)	56.82	11.83	50	55.43	15.91	30	0.10 [-0.35, 0.55]	-
COPE Gitlin	7.5	1.9	102	6.9	2.5	107	0.27 [-0.00, 0.54]	- + -
COTiD Voigt-Radloff	-103	18.7	50	-108.6	17.2	47	0.31 [-0.09, 0.71]	++-
2.2.2 Extensive, long	lasting							
Coll-care Callahan	3.5	5.8	84	7.7	8.7	69	-0.58 [-0.90, -0.25]	-+-
Home Care Dias	19	13	33	21.4	16.2	26	-0.16 [-0.68, 0.35]	
MADDE Newcomer	14.1	8.07	1702	14.4	8.62	1579	-0.04 [-0.10, 0.03]	+
Early homecare Chu	27.1	0	27	29.5	0	21	Not estimable	
ROT Onder	2	11.71	70	1.3	12.27	67	0.06 [-0.28, 0.39]	+
Caseman Jansen	-47.4	6.44	41	-48.4	6.4	38	0.15 [-0.29, 0.60]	-+
2.2.3 Temporary hospitalization								
IRR Bakker	36.75	25.81	60	44.58	28.24	65	-0.29 [-0.64, 0.07]	-+-
							F	avours experimental Eavours control

(1) Chien partly groupprogram

(2) Logsdon groupprogram

(3) Ostwald groupprogram

SD: Standard Deviation; CI: Confidence Interval

Figure 7 Caregiver outcomes: burden and/or competence

Quality of life

Ten studies measured the *quality of life* of the caregiver [1, 4a, 4b, 6a, 6b, 8a, 11, 13, 15, <u>20</u>] (Figure 8). The body of evidence for these studies is moderate. Seven of the ten studies found statistically significant effects [1, 4a, 4b, 6a, 8a, 13, <u>20</u>]. The programs are from all three categories. These seven studies also showed significant effects on other outcomes. Three other studies showed no statistically significant effect on the quality of life of the caregiver [6b, 11, 15]. Two of the programs, Case Management [11] and COTiD program [6b] showed no significant effects on any outcome, and the Reality Orientation program only showed significant effects on outcomes for the person with dementia [15]. The intervention components of this program focus primarily on the "cognition" of the person with dementia, and do not involve caregiver feelings.

	Exp	eriment	tal	c	Control	:	Std. Mean Difference	Std. Mean Difference
Study or Subgroup	Mean	SD	Total	Mean	SD	Total	IV, Random, 95% CI	IV, Random, 95% CI
2.4.1 Intensive, limite	d							
COTiD Voigt-Radloff	3	0.6	51	3.1	0.7	48	-0.15 [-0.55, 0.24]	_ +
ESP Gitlin 03	3	0.54	89	2.89	0.56	101	0.20 [-0.09, 0.48]	++
DFCP Chien	74.1	14.8	46	69.8	16.7	46	0.27 [-0.14, 0.68]	++-
COPE Gitlin	3.1	0.6	102	2.9	0.5	107	0.36 [0.09, 0.64]	-+-
ACT Gitlin	3.3	0.7	117	2.8	0.7	122	0.71 [0.45, 0.97]	+-
COTiD Graff 07	4	0.6	67	3.4	0.7	65	0.92 [0.56, 1.28]	
2.4.2 Extensive, long	lasting							
ROT Onder	1.3	11.71	70	1.1	11.45	67	0.02 [-0.32, 0.35]	+
Caseman Jansen	48.2	9.84	45	47.7	11.25	37	0.05 [-0.39, 0.48]	_ _
Home Care Dias	-2.6	2.3	34	-3.3	3.6	31	0.23 [-0.26, 0.72]	-+ +
2.4.5 Temporary nos		011					0.4450.00.0.001	
Training Brodaty	5.3	6.2	33	6.1	8.4	31	-0.11 [-0.60, 0.38]	
							_	
								-2 -1 0 1 2
								Eavours control Eavours experimental

SD: Standard Deviation; CI: Confidence Interval

Figure 8

Caregiver outcomes: quality of life

Discussion

Psychosocial interventions for both people with dementia and their caregivers may have a beneficial impact for both members of the dyad or only one of them. We targeted psychosocial interventions for the dyad in this review. In addition to the meta-analysis of Brodaty and Asaratnam⁹, this meta-analysis builds on previous reviews by extending the focus to also include outcomes for the people with dementia. The terms "nonpharmacological" and "psychosocial" are commonly used interchangeably. In this study, we used the term psychosocial intervention to indicate interventions in which contact between the professional and both the person with dementia and the caregiver was central, with or without other treatment components. In practice these interventions are complementary to other interventions that may be pharmacological, technological, or providing respite care for the caregiver. We searched for effects for both members of the dyad, but unfortunately the data were too diverse, and it was not possible to relate outcomes for both the person with dementia and the caregiver.

Smits et al. conclude that "general health appears to be the most promising target for dyadic programs" for caregivers.¹² In our review, we can be more specific: Programs with intervention components that are related to the targeted functional domains are promising, especially for the outcomes ADL/IADL dependency and competence, adding to better quality of life for both members of the dyad. The increasing number of moderate to good effect studies of psychosocial interventions for people with dementia and their informal caregivers reflects the need for evidence-based interventions for this target group.

Many studies show domain-specific effects, that is, there are statistically significant effects on the outcomes corresponding with the functional domain that the intervention aims to improve. This is the case especially for the outcomes "activity and functional dependence of the person with dementia" and "competence of the caregiver" [4b, 5, 6a, 9], and for the outcome "sleep" in the study of the sleep program [2]. The effects of other outcomes are more heterogeneous, and any relation with the treatment components and delivery characteristics is less evident. Programs of all three categories, i.e. short-period programs, long-lasting programs, and programs with hospitalization, had positive effects on behavioral problems of the person with dementia and on mood and burden of the caregiver. Pinguart and Sorensen have already mentioned the domain-specific effects of intervention components for the caregiver, such as counseling and psychoeducation for active engagement of the caregiver.¹⁸ In this review, we found support for the effectiveness of skills training for the activities of the person with dementia and communication skills training for the caregiver. This can explain the effect sizes being larger than those of the other programs that merely emphasize the role of the caregiver to train the person with dementia. Active training for activities focused directly on both members of the dyad adds to the beneficial effects. All psychosocial intervention programs in this review tailored the intervention to the needs and personal situation of the dyad. Based on this review, we did not find that a structured needs assessment at the start of an intervention is better than tailoring the intervention during the sessions. Further, no effects can be unequivocally attributed to the intensity and duration of the program. Perhaps this will depend on the needs of the person with dementia and/or caregiver and the targeted functional domain of the program. Pinquart and Sorensen, and Brodaty et al. found that longer interventions are more likely to improve mood of the caregiver.^{18,10} Next, Brodaty and Arasaratnam recommend in their review on caregiver interventions that short-period programs were most effective for behavioral and psychological problems of the person with dementia, and also the most effective for botheration, stress, or self-efficacy of the caregiver.9

Heterogeneity of effects

Our review also yielded some conflicting results. COTiD is probably the most illustrative example of this. In the COTiD study [6], Graff et al. show large effect

sizes for all outcome measures, including mood, guality of life, daily functioning, and competence, but these effects were not replicated in successive studies of the same program.^{19,20} Next, Gitlin et al. have demonstrated a decrease in behavioral occurrences of the person with dementia in two studies [4a, 5].^{21,22}; however, the Care of Persons with dementia in their Environments (COPE) study [4b], with comparable intervention components, does not show this effect.²³ Differences in findings in interventions with comparable treatment components might be explained in several ways. Contrary to pharmacological treatment, psychosocial interventions in general, and dyadic interventions in particular, are of a different nature. The question is, can we expect to find clearly defined effects on the total group of caregivers of such complex interventions as included in this metaanalysis. These interventions consist of various treatment components, and the effects will depend on different aspects, such as the specific needs of the caregivers and people with dementia and the fit with the targeted functional domain of the program. In addition, the competencies of the professionals involved and the relationship between care professional, person with dementia and informal caregiver may also play a role. Next, the quality of the intervention may influence outcomes. Leontjevas et al. advocate a process evaluation of the sampling quality and of the intervention quality of trials²⁴. Such evaluations guarantee valid outcomes and, in the case of effective programs, guide implementation of the program.^{25,26} Recruitment strategies have to take into account the fact that participants may differ in several respects, which results in different amounts of improvement. For example, the stage of dementia, the extent of behavioral problems, the mental health of the caregivers, their knowledge, skills, and earlier support may all differ. If caregivers do not know much about dementia, psychoeducation may be a very effective tool. However, if they already know all the ins and outs of the disease, one can expect that psycho-education will not be very effective. Often research participants are already involved in a support network and belong to a help-seeking group.²⁷ The intervention quality depends on experienced trainers and support during the intervention. Next cultural background of participants and trainers as well as the healthcare system in a country influences the effects of a program.²⁸

We did not distinguish between the perceived burden and competence as separate outcomes because the instruments used to measure these concepts in the studies overlap in content. Since it may be important to differentiate between these concepts, developing a new instrument with good psychometric properties may be important. Perceived burden and competence may have different relationships with stressors such as behavior problems.^{21,29} and perceived competence might be easier to improve because it concerns a skill.

We found a relationship with other functional domains: Studies with positive effects on quality of life also had positive effects on other outcome domains for the caregiver as well as for the person with dementia.

Limitations

Results should be interpreted cautiously because of the differences in the programs and studies. First, all studies used validated measurement instruments, but sometimes they used different combinations of subscales, which made results difficult to compare [4a, 4b, 5, 7, 8a, 8b]. Second, the duration of the programs was five weeks to two years. We chose to compare the results up to one year after the intervention because this was the time most likely to do justice to the program. The natural decline due to the dementia process may decrease the effects of long-lasting programs more than the effects of short-period programs. Two long programs [12, 14] and one short program [7] had no significant effects directly after completion of the program, but they did at later follow-up times. The quantitative comparison was also made within more homogeneous subgroups according to the program characteristics, the stage of dementia, and the duration of follow-up, but that did not change the overall figure.

Implications for research

Functional domains of a psychosocial intervention should focus on the needs of the dyad. A structured needs assessment of both members of the dyad before the start to determine whether the intervention is appropriate is challenging because of difficulties in recruitment. However, it might be expected that focusing an intervention on the needs of the dyad would lead to stronger and more realistic effects, which is important in daily practice.

More research into measuring the effects of different dosages, frequency, and intervention components or combinations of components is needed. The outcome "institutionalization" needs more attention. Although studies with a long followup are expensive, this is a minor problem compared to the huge costs for institutional care for a growing number of people with dementia in the next decades. For some programs cost-effectiveness studies are available [5, 6a, 17], and results about cost-effectiveness are needed for choosing appropriate programs for clients with dementia and their caregivers. The heterogeneity of results in this review raises several questions. Did the psychosocial interventions meet the specific needs of the caregivers and people with dementia who were involved in the different studies? Have the interventions been delivered and received in the proposed way? Were the appropriate instruments used for measuring the effects of the complex interventions included in this review, or do we need more individualized outcomes to measure the effects of this kind of interventions? Moreover, individualized analyses, such as time series might also be an option, in which the person has its own control.

Implications for practice

A generic conclusion about the program that works best is not possible because of the broad range of outcomes. Programs that target behavioral problems and/or ADL/IADL dependency seem to be promising. Active training for activities and communication skills improves results for both members of the dyad. This review underscores the need to evaluate key treatment variables and key characteristics of the dyad in order to determine which form of treatment may be more compatible and thus more likely to be beneficial to the person with dementia and the caregiver.³⁰ Choosing an intervention depends on several arguments. The intervention has to meet the problems that a dyad experiences, thus the primary targeted functional domain should match with those problems. Other criteria for choosing an intervention for a dyad are stage of mild or moderate dementia, costs, and availability and feasibility of the program. For example, the IRR program is effective for reducing behavioral problems for persons with relatively mild dementia, but the IRR program with (temporarily) hospitalization is an expensive and intrusive program [18]. Thus this program should be restricted to persons with dementia with severe neuropsychiatric symptoms. Other programs are merely aimed at maintaining functional abilities for the person with dementia and supporting the caregiver to handle the behavioral symptoms and cope with the dementia process and their role as a caregiver [5, 6a, 4a, 4b, 3, 8a, 9, 10]. The preventive character of these less expensive programs over limited time makes these programs suitable for broad application for dyads that recognize these problems. Next, some interventions aimed chiefly at supporting the caregiver in his/her caregiver role for a longer time [1, 12, 19, 13, 14, 7, 17]. When needs of a couple on a certain moment primarily are experienced by the caregiver, these interventions are appropriate, when needed in combination with the limited, short-term programs.

Since a clinician has to determine which program works for which dyad, matching the goals of a program with the needs of both members of the dyad is necessary to support them in their daily lives. Reflection on the results during the program is also necessary, and adaptation or a change to other support programs may be required.

References

- 1. Cuijpers P. Depressive disorders in caregivers of dementia patients: a systematic review. Aging & Mental Health. 2005;9(4): 325-30.
- 2. Peeters J, Van Beek A, Meerveld J, Spreeuwenberg P, Francke A. Informal caregivers of persons with dementia, their use of and needs for specific professional support: a survey of the National Dementia Programme. *BMC Nursing*. 2010;9.
- Pinquart M, Sorensen S. Correlates of physical health of informal caregivers: a meta-analysis. The Journals of Gerontology - Series B, Psychological Sciences and Social Sciences. 2007;62:126-37.
- 4. Aalten P. Behavioural problems in dementia: course and risk factors. (PhD thesis). Maastricht, Universiteit Maastricht, 2004.
- 5. Lyketsos CG, Lopez O, Jones B, Fitzpatrick AL, Breitner J, Dekosky S. Prevalence of neuropsychiatric symptoms in dementia and mild cognitive impairment: results from the cardiovascular health study. *JAMA*. 2002;288:1475-83.
- 6. Dröes R. Focus and effectiveness of psychosocial interventions for people with dementia in institutional care settings from the perspective of coping with the disease. *Non-pharmacological Therapies in Dementia*. 2010;1: 139-161.
- Moniz-Cook E, Vernooij-Dassen M, Woods B, Orrell M. Psychosocial interventions in dementia care research: the INTERDEM manifesto. *Aging & Mental Health*. 2011;15: 283-90.
- 8. Acton GJ, Kang J. Interventions to reduce the burden of caregiving for an adult with dementia: a meta-analysis. *Researh in Nursing & Health*. 2001;24: 349-60.
- Brodaty H, Arasaratnam C. Meta/analysis of Nonpharmacological Interventions for neuropsychiatric Symptoms of Dementia. *American Journal of Psychiatry*. 2012;169(9): 946/953.
- 10. Brodaty H, Green A, Koschera A. Meta-analysis of psychosocial interventions for caregivers of people with dementia. *Journal of the American Geriatrics Society*. 2003; 5: 657-64.
- de Vugt ME, Stevens F, Aalten P, Lousberg R, Jaspers N, Winkens I... Verhey FR. Do caregiver management strategies influence patient behaviour in dementia? *International Journal of Geriatric Psychiatry*. 2004;19(1): 85-92.
- Smits CHM, de Lange J, Dröes RM, Meiland F, Vernooij-Dassen M, Pot AM. Effects of combined intervention programmes for people with dementia living at home and their caregivers: A systematic review. *International Journal of Geriatric Psychiatry*. 2007;22(12):1181-1193.
- 13. Furlan AD, Pennick V, Bombardier C, Van Tulder M. Updated method guidelines for systematic reviews in the Cochrane Back Review Group. *Spine* 2009;34:1929-41.
- 14. Higgins JTP, Green S (eds.) Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration, 2011.

- 15. Thompson CA, Spilsbury K, Hall J, Birks Y, Barnes C, Adamson J. Systematic review of information and support interventions for caregivers of people with dementia. *BMC Geriatrics*. 2007;7:18.
- Olazarán J. Reisberg B, Clare L, Cruz I, Peña-Casanova J, Del Ser T ... Spector A. Nonpharmacological therapies in Alzheimer's disease: a systematic review of efficacy. *Dementia and Geriatric Cognitive Disorders*. 2010; 30(2): 161-178.
- 17. Czaja SJ, Schulz R, Lee CC, Belle SH. A methodology for describing and decomposing complex psychosocial and behavioral interventions. *Psychology and Aging*. 2003;18: 385-95.
- 18. Pinquart M, Sorensen S. Helping caregivers of persons with dementia: which interventions work and how large are their effects? *International Psychogeriatrics*. 2006;18: 577-95.
- Graff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM. Community based occupational therapy for patients with dementia and their care givers: Randomised controlled trial. British Medical Journal. 2006;333:1196-1199.
- Voigt-Radloff S, Graff M, Leonhart R, Schornstein K, Jessen F, Bohlken J ... Vernooij-Dassen M. A multicentre RCT on community occupational therapy in Alzheimer's disease: 10 sessions are not better than one consultation. *British Medical Journal Open*. (2011;1(1): e000096.
- 21. Gitlin LN, Winter L, Burke J, Chemett N, Dennis MP, Hauck WW. Tailored activities to manage neuropsychiatric behaviors in persons with dementia and reduce caregiver burden: A randomized pilot study. *American Journal of Geriatric Psychiatry*. 2008;16: 229-239.
- 22. Gitlin LN, Winter L, Dennis MP, Hodgson N, Hauck WW. Targeting and managing behavioral symptoms in individuals with dementia: A randomized trial of a nonpharmacological intervention. *Journal of the American Geriatrics Society*. 2010; 58:1465-1474.
- 23. Gitlin LN, Winter L, Dennis MP, Hodgson N, Hauck WW. A biobehavioral home-based intervention and the well-being of patients with dementia and their caregivers: the COPE randomized trial. *JAMA*. 2010;304: 983-91.
- 24. Leontjevas R, Gerritsen DL, Koopmans RT, Smalbrugge M, Vernooij-Dassen MJ. Process evaluation to explore internal and external validity of the "Act in Case of Depression" care program in nursing homes. *Journal of the American Medical Directors Association*. 2012;13: 488 e1-8.
- 25. Moniz-Cook E, Manthorpe J. Introduction: Personalising Psychosocial Interventions to Individual need and context. In *Psychosocial Interventions in Early Dementia: Evidence-Based Practice*. London: Jessica Kingsley, 2009.
- 26. Grol R, Wensing M. Implementatie: effectieve verbetering van de patiëntenzorg (Implementation: Effective Improvement of Patient Care]. Amsterdam, Netherlands: Reed Business, 2011.
- Schoenmakers B, Buntinx F, Delepeleire J. Supporting the dementia family caregiver: The effect of home care intervention on general well-being. *Aging & Mental Health*. 2010;14: 44-56.

- 28. Voigt-Radloff S, Graff M, Leonhart R, Hüll M, Olde Rikkert M & Vernooij-Dassen M. Why did an effective Dutch complex psycho-social intervention for people with dementia not work in the German healthcare context? Lessons learnt from a process evaluation alongside a multicentre RCT. *British Medical Journal Open*. 2011;1(1): e000094.
- Graff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM.
 Effects of community occupational therapy on quality of life, mood, and health status in dementia patients and their caregivers: A randomized controlled trial. *Journals of Gerontology* - Series A Biological Sciences and Medical Sciences. 2007; 62:1002-1009.
- Coon DW, Thompson L, Steffen A, Sorocco K, Gallagher-Thompson D. Anger and depression management: psychoeducational skill training interventions for women caregivers of a relative with dementia. *Gerontologist*. 2003;5: 678-89.
- 31. Jansen AP, van Hout HP, Nijpels G, Rijmen F, Dröes RM, Pot AM ... van Marwijk HW. Effectiveness of case management among older adults with early symptoms of dementia and their primary informal caregivers: a randomized clinical trial. *International Journal of Nursing Studies*. 2011;48(8): 933-943.
- Clare L, Linden DE, Woods RT, Whitaker R, Evans SJ, Parkinson CH ... Rugg MD. Goal-oriented cognitive rehabilitation for people with early-stage Alzheimer disease: a single-blind randomized controlled trial of clinical efficacy. *The American Journal of Geriatric Psychiatry*. 2010; 18(10): 928-939.
- 33. Chien WT, Lee IY. Randomized controlled trial of a dementia care programme for families of home-resided older people with dementia. *Journal of Advanced Nursing*. 2011;67: 774-787.
- Carbonneau H, Caron CD, Desrosiers J. Effects of an adapted leisure education program as a means of support for caregivers of people with dementia. Archives of Gerontology and Geriatrics. 2011;53: 31–39.
- 35. Bakker TJ, Duivenvoorden HJ, Van Der Lee J, Olde Rikkert MG, Beekman AT, Ribbe MW. Integrative psychotherapeutic nursing home program to reduce multiple psychiatric symptoms of cognitively impaired patients and caregiver burden: randomized controlled trial. *American Journal of Geriatric Psychiatry*. 2011;19 507-20.
- 36. McCurry S, Pike KC, Logsdon RG, Larson EB, Vitiello MV, Teri L. Increasing walking and bright light exposure to improve sleep in community-dwelling persons with Alzheimer's disease: results of a randomized, controlled trial. *Journal of the American Geriatrics Society*. 2011;59: 1393–1402.
- 37. McCurry S, Pike KC, Logsdon RG, Vitiello MV, Larson EB, Teri L. Walking, bright light, and a combination intervention all improve sleep in communitydwelling persons with alzheimer's disease and caregivers: Results of a randomized, controlled trial. *Sleep.* 2010;33: A353.
- 38. McCurry SM, Logsdon RG, Vitiello MV, Teri L. Nighttime insomnia treatment and education for Alzheimer's disease: a randomized, controlled trial. *Journal of the American Geriatrics Socie*ty. 2005;53: 793–802.

- 39. Logsdon RG, Pike KC, McCurry SM, Hunter P, Maher J, Snyder L, Teri L. Early-stage memory loss support groups: outcomes from a randomized controlled clinical trial. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences*. 2010;65(6): 691-697.
- 40. Logsdon RG, McCurry SM, Teri L. Time-limited support groups for individuals with early stage dementia and their care partners: Preliminary outcomes from a controlled clinical trial. *Clinical Gerontologist*. 2007;30: 5–19.
- 41. Neely AS, Vikstrom S, Josephsson S. Collaborative memory intervention in dementia: caregiver participation matters. *Neuropsychological Rehabilitation*. 2009;19: 696–715.
- 42. Eloniemi-Sulkava U, Saarenheimo M, Laakkonen ML, Pietilä M, Savikko N, Kautiainen H... Pitkälä KH. Family care as collaboration: effectiveness of a multicomponent support program for elderly couples with dementia. Randomized controlled intervention study. Journal of the American Geriatrics Society. 2009;57(12): 2200-2208.
- 43. Dias A, Dewey ME, D'Souza J, Dhume R, Motghare DD, Shaji KS ... Patel V. The effectiveness of a home care program for supporting caregivers of persons with dementia in developing countries: a randomised controlled trial from Goa, India. *PloS one*. 2008;3(6): e2333.
- 44. Onor ML, Trevisiol M, Negro C, Alessandra S, Saina M, Aguglia E. Impact of a multimodal rehabilitative intervention on demented patients and their caregivers. *American Journal of Alzheimer's Disease and other Dementias*. 2007;22: 261–272.
- 45. Callahan CM, Boustani MA, Unverzagt FW, Austrom MG, Damush TM, Perkins AJ ... Hendrie HC. Effectiveness of collaborative care for older adults with Alzheimer disease in primary care: a randomized controlled trial. *JAMA*. 2006;295(18): 2148-2157.
- 46. Dröes RM, Meiland F. Schmitz M, Van Tilburg W. Effect of the Meeting Centres Support Program on informal carers of people with dementia: results from a multi-centre study. Aging & Mental Health. 2006;10: 112–124.
- 47. Onder G, Zanetti O, Giacobini E, Frisoni GB, Bartorelli L, Carbone G ... Bernabei R. Reality orientation therapy combined with cholinesterase inhibitors in Alzheimer's disease: randomised controlled trial. *The British Journal of Psychiatry*. 2005;187(5): 450-455.
- 48. Martin-Cook K, Davis BA, Hynan LS, Weiner MF. A randomized, controlled study of an Alzheimer's caregiver skills training program. *American Journal of Alzheimer's Disease and other* Dementias. 2005;20: 204–210.
- 49. Hepburn KW, Lewis M, Kirk LN, Narayan S, Center B, Bremer KL, Tornatore J. Partners in caregiving: A psychoeducation program affecting dementia family caregivers' distress and caregiving outlook. *Clinical Gerontologist*. (2006;29(1): 53-69.
- 50. Berger G, Bernhardt T, Schramm U, Müller R, Landsiedel-Anders S, Peters J... Frolich L. No effects of a combination of caregivers support group and memory training/music therapy in dementia patients from a memory clinic population. *International Journal of Geriatric Psychiatry*. 2004;19(3): 223-231.

- 51. Dröes RM, Breebaart E, Meiland FJ, Van Tilburg W, Mellenbergh GJ. Effect of Meeting Centres Support Program on feelings of competence of family carers and delay of institutionalization of people with dementia. *Aging & Mental Health.* 2004;8: 201–211.
- 52. Dröes RM, Meiland F, Schmitz M, Van Tilburg W. Effect of combined support for people with dementia and carers versus regular day care on behaviour and mood of persons with dementia: results from a multi-centre implementation study. *International Journal of Geriatric Psychiatry*. 2004;19: 673–684.
- 53. Dröes RM, Breebaart E, Ettema TP, Van Tilburg W, Mellenbergh GJ. Effect of integrated family support versus day care only on behavior and mood of patients with dementia. *International Psychogeriatrics*. 2000;12: 99–115.
- 54. Gitlin LN, Winter L, Corcoran M, Dennis MP, Schinfeld S, Hauck WW. Effects of the home environmental skill-building program on the caregiver–care recipient dyad: 6-month outcomes from the Philadelphia REACH initiative. *Gerontologist*. 2003;43(4): 532-546.
- 55. Gitlin LN, Corcoran M, Winter L, Boyce A, Hauck WW. A randomized, controlled trial of a home environmental intervention: effect on efficacy and upset in caregivers and on daily function of persons with dementia. *Gerontologist*, 2001;41: 4-14.
- 56. Teri L, Gibbons LE, McCurry SM, Logsdon RG, Buchner DM, Barlow WE ... Larson EB. Exercise plus behavioral management in patients with Alzheimer disease: a randomized controlled trial. *JAMA*. 2003;290(15): 2015-2022.
- 57. Romero B, Wenz M. Concept and effectiveness of a treatment program for patients with dementia and their relatives: Results from the Bad Aibling Alzheimer Disease Therapy Center (in German). Zeitschrift für Gerontology und Geriatrie. 2002;35:118–128.
- 58. Eloniemi-Sulkava U, Notkola IL, Hentinen M, Kivela SL, Sivenius J, Sulkava R. Effects of supporting community-living demented patients and their caregivers: a randomized trial. *Journal of the American Geriatrics Society*. 2001;49: 1282-7.
- 59. Quayhagen MP, Quayhagen M. Testing of a cognitive stimulation intervention for dementia caregiving dyads. *Neuropsychological Rehabilitation*. 2001;11: 319–332.
- 60. Chu P, Edwards J, Levin R, Thomson J. The use of clinical case management for early stage Alzheimer's patients and their families. *American Journal of Alzheimer's Disease and other Dementias*. 2000;15: 284-290.
- 61. Aupperle PM, Coyne AC. Primary vs subspecialty care: a structured follow-up of dementia patients and their caregivers. *American Journal of Geriatric Psychiatry*. 2000;8:167–170.
- 62. Ostwald SK, Hepburn KW, Caron W, Burns T, Mantell R. Reducing caregiver burden: a randomized psychoeducational intervention for caregivers of persons with dementia. *Gerontologist.* 1999;39: 299-309.
- 63. Logiudice D, Waltrowicz W, Brown K, Burrows C, Ames D, Flicker L. Do memory clinics improve the quality of life of carers? A randomized pilot trial. *International Journal of Geriatric Psychiatry*. 1999;14: 626–632.

- 64. Miller R, Newcomer R, Fox P. Effects of the Medicare Alzheimer's Disease Demonstration on nursing home entry. *Health Services Research*. 1999;34: 691-714.
- 65. Newcomer R, Miller R, Clay T, Fox P. Effects of the Medicare Alzheimer's Disease Demonstration on Caregiver Burden & Depression. *Health Services Research*. 1999;34: 669-689.
- 66. Yordi C, Dunah R, Bostrom A, Fox P, Wilkinson A, Newcomer R. Caregiver supports: outcomes from the medicare Alzheimer's disease demonstration. *Health Care Financing Review*. 1997;19: 97–117.
- 67. Moniz-Cook E, Agar S, Gibson G, Win T & Wang M. A preliminary study of the effects of early intervention with people with dementia and their families in a memory clinic. *Aging & Mental Health.* 1998;2: 199–211.
- 68. Riordan JM, Bennett AV. An evaluation of an augmented domiciliary service to older people with dementia and their carers. *Aging & Mental Health*. 1998;2:137–143.
- 69. Teri L, Logsdon RG, Uomoto J, McCurry SM. Behavioral treatment of depression in dementia patients: a controlled clinical trial. *Journals of Gerontology Series B: Psychological Sciences and Social Sciences*. 1997;52: 159–P166.
- 70. Brodaty H, Gresham M, Luscombe G. The Prince Henry Hospital dementia caregivers' training programme. International Journal of Geriatric Psychiatry. 1997;12: 183-92.
- 71. Brodaty H, Gresham M. Effect of a training programme to reduce stress in carers of patients with dementia. *British Medical Journal*. 1989;299: 1375-9.
- 72. Hincliffe AC, Hyman IL, Blizard B, Livingston G. Behavioural complications of dementia—can they be treated? International Journal of Geriatric Psychiatry. 1995;10: 839–847.
- 73. Vernooij-Dassen M, Huygen F, Felling A, Persoon J. Home care for dementia patients. *Journal* of the American Geriatrics Society. 1995;43: 456–457.
- 74. Vernooij-Dassen M. Dementie en thuiszorg: een onderzoek naar determinanten van het competentiegevoel van centrale verzorgers en het effect van professionele interventie [Dementia and Homecare: A Study into Determinants of The sense of Competence of Caregivers and the Effect of Professional Intervention]. Amsterdam: Swets & Zeitlinger, 1993.



CHAPTER 3 -

How do activating interventions fit the personal needs, characteristics and preferences of people with dementia living in the community and their informal caregivers?

Published as:

Van't Leven N, de Lange J, Prick AE, Pot AM. How do activating interventions fit the personal needs, characteristics and preferences of people with dementia living in the community and their informal caregivers? *Dementia: The International Journal of Social Research and Practice*. (London). 2019; 18(1), 157-177. doi: 10.1177/1471301216662378, 0(0) 1–21. Epub 2016 Aug 10.

Abstract

Background

Psychosocial interventions aim to mitigate the serious consequences of dementia for the daily life of people with dementia and their informal caregivers. To deliver a person-centred approach, it is crucial to take needs, characteristics and preferences of people with dementia and their informal caregivers into account. However, these factors are generally not systematically checked in order to determine which intervention will be most appropriate. Additionally, little is known about which intervention suits which needs, characteristics and preferences. Therefore, this study examined how three multiple-component, activating dyadic interventions fitted needs, characteristics, and preferences of both the people with dementia and their informal caregivers: the Pleasant Events Program, the Exercise and Support Intervention for People with Dementia and Their Caregivers, and Occupational Therapy.

Methods

Semi-structured interviews were held with participants in either one of the interventions, 34 dyads and 19 professionals. The constant comparative method was used for the analysis.

Results

Five factors influenced the dyad's 'fit': timing, need for activity, lifestyle, apart-ortogether and meaning of (lost) activity. The factors 'timing' and a 'need for activity' were conditional for these activating interventions. Dyads in an early stage of dementia, who were aware of the effects on daily life, were open to a change in routine, and had a need to maintain activities profited from these interventions. Three distinctive factors were important for the fit of one of the three interventions in particular: 'lifestyle', 'apart or together' and 'meaning of (lost) activity'. The Pleasant Events program and the Exercise and Support intervention properly addressed the need for activities that afforded daily pastimes or structure. The Exercise and Support Intervention addressed the need for physical activity and emphasized shared activity. Occupational Therapy properly addressed the need for self-sufficiency, maintaining activities and adjustment to physical limitations.

Conclusion

The five identified factors can contribute to a more person-centred application of the interventions.

Introduction

Dementia reduces the quality of life of people with dementia and their informal caregivers.¹⁻³ Psychosocial interventions aim to mitigate the results of this process. The importance of a person-centred approach of these interventions has been increasingly recognised. Person-centred care was introduced almost two decades ago by Kitwood who emphasized the importance of addressing the psychological needs of people with dementia and the maintenance of their personhood.⁴ Understanding the history and individual needs, characteristics and preferences of people with dementia is crucial for a person-centred approach.⁵ Unfortunately, when interventions are offered to people with dementia in clinical practice or research trials, their needs, characteristics and preferences are generally not systematically checked to determine which intervention will be the most appropriate one. Insight into the applicability, or 'fit', of an intervention with the needs, characteristics, and preferences may help to apply interventions in a more person-centred manner. Fit indicates the extent to which an intervention fulfils individual needs and is attuned to characteristics and preferences.

In recent years, psychosocial interventions have increasingly focused on 'dyads', consisting of both the person with dementia and the primary informal caregiver, often a spouse or child. This was the result of an increasing awareness that the functioning of people with dementia was related to the functioning of their primary informal caregivers.^{6,7} Dyadic interventions take this mutual influence between the person with dementia and the informal caregiver into account. The effects of dyadic psychosocial intervention programs have been studied in previous reviews.^{8,9} Results indicate the beneficial potential of these interventions for maintaining independency in activities of daily living and competence for caregivers, contributing to a better quality of life.

This study is focused on three dyadic, multi-component interventions, to support the dyad in recognizing which activities are still possible to engage in, despite the challenges posed by dementia. Those were: the Pleasant Events Program¹⁰, the Exercise and Support Intervention for People with Dementia Living at Home and their Caregivers¹¹, and Occupational Therapy according to the 'Community Occupational Therapy in Dementia Guideline' (COTiD).¹² These interventions make use of various activities, offer psycho-education, and provide emotional care that help the dyad adapt to the effects of dementia in their daily lives. The person with dementia and the informal caregiver practise these activities together. The interventions have a basic structure to start with and are adjusted to the dyad's wishes and needs. All three are short-term interventions consisting of six to ten home visits. They differ in the nature of their activities. In the next paragraph we describe each of the three interventions in more detail. For additional details of the interventions, see Table 1.

	Pleasant Events Program	Exercise and Support Intervention for People with Dementia and their Caregivers	Community Occupational Therapy in Dementia (COTiD)
Original interventions	This program is based on Teri et al. ¹³ The Dutch manual for the program was published by Nivel [Dutch Institute for Primary Healthcare] ¹⁰ , and is supplemented by examples from Gitlin et al. 's program. ¹⁴	This intervention is based on Teri et al. ¹⁵ After a pilot study with the original program, this program was adapted. Attributes like a ball, weights, and elastic equipment made the exercises more attractive for people with dementia. The amount of sessions was limited to eight. The training for activating events, beliefs, and consequences (ABC) was skipped and this program emphasized the pleasant events component more. ¹¹	This therapy is described in Graff et al. ^{12,16,17}
Primary goal	Maintaining activity, having pleasant activities, and preventing depression of people with dementia and the informal caregiver.	Preventing depression of people with dementia and the informal caregiver.	Improving the performance of daily activities of people with dementia, and mitigating the care burden of the informal caregiver.
Components	 Choosing and planning pleasant activities for the person with dementia, the informal caregiver or both, Practicing these activities, Adjusting activities to the capabilities of the person with dementia, Psycho-education, Homework: pleasant activities. 	 Exercises for flexibility, balance, power, and stamina of the person with dementia and the informal caregiver, supervised by a coach at home. Identifying pleasant activities for both, Psycho-education and communication training to manage behavioral problems, Homework: repeating mobility exercises and pleasant activities (preferred at least 3 times a week). 	 Needs assessment and observation, Practicing meaningful, daily activities (self sufficiency) with compen- sation strategies and adjustments for the person with dementia, Psycho-education, Practicing different ways of approaching the person with dementia for the informal caregiver, Homework: individual appointments for practicing activities and approach.

>>

Table 1 Interventions: content, evidence and provision

	Pleasant Events Program	Exercise and Support Intervention for People with Dementia and their Caregivers	Community Occupational Therapy in Dementia (COTID)
Duration	A maximum of 6 home visits of 1,5 hours each.	A maximum of 8 home visits of 1 hour each.	A maximum of 10 home visits of 1 hour each.
Evidence	Positive effects on depression for people with dementia and informal caregivers and on occupatio- nal engagement and agitated behavior. ^{13,14}	Positive effects on depression and general health of people with dementia. ¹⁵ An RCT with the adjusted program found only a quantitative effect on cognition. ¹⁸	Positive effects on daily functioning of people with dementia and competence of informal caregivers, as well as on depression, general health and quality of life of both. ^{17,19} These effects were not demonstrated in studies by Voigt-Radloff, Graff, Leonhart, Schornstein, et al. and Döpp et al. ^{20,21}
Provision	The program was offered as part of the University Collaborative Centre for Dementia, which entails the collaboration of Rotterdam University, the Home Care Organization 'De Zellingen' and the Van Kleef Institute (for generating and disseminating knowledge for home care professionals). The coaches were students graduating in healthcare studies and home-care workers from De Zellingen. They were supervised during the intervention. The project leader and case manager in the region recruited participants.	The program was offered within an RCT from the Department of Clinical Psychology, VU University, Amsterdam. ^{11,15} The coaches were Master students from the Depart- ment of Clinical Psychology, VU University, who completed a special training program for geropsychology. They were supervised during the project. The project leader recruited participants for the program with the help of caregiver organizations, local Alzheimer cafes (public meetings for people with dementia, their caregivers, and others) and case managers throughout the Netherlands.	Occupational Therapy is regularly offered and delivered by certified occupational therapists, who followed an additional training for this program. The participants were referred by medical doctors and geriatricians as well as case managers throughout the Netherlands.

RCT = Randomized Controlled Trial

>>

The Pleasant Events Program provided support in adapting and planning pleasant activities, e.g. learning to use skype with children and grandchildren with a stepwise manual, occasionally taking public transport to a historic city part when going shopping instead of the usual shops, watching national geographic on television or having a walk in the neighbourhood. The Exercise and Support Intervention contained exercises for flexibility, balance, strength and/or endurance

61

exercises using a ball, weights and elastic bands and combined this physical exercise with support for planning pleasant activities, as well as psycho-education and communication skill-training. Occupational Therapy consisted of the improvement of self-care in a broad spectrum, such as learning to use the remote control for the television with orientation marks, making a bicycle-tour that is worry-free for the informal caregiver at home by taking a well-known route and using a mobile phone, helping to keep a daily structure and schedule with an individually adapted agenda.

The three dyadic, multi-component interventions described above have been proven effective in scientific research.^{13-15,17,19} However, recent studies could not confirm the original results.^{18,20,21} This might be due to the adaptation or implementation of the interventions or to the different characteristics of the settings in which they were re-tested.²¹⁻²⁴ Process evaluations of these trials showed positive experiences of participants with these intervention.^{22,23,25} Finding no significant results of interventions might also be a result of not fitting the needs, characteristics and preferences of people with dementia. When choosing an intervention, 'Which intervention works?' is not the only question, but also 'What works, for whom, and at what times?^{9,26-28} Therefore our research question for this study was: which factors explain the appropriateness or fit of the three interventions with the needs, characteristics, and preferences of the participating dyads?

Method

We used a qualitative design with semi-structured interviews to gather in-depth information about the fit of these interventions in relation to the needs, characteristics and preferences of the dyads.²⁹⁻³² We wanted to integrate the perspectives of people with dementia, their informal caregivers, and the professionals who guided them through the intervention, and we included and studied them as cases.

The Pleasant Events Program and the Exercise and Support Intervention were both evaluated in a larger study (see Table 1). Ethical approval for the interviews in our study was already included in the study designs (Pleasant Events Program: Medical Ethics Review Committee of the Erasmus Medical Center, number 2009-117; Exercise and Support Intervention: Medical Ethics Review Committee of the VU University Medical Center, number 2008/320). For the interviews about the Occupational Therapy Intervention ethical approval was attained separately (Medical-Ethical Review Committee for Mental Health Care, number 11.123).

Participants

We recruited participants who had either taken part in the Pleasant Events project, the trial for the Exercise and Support Intervention, both already research participants, or had Occupational Therapy according to the COTiD-guideline as a clinical service. The participants had been recruited for or referred to that particular intervention (they were not offered a choice of which intervention they thought would best suit them). Participants for the Pleasant Events Program and the Exercise and Support Intervention were recruited via the project managers, coordinating those studies. Participants for occupational therapy were recruited via occupational therapists. The project managers asked participants in both studies, after the intervention had ended, if they were open for an interview with an independent researcher. The occupational therapists asked their clients after the intervention if they were open for an interview too. They had leaflets with information about the study and the interviewer available for a dyad. When participants were positive to an interview they asked permission to share their name and telephone number with the interviewers. After receiving names, the researcher contacted the dyad per telephone, explaining again the goal of the interview and the study. If the dyad was still open for an interview, an appointment was made. This appointment was confirmed by post with an informative letter and a form for informed consent for the interview and audio-recording was already sent. Four dyads still cancelled their interviews, because of illness or being otherwise engaged. At the time of the interview at home the information was repeated and the informed consent was signed. We used one form with both names of the dyad. The people with dementia signed if they wanted and it did not confuse them. The informal caregiver also signed. If the people with dementia could not sign, the informal caregiver signed for both of them as a couple.

Our intention was to obtain a variety in cases to include a broad spectrum of experiences, needs and characteristics in the data. So we asked them for both successful and less successful cases and also those that had not finished the program. In this way convenient samples were composed for each intervention.

Data collection

We aimed to interview both members of the dyad, either together or separately, depending on their preference, so that they would feel at ease. Seven informal caregivers indicated that their partner or parent with dementia could no longer reflect on their experiences and a joint interview would be too burdensome for them. These informal caregivers agreed to an interview on a day that the person with dementia was in day care. The interviews with people with dementia and informal caregivers took place in the participants' homes. An interview with a couple started with an open question about the interviewees' experiences with the intervention. Follow-up questions addressed initial expectations, continuation of exercises and activities, usefulness of the advice, what the couple had gained from the intervention, and what was still missing. Especially in the interviews with people with dementia, the interviewers used short sentences and prompts, and tried to follow their reasoning through, repeated parts of their answers to stick to their story, and sometimes presented a photo of the professional. They spent sufficient time with small talk between questions to let the person with dementia rest and followed their track of thought.^{33,34} We did not mention the term dementia or Alzheimer unless the person with dementia or caregiver had first used this term. Some caregivers added their perspective afterwards. The interviews lasted 50 to 120 minutes. The first and third authors (NL, AEP) conducted the interviews. In case the need for information or a helpful conversation emerged, we had leaflets for the dyads for telephone support, a service of the national Alzheimer's Association, where they could talk with peers.

The interview with the professional took place after the interviews with the dyad, at their workplace or at a public space. Some professionals were interviewed about their experiences with several dyads during one appointment, but each dyad was discussed separately. This interview started with an open question about the professional's experience of the intervention related to the dyad. Follow-up questions addressed what results were achieved, how the professional tailored the intervention to the needs and preferences of the dyad, and what was not achieved. These interviews lasted 20 to 30 minutes. The recordings of the interviews were transcribed verbatim and imported into Atlas-ti-6/7.1 for qualitative analysis.³⁵

Analysis

Data collection and analysis was an iterative and reflexive process, based on the constant comparative method.^{32,36,37} An interview with a person with dementia and a caregiver was linked to the interview with the professional, and was included



PE=Pleasant Events program, EP=Exercise program, OT=Occupational Therapy program PwD= Person with Dementia, CG=Caregiver

Figure 1 Cases and analysis process

as a case (Figure 1). Transcripts were read to capture an overall impression and a preliminary summary was made with the perspectives of the person with dementia, caregiver and professional, similarities and differences. The transcripts were unravelled with open codes, to describe all aspects of the content.²⁹ The analysis consisted of three phases in a back and forth process. In the first phase, eleven cases were analysed exhaustively with open coding. Some codes referring to the same phenomenon were already grouped together to include convergence and divergence. We summarized the degree of fit for each case. In the second phase, we grouped the codes into categories under the themes: needs, personal characteristics, contributing elements, perceived results, and fit. NL coded 10 new cases and recoded the 11 earlier cases. We extracted within these themes what kind of needs and which characteristics were important for the fit of the intervention. This led to an analytical framework with preliminary factors that affected the fit. In the third phase, we used the remaining 13 cases to test this analytical framework and to search for new perspectives. No new codes were needed, which showed that saturation had been reached. The preliminary factors were condensed to five factors that were important for the fit of the interventions for the participants.

Methods to enhance credibility in the analysis

Three researchers (NL, JL, AEP) independently coded the interviews of four cases. They discussed any coding differences. NL analysed all data in detail. NL and JL frequently discussed results during all phases for plausibility and consistency. Several independent researchers conducted peer reviews on preliminary results. We offered every respondent at the end of the interview the possibility to read the transcript or receive a copy of the article at the time of publishing. All authors discussed methods, summaries, analyses and results in the study process. Table 2 shows details of the participating couples and professionals to conclude on the degree of transferability.

Findings

Characteristics of the participants

We included a total of 34 cases, although some cases missed the interview of the person with dementia, and some the interview of the coach (Table 2). The interviewees indicated that the onset of the dementia's symptoms had begun between 1 to 5 years ago at the time of the intervention, although the syndrome was often formally diagnosed later. The Pleasant Events Program included comparatively more participants with advanced dementia and poor communication ability, who needed assistance for personal care and attended day-care for several days. The Occupational Therapy intervention included more people with physical limitations, such as poor eyesight or difficulty with walking, than the other interventions. Sometimes physical limitations were the reason for the referral rather than cognitive limitations. The overall professional support for the participating dyads ranged from supervision by a geriatrician and case manager only, to day-care

Table 2 Participants

Program		Pleasant Events Program 9 Cases:	Exercise and Support Program 11 Cases:	Community Occupational Therapy 14 Cases:	34 Cases:
People with	Mean age (years),	78.1	75	80	
dementia	Sex	7 M/2W	6 M/5W	9 M/5W	
	Onset of dementia: Range (Mean)	1-4 Years (2.4)	1–5 Years (2.9)	1-4 Years (2,4)	
	Interview	6 PwD	10 PwD	11 PwD	27 PwD
Caregivers	Mean age (years),	Partners: 76	Partners: 72.6 Child: 53	Partners: 77.3 Children: 53	
	Sex	2 M/7 W	4 M/7 W	3 M/10W	
	Relation	9 x Partners	10 Partners 1 Parent/child	9 Partners 5 Parent/child	34 CG
Coaches/ Professionals	Background	3 Pairs of 4-year students; 2 Homecare workers, 9 cases	2 Master students clinical psychology, 5 cases	9 Occupational therapists, 14 cases	19 Coaches/ professional, involved in 28 Cases
	Dyads per professional	1or 2 Dyads per pair; 2or 3 Dyads per homecare worker	2 or 3 Dyads per coach	1, 2 or 5 Dyads per OT	

PwD = Person with dementia; CG = primary informal caregiver, M= Men, W= Women

and personal care several times a week. Most of the participating dyads were partners, others parent-child. The majority of the people with dementia were men in all three interventions.

Five factors for fit of the interventions

The analysis of the interviews led to five client-related factors that affected the fit of the interventions to needs, characteristics, and preferences of a dyad: two conditional factors, 'timing' and 'need for activity', and three distinctive factors, 'lifestyle', 'apart or together', and 'meaning of (lost) activities' (Table 3).

Timing

Most interviewees who took part in the intervention when the dementia was in an early stage claimed that it had helped them continue their activities. The professionals confirmed this in their interviews. Those who could accept change in order to cope with the consequences of the dementia process and were willing to put suggestions into practice benefitted considerably from the intervention. Some people with dementia said they wanted to slow down the dementia process as much as possible. They felt responsible for their health. The intervention offered them the opportunity to help combat the onset of the dementia disease. One man with dementia explained:

It's simply a nasty disease. We have to face many problems. But my wife gets the most of it. And I simply can't help all the time... The good news is that there are some small stepping stones (such as this intervention, the Exercise and Support Intervention), where I can make an effort 'to slow the dementia down'. I do hope so.

We concluded that participating in the intervention and perform the exercises were meaningful for him. It supports him in coping with his illness. Many informal caregivers stated that they had already actively sought information or other help. One informal caregiver said:

I believe that the more tips and help and guidelines you can obtain for dementia, the better it is for everybody. I have often seen informal caregivers in my network who have the notion that they have to be able to do it all on their own, because accepting help is a sign of weakness. I've never seen it that way myself.

Some participants in the Pleasant Events Program and the Exercise and Support Intervention did not want to emphasize 'problems' they needed help with, but were very positive about the intervention. They perceived it as a course in staying active and learning about living with dementia. For them, the intervention had been offered well on time.

Many professionals mentioned that openness towards the intervention was related to addressing the real needs of a dyad. It was quite a challenge for them because a question often illustrated a deeper need. When they managed to make this deeper need explicit, most dyads were open and willingly to get involved in the intervention. Most of the professionals stated that the intervention should have been available for their clients at an earlier stage. An occupational therapist explained:

Ideally, we would like to get involved when people with dementia can still learn a bit and we can anticipate future problems. For example, it can be worthwhile to install a telephone with photo buttons in a fixed location when they can still make a telephone call themselves. This makes it easier for them to operate. The same is true for the timely installation of a date clock. We're often brought in too late. My colleagues and I consider cases where the family needed to 'turn off the gas for safety' the most difficult. This means you are a little too late. Perhaps in an earlier phase we could have helped the person find a way to compensate.

The intervention was deemed too late if a person with dementia could no longer absorb new information or could no longer retain focus. The intervention was also too late if the informal caregiver was overburdened and lacked the energy to attempt new behaviour, which was particularly important for these activating interventions. These informal caregivers did feel the professional supported them, but they had benefitted little from the intervention.

In contrast, the intervention came too early for some dyads. One dyad needed more time to acknowledge the diagnosis and primarily needed information at that point in time. Another dyad wished to retain their normal way of life for as long as possible, and was not yet open to the advice. Dyads that were less able to change had fewer results. These interventions, that mobilised them to adapt to the effects of dementia in their daily lives, called for the adjustment of their existing habits. Some found that the activities and advice did not apply to their situation, despite the professional's attempts to motivate them. An informal caregiver reported that the advice was not very useful for her because she just wanted her husband to do the activities the way he always did them:

I was advised to ask him directly for activity because of his lack of initiative. I did try this. He helps dry the dishes, but puts everything on the counter, right, rather than where it's supposed to go... That makes me think, "Well, I'll just do it myself".

Need for activity

The interventions addressed the dyads' need to maintain activities. People with dementia often mentioned activity in general, such as 'doing what I was used to doing', or 'keeping a hold of things'. Some caregivers explained that the people they cared for had difficulty with various activities: hobbies, daily tasks or getting out. CGs also had general questions about 'how to support the person with dementia best to perform activities'. One informal caregiver spoke about the problems she encountered:

Recently I asked you "Could you make a cup of tea?" Then you did your utmost best and came back with two plates of strawberries. And then I repeated, "You need to make a cup of tea, dear." And then, you came back with... another bowl of strawberries. The third one! And usually, you've never had a problem doing something, and now, when I ask you for something, you just can't figure it out.

This caregiver required help in helping her husband complete activities by himself, as many informal caregivers did. The informal caregivers needed advice for their new role of supporting the person with dementia in carrying out activities. The problems included 'Which instructions are required?' and 'Should I take charge or let him/her figure it out and do it alone?' Many informal caregivers also mentioned their difficulty with the apathy of the person with dementia: 'He just sits on the couch'. They needed advice about how to motivate the person with dementia and how to adapt activities to their limitations.

All professionals mentioned efforts to explore the latent needs of a dyad for activities.

There were also a few dyads who did not experience a need for activities, and for them, the intervention had less impact. An informal caregiver reflected:

Yes, we are very busy. We have a pretty large family. We are both married for the second time. So we've got four families... And the kids, of course. Plenty of work to go around... She (the partner with dementia) is still pretty present. We go everywhere. So we said to each other: "Actually, we do not think this (the intervention) is for us." We are not far enough along yet.

Lifestyle

Lifestyle was shown to be a factor for the fit of the interventions. In this context, lifestyle means activities and habits. Some dyads explained how they usually spent their time, both in the past and in the present. This gave us insight into their interests and what was important to them. Expressions such as 'Well, you know, staying active, that's important...' or 'we like to make an outing regularly, visit a museum for example' or 'My wife and I have always taken part in sports' or: 'He's a real outdoor man' show their activity preferences. Many dyads shared values such as 'keeping active', 'staying mobile', or 'getting out of the house every day'. Some dyads shared interests, whereas others had differing interests. Either way, the professional could adapt the intervention to address different needs. At least one of them should have an active lifestyle for the fit of these interventions. It appeared that the caregiver tried to facilitate the person with dementia in continuing his or

her usual activities, but also tried to activate the person with dementia with activities the caregiver liked to do him- or herself.

Table 3	actors affecting how the interventions fit the needs and characteristics of the					
	dyad					

	Fit-factors	Pleasant Events Program	Exercise and Support Intervention for People with Dementia Living at Home and Their Caregivers	Occupational Therapy According to the COTiD Guideline					
Conditional factors	Timing	Early phase of dementia Openness to change Accepts change to cope with consequences of dementia							
	Need for activity	Maintain activity Training of lost activities Support for caregiver to instruct the PwD to carry out activities							
Distinctive Factors	Lifestyle, For both or for one of them	Active Link to: all kinds of activity at home or outside; conviviality	Active Link to: exercise, sports, walking, etc.; physically active	Active Link to: all kinds of daily activities at home or outside; "do what you can yourself"					
	Apart or together	Apart or together	Together	Apart or together					
	Meaning of (lost) activities	Need for maintaining activity, daily pastimes, structure	Need for training and physical activity, daily pastimes, structure	Need for doing things oneself and self-sufficiency, adjustment to physical limitations, daily pastimes, structure					

Apart or together

The interventions seemed to satisfy the need to do something together very well. Many of the people with dementia enjoyed spending time with their informal caregivers; they felt at ease. The informal caregivers expressed this need another way. They often spent a lot of time caring for the person with dementia, but missed the contact and mutuality they had before. Particularly, informal caregivers of dyads who were used to doing a lot together were still looking for moments to share their experiences. The intervention helped them to do things together and to get a feeling of togetherness again. Some informal caregiver s had no need for additional joined activities. The time spent on providing care was enough. For
them, it was especially important to keep the person with dementia busy for a while or do the exercises on their own so they could attend to their own activities. The professionals adjusted the intervention to these needs of a dyad.

All three interventions required input from both the person with dementia and the informal caregiver, and they required some togetherness and cooperative action. If there was a limited need to do things together the Occupational Therapy and the Pleasant Events Program was more applicable than the Exercise and Support Intervention. The Exercise and Support Intervention requires participants to do the exercises together three times a week and to plan pleasant activities. Some of the dyads particularly appreciated having an additional shared activity. For them, the intervention fulfilled the need to do something together. One professional said: 'They were keen on doing something together, being active, so the exercise program was a good offer for them'. For other dyads, practicing three times a week was a burden on the informal caregiver and did not fit their needs for togetherness and independence. The Pleasant Events Program is oriented on fun pastimes and Occupational Therapy on the ability of people with dementia to do activities more independently. Depending on their needs and abilities, a dyad can choose to do more or less together, and the therapist can choose to be more or less oriented on joint activities.

Meaning of (lost) activity

The result of the activity itself, such as making a cup of coffee, or going outside for a walk was important for a dyad, of course. However, the corresponding meaning of these activities for a person with dementia or the informal caregiver determined the degree of feeling a loss of activities. Making a cup of coffee could mean independency but could also mean caring for the partner, while going for a walk could mean just passing time but could also mean being physically active or having social contact. The activity could mean a pastime, maintaining structure, being physically active, having social contact, being self-sufficient, or may add to satisfaction and self-appreciation. For example, for one lady with dementia who lived by herself, the lack of activities meant a lack of structure in the day. She was facilitated in performing some activities again through memory-aids and a telephone call from her daughter. The rhythm and structure in the day was enough for her to also pick up other activities like making coffee and making bread, so she could maintain living on her own. If the lost activities primarily were related to the meaning of the activity, couples often did not specifically call it loss of an activity. For example, the dementia patients said they had nothing to do anymore, or caregivers explained

they felt powerless to handle the situation. For others the result of the activity itself was important. In those cases, people tended to name specific activities during which they experienced a lack of self-sufficiency: hobbies, household activities, getting dressed, or operating devices in the house. One woman with dementia said:

I like watching television. I tend to turn it on for a bit in the morning, and then at the end of the afternoon. Well, that didn't go well any more. Her daughter added: My mother had another device with a remote control, and she had a remote control for the television. She couldn't keep them apart.

Some of the participants also suffered from physical limitations, apart from the cognitive or behavioural problems caused by the dementia. This made it difficult for them to maintain the same level of activity. They needed to adjust their standard activities to both their physical and cognitive limitations.

Discussion

Main findings and reflections

This study presented five factors that reflect the fit of the three studied interventions to the dyads' needs, characteristics and preferences. Timing and a need for activities are both conditional factors for the fit of any of these activating multiple-component interventions with dyads' needs and characteristics. The factors lifestyle, the need to do something together or apart, and the individual meaning of activities are indications of the fit of one of the three interventions in particular. The Pleasant Events Program was well suited to dyads that enjoyed various activities at home or outside, together, but also apart, for pleasure and daily pastimes. The Exercise and Support Intervention was appropriate for dyads of whom one of the two preferred exercise or sports, were open to exercise together a few times a week, and had a need for daily pastimes and keeping active. Occupational Therapy was well suited to dyads for which self-sufficiency of the person with dementia was the primary goal, in shared as well as individual activities. Occupational Therapy also took into account the need of advice about activities for dyads with physical limitations.

In clinical practice and in inclusion for trials these factors are often not taken into account. Characteristics such as age, level of education, social economic status and

social network may be collected, but it is not common to discuss preferences, habits and values systematically with a dyad. In this study we also collected data about socio-economic status, level of education, (former) job and religion of the participating dyads, but these data did not reveal factors for the fit of the interventions. The five factors found in this study are all related to preferences (although lifestyle is related to educational level and socio-economic status too).

Clinicians need to take a person-centred approach by carefully considering a dyad's needs, characteristics, and preferences to determine the intervention that is likely to be the best fit. Based on the results of this study, a two-step process may be advised. The first step, for professionals like physicians and case-managers who would like to refer people with dementia and their caregivers to an activating intervention, is about assessing the needs of a dyad before referral to a specific intervention. Clinicians need to be alert on the dyad's openness or readiness for change and ask what their need for activities is. The 'timing' requires careful monitoring of a dyad, because their needs for support are not static and change due to the progress of dementia. In a second step, a specific activating, multicomponent intervention needs to be selected. Clinicians need to ask both the person with dementia and the caregiver whether they prefer to do things together or apart, what there life-style has been (active, physically active, doing sports, outings) and the meaning of activity for them conviviality, pastime, selfsufficiency). This might be a challenge for clinicians, because although some dyads ask for advice or support in a direct way, often help is needed to make their latent needs explicit.³⁸

The three interventions aim at maintaining the functional abilities of people with dementia, supporting the caregiver, and helping both to cope with dementia progression. The preventive character of these programs, which are limited in both duration and cost, makes them suitable for all dyads who recognize difficulty with activities. Emphasising possibilities despite apparent limitations may strengthen a dyad, as well as attention to maintaining activity; but this has not been made a part of standard care yet. Some dyads who participated in the Exercise and Support Intervention did not want to talk about 'problems'. They stated that the intervention trained them to cope with their situation by improving their capacities instead of emphasising problems. In line with Haberstroh and colleagues the term 'support' was associated with helplessness, whereas the term 'training' was associated with maintaining capacities and empowering.³⁹ This finding suggests offering these interventions to dyads as training to improve their skills in dealing with the effects

of dementia, as to distinguish this type of support from other support such as respite care. Such an approach may stimulate openness to an intervention.

Strengths and limitations of the study

The dyads involved in this study may not be representative for the total group of people with dementia and their informal caregivers who participated in the three interventions, although the respondents did represent a group of people of diverse age, education, work, and dementia duration. Both the dyads and the professionals in this study were willing to be interviewed because they wanted to contribute to good care for people with dementia. Furthermore, only a few of them dropped out during the intervention. In recruitment of dyads for this study we asked specifically for dyads and professionals with positive and negative experiences. We included five dyads (3 Pleasant Events Program, 2 Occupational Therapy) who ended the intervention after one or two home visits. The caregiver was overburdened, there was no need for activity (not yet or not anymore) or there were personal problems. So most dyads found enough benefit of the intervention to continue to apply it to at least some degree. We recommend that future studies interview more dyads who chose not to take part or dropped out early, because they can afford more insight into the fit of interventions.

One particular strength of the study is the involvement of people with dementia. A significant number of interviewees with dementia related their intervention experiences in the interview. Some were no longer able to reflect on the intervention, but they could still indicate what activities were important to them and why. The period between the completion of the intervention and the interview varied from 1 to 10 months. Many people with dementia who were able to participate in an interview could still talk about the intervention or the professional, even after 6 or 9 months, which enabled us to include experiences over a longer time.

It is unclear to what extent a joint interview versus separate interviews affected the study. The informal caregivers who were present frequently added to accounts of the people with dementia and helped them express themselves. However, the people with dementia or the informal caregivers may have felt difficulty expressing themselves freely in joint interviews. Some informal caregivers resolved this by speaking to the interviewer alone after the interview to freely express their opinion. In some of the cases we realized at the time of (re)reading the transcript that the content of what the caregiver said was not supportive for the person with dementia's mood, although the atmosphere and non-verbal signs had been positive. Based on this and other experience we would prefer to interview both members of the dyad separately in future situations, perhaps performing three interviews, both separately and jointly.²³

Implications for practice

In clinical practice, programs with treatment components that are more or less comparable to the studied interventions are becoming increasingly available. The experiences of participants in this study address the importance for researchers and clinicians to take a person-centered approach and assess the needs, preferences, and characteristics of dyads first, both separately and jointly, to decide through shared decision-making which intervention would best suit these dyads. Referring professionals may use the two conditional factors to discuss with a dyad the appropriateness of any of these activating interventions.

Timing concerns questions like:

- is the dyad still overwhelmed by the diagnosis? Then referral to an activating intervention may come too early.
- do both persons have enough energy to participate in an activating intervention? For example, when the CG is overburdened, an activating intervention may be too late.

Need for activity may hold questions like:

- does the person with dementia need to maintain activities or pick them up again? Is he or she able to do more activities?
- does the caregiver need support in assisting or instructing the person with dementia to perform activities?
- does the caregiver need more insight into the capacities of the person with dementia, what he or she is able to do?

The other factors could help to choose an intervention in particular.

Lifestyle holds questions like:

- does the dyad or one of them (person with dementia or caregiver) have, or did he have an active lifestyle?
- what kind of activity he or she used to do: physical activity, sports, days outings?

Apart-together concerns questions like:

- is the person with dementia accustomed to spending time alone?
- does the person with dementia depend on the caregiver?
- does the person with dementia and/or the caregiver like to have shared activities?
- does the caregiver need time for him-/herself?

Meaning of (lost) activity concerns questions like:

- is there a need for more daily routine in the activities of the person with dementia?
- are lost activities related to a lost pastime or self-sufficiency for the person with dementia?
- is there a need for more safety inside or outside the house in order to be able to continue activities?
- is there a need for adaptations of activities to the physical limitations of the person with dementia in order to be able to continue these activities?

Conclusions

Dyads in an early stage of dementia, who were open to change their habits and routines because of the consequences of the dementia process and had a need to maintain activity, especially profited of these interventions. The intervention Pleasant Events as well as the intervention Exercise and Support properly addressed the need for daily pastimes, structure, and fulfilment. The Exercise and Support Intervention addressed the need for physical activity and emphasised shared activity of the dyad as well. Occupational Therapy (COTiD) properly addressed the need for additional self-sufficiency, maintaining activities and adjustment to physical limitations.

Further study of the validity and feasibility of these factors is needed before they can be used for choosing the appropriate intervention. With the results of this study we are able to formulate hypothetic indications for these activating interventions directed at maintaining daily activity and coping with changing roles and diminishing capacities. Consequently, it would be possible to design a tool for the needs assessment of the activities for a dyad.

References

- 1. Pinquart M, Sorensen S. Helping caregivers of persons with dementia: which interventions work and how large are their effects? *International Psychogeriatrics*. 2006;18(4):577-595.
- von Kutzleben M, Schmid W, Halek M, Holle B, Bartholomeyczik S. Community-dwelling persons with dementia: what do they need? What do they demand? What do they do? A systematic review on the subjective experiences of persons with dementia. Aging & Mental Health. 2012;16(3):378-390.
- 3. Zwaanswijk M, Peeters JM, van Beek AP, Meerveld JH, Francke AL. Informal caregivers of people with dementia: problems, needs and support in the initial stage and in subsequent stages of dementia: a questionnaire survey. *Open Nursing Journal*. 2013;7:6-13.
- 4. Kitwood TM. Dementia Reconsidered: The Person Comes First Buckingham UK: Open University Press, 1997.
- 5. Brooker D, Latham I. Person-Centred Dementia Care, Making Services Better with the VIPS Framework. Second Edition: ed. London: Jessica Kingsley Publishers, 2016.
- 6. Ablitt A, Jones GV, Muers J. Living with dementia: a systematic review of the influence of relationship factors. *Aging & Mental Health*. 2009;13(4):497-511.
- 7. De Vugt M, Stevens, F., Aalten, P. et al. Do caregiver management strategies influences patient behaviour in dementia? *International Journal of Geriatric Psychiatry* 2004;19:85-92.
- 8. Smits C, Lange J de, Dröes R, Meiland F, Vernooij-Dassen M, Pot A. Effects of combined intervention programmes for people with dementia living at home and their caregivers : a systematic review. *International Journal of Geriatric Psychiatry*. 2007;22(12):1181-1193.
- 9. Van't Leven N, Prick AE, Groenewoud JG, Roelofs PD, de Lange J, Pot AM. Dyadic interventions for community-dwelling people with dementia and their family caregivers: a systematic review. *International Psychogeriatrics*. 2013;25(10):1581-1603.
- Verkaik R, Francke, A, Lahr, M, Zuidervaart, M. . Huisbezoeken; Voorkom Somberheid bij Geheugenproblemen, handboek voor de hulpverlener (plezierige activiteitenmethode). [Home visits: Preventing depressive mood in memory problems, manual for the professional: Pleasant activities]. retrieved from https://www.nivel.nl/nl/publicatie/huisbezoeken-voorkom-somberheid-bij-geheugenproblemen-handboek-voor-hulpverlener-en. Utrecht: Nivel;2005.
- Prick AE, de Lange J, Scherder E, Pot AM. Home-based exercise and support programme for people with dementia and their caregivers: study protocol of a randomised controlled trial. BMC Public Health. 2011;11:894.
- Graff M, Melick, M.van, Thijssen, M., Verstraten, P., Zajec, J. . Ergotherapie bij ouderen met dementie en hun mantelzorgers: Het EDOMAH-programma. Houten: Bohn Stafleu van Loghum, 2010.

- 13. Teri L, Logsdon, R. G., Uomoto, J., McCurry, S. M. Behavioral treatment of depression in dementia patients: a controlled clinical trial. *Journals of Gerontology- series B: Psychological Sciences and Social Sciences*. 1997;52(4):P159-166.
- 14. Gitlin LN, Winter L, Burke J, Chemett N, Dennis MP, Hauck WW. Tailored activities to manage neuropsychiatric behaviors in persons with dementia and reduce caregiver burden: A randomized pilot study. *American Journal of Geriatric Psychiatry*. 2008;16(3):229-239.
- 15. Teri L, Gibbons LE, McCurry SM, et al. Exercise plus behavioral management in patients with Alzheimer disease: a randomized controlled trial. *JAMA*. 2003;290(15):2015-2022.
- 16. Graff MJ, Vernooij-Dassen MJ, Zajec J, Olde Rikkert MG, Hoefnagels WH, Dekker J. How can occupational therapy improve the daily performance and communication of an older patient with dementia and his primary caregiver? A case study. Dementia: The International Journal of Social Research and Practice. 2006;5(4):503-532.
- Graff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM. Community based occupational therapy for patients with dementia and their care givers: Randomised controlled trial. British Medical Journal. 2006;333(7580):1196-1199.
- 18. Prick AE, de Lange J, Twisk J, Pot AM. The effects of a multi-component dyadic intervention on the psychological distress of family caregivers providing care to people with dementia: a randomized controlled trial. *International Psychogeriatrics*. 2015;27(12):2031-2044.
- Graff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM. Effects of community occupational therapy on quality of life, mood, and health status in dementia patients and their caregivers: A randomized controlled trial. *Journals of Gerontology* - Series A Biological Sciences and Medical Sciences. 2007;62(9):1002-1009.
- Döpp CM, Graff MJ, Teerenstra S, Olde Rikkert MG, Nijhuis-van der Sanden MW, Vernooij-Dassen MJ. Effectiveness of a training package for implementing a community-based occupational therapy program in dementia: a cluster randomized controlled trial. *Clinical Rehabilitation*. 2015;29(10):974-986.
- 21. Voigt-Radloff S, Graff M, Leonhart R, et al. A multicentre RCT on community occupational therapy in Alzheimer's disease: 10 sessions are not better than one consultation. *British Medical Journal open*. 2011.
- 22. Döpp CM, Graff MJ, Olde Rikkert MG, Nijhuis van der Sanden MW, Vernooij-Dassen MJ. Determinants for the effectiveness of implementing an occupational therapy intervention in routine dementia care. *Implementation Science*. 2013;8:131.
- 23. Prick AE, de Lange J, Van 't Leven N, Pot AM. Process evaluation of a multicomponent dyadic intervention study with exercise and support for people with dementia and their family caregivers. *Trials.* 2014;15(1):401.
- 24. Vernooij-Dassen M, Moniz-Cook E. Raising the standard of applied dementia care research: addressing the implementation error. *Aging & Mental Health*. 2014;18(7):809-814.

- 25. Voigt-Radloff S, Graff M, Leonhart R, Hull M, Rikkert MO, Vernooij-Dassen M. Why did an effective Dutch complex psycho-social intervention for people with dementia not work in the German healthcare context? Lessons learnt from a process evaluation alongside a multicentre RCT. *British Medical Journal Open*. 2011;1(1):e000094.
- 26. Brodaty H, Arasaratnam C. Meta-analysis of nonpharmacological interventions for neuropsychiatric symptoms of dementia. *American Journal of Psychiatry*. 2012;169(9):946-953.
- 27. Gitlin LN. Good news for dementia care: caregiver interventions reduce behavioral symptoms in people with dementia and family distress. *American Journal of Psychiatry*. 2012;169(9):894-897.
- Van Mierlo LD, Van der Roest HG, Meiland FJ, Dröes RM. Personalized dementia care: proven effectiveness of psychosocial interventions in subgroups. *Ageing Research Reviews*. 2010;9(2):163-183.
- 29. Hsieh HF, Shannon SE. Three approaches to qualitative content analysis. *Qualitative Health Research*. 2005;15(9):1277-1288.
- Hunter A, Murphy K, Grealish A, Casey D, Keady J. Navigating the grounded theory terrain. Part 1. Nursing Research. 2011;18(4):6-10.
- 31. Kvale S. Interviews. London: SAGE Publications, 1996.
- 32. Maso I, Smaling, A. Kwalitatief onderzoek: praktijk en theorie [Qualitative research: practice and theory]. Amsterdam: Boom, 1998.
- 33. Beuscher L, Grando VT. Challenges in conducting qualitative research with individuals with dementia. *Research in Gerontological Nursing*. 2009;2(1):6-11.
- 34. van Baalen A, Vingerhoets AJJM, Sixma HJ, de Lange J. How to evaluate quality of care from the perspective of people with dementia: An overview of the literature. *Dementia: The International Journal of Social Research and Practice.* 2011;10(1):112-137.
- 35. Friese S. Qualitative data Analysis with ATLAS.ti. London: SAGE Publications, 2012.
- 36. Boeije H. Analysis in qualitative research. London:Sage, 2010.
- Bogdan RC, Biklen SK. Qualitative research for education: an introduction to theory and methods. Boston: Allyn & Bacon, 1997.
- van der Roest HG, Meiland FJM, Maroccini R, Comijs HC, Jonker C, Dröes RM. Subjective needs of people with dementia: a review of the literature. *International Psychogeriatrics*. 2007;19:559-592.
- Haberstroh J, Neumeyer K, Krause K, Franzmann J, Pantel J. TANDEM: Communication training for informal caregivers of people with dementia. *Aging & Mental Health*. 2011;15(3):405-413.

CHAPTER 4 -

Working mechanisms of dyadic, psychosocial, activating interventions for people with dementia and informal caregivers: a qualitative study

Published as:

Van't Leven N, de Lange J, Ploeg ES van der, Pot AM. Working mechanisms of dyadic, psychosocial, activating interventions for people with dementia and informal caregivers: a qualitative study. *Clinical Interventions in Aging*, 2018; 13, 1847-1857. doi: 10.2147/CIA.S160363. eCollection 2018.

Background:

People living with dementia and their informal caregivers often report difficulties in maintaining their usual activities. Several dyadic, psychosocial, activating interventions were developed to create insight into how to cope with limitations in a practical way and to increase skills for maintaining personal activities. Effects of these interventions varied so far. More knowledge on the working mechanisms of activating interventions might help to increase their impact. This study will provide more insight into the working mechanisms of three dyadic, activating interventions as the participating people with dementia, caregivers, and coaches perceived them during the intervention.

Methods:

We used a qualitative design with semi-structured interviews of 34 dyads that included 27 persons with dementia, 34 caregivers, and 19 coaches. The data were analyzed with the constant comparative method.

Results:

Many people with dementia and caregivers found capacity-building approaches beneficial. We identified *empowerment* as the core theme. Three working mechanisms explained why the interventions were perceived having a beneficial impact. They were: 1. enabling activities without providing false hope; 2. exploring the most important personal activity needs of the people with dementia and caregivers; and 3. a solution-focused approach to adapt, test, and practice activities. An individualized approach contributed to positive change.

Conclusion:

For the overarching working mechanism empowerment, expertise in the field of dementia and psychological skills of coaches seem to be important. Assessing the needs, capacities, and limitations of both people with dementia and caregivers to take part in activities seems to be key, as well as the communication about it and the skills to teach a solution-focused approach. Studies on the impact of psychosocial and activating interventions might benefit from considering outcome measures directed at empowerment, along with aspects such as hope and belief in one's own capacities versus traditional outcome measures as mood, burden, or quality of life.

Introduction

Worldwide, 50 million persons with dementia have to cope with the daily consequences.¹ There is currently no cure for dementia. Dyadic psychosocial interventions aim to support both people living with dementia and their informal caregivers (usually spouses or children).^{2,3} People with dementia often report difficulties in performing their usual activities.^{4,5} Daily activities are important because they can create structure, satisfaction, and continuity in life, and they contribute to well-being and health.⁶⁻⁸ It is difficult for caregivers to engage in joint social activities with people with dementia.⁹ Furthermore, due to their care-giving tasks, they have difficulty maintaining their own activities as well.^{10,11} Some psychosocial interventions take these challenges as a starting point and are focused on engaging both people with dementia and caregivers in activities.¹²⁻²⁰ They aim to increase skills for maintaining participation in activities and how to cope with declines capacities in a practical way, to increase self-management.^{21,22} These multicomponent interventions comprise performing activities, psychoeducation, and emotional support. In this study, we label these interventions "activating interventions".

Studies show benefits of activating interventions, but the effects vary widely, from no effects at all to large effects on people with dementia-outcomes, such as outcomes for mood, behavior, activities of daily living (ADL) dependency, and quality of life. This is also true for caregiver outcomes such as mood, burden, competency, and quality of life.^{12-15,18,19,23-29} Some authors address the issue that knowledge on the impact of the different components in multi-component interventions is lacking.^{3,30-32} A few studies provide information about the relation between treatment components and effects. Pinguart showed positive effects of counseling and psycho-education for the caregiver's active engagement.³³ Van 't Leven et al showed a relation between skills training of people with dementia and their ADL independence, and a relation between training of caregivers' communication skills and their competence.³ Furthermore Brodaty and Asaratnam reported indications for delivery characteristics of successful multi-component interventions. These indications are the interventions are multi-component, tailored to the needs of people with dementia and caregivers, include 9-12 sessions, and are delivered at home. They are interspersed with telephone sessions and individual or group follow-ups.³⁰ Olazeran et al., however, could not conclude that any relation existed between the effects of either treatment components or delivery characteristics in their review.³²

More knowledge on working mechanisms of the interventions may add to more effective support for people with dementia and caregivers. In this study, we concentrate on dyadic, psychosocial, activating interventions because of the people with dementia's and caregivers' activity losses that are consequences of dementia in daily life. We used data from studies about three activating interventions: the Pleasant Events Program, the Exercise and Support Intervention for People with Dementia and their Caregivers, and the Community Occupational Therapy in Dementia guideline.^{12,16,17,19,20} These interventions aim at increasing skills to cope with the consequences of dementia. They are all dyadic, training activities, and they are delivered at home. Table 1 provides a detailed description of the treatment components, the evidence found in effect studies, and the organizational structure in which the interventions were offered to person with dementia-caregiver dyads.

	Pleasant Events Program 12,19,20	Exercise and Support Intervention for People with Dementia and their Caregivers ^{17,18}	Community Occupational Therapy in Dementia (COTiD) ^{16,24,29}
Primary goal	Maintaining activity, having pleasant activities, and preventing depression of PWD and CGs.	Improving mood and physical health of PWD and CGs. Decreasing burden of CGs.	Improving the performance of daily activities of PWD, and mitigating the care burden of CGs.
Components	 Choosing and planning pleasant activities for PWD and CGs or both, Practicing these activities, Adjusting activities to the capabilities of PWD, Psycho-education, Homework: pleasant activities. 	 Exercises for flexibility, balance, power, and stamina of PWD and the CGs, supervised by a coach at home. Identifying pleasant activities for both, Psycho-education and communication training, Homework: repeating mobility exercises and pleasant activities (preferred at least 3 times a week). 	 Needs assessment and observation, Practicing meaningful, daily activities (self-suffici- ency) with compensation strategies and adjustments for PWD, Psycho-education, Practicing different ways with CGs of approaching PWD, Homework: individual appointments for practicing activities and approach.
Duration	A maximum of 6 home visits of 1.5 hours each, weekly and biweekly.	A maximum of 8 home visits of 1 hour each, 4x weekly, 4x biweekly.	A maximum of 10 home visits of 1 hour each, 2x a week, weekly and biweekly.

>>

 Table 1
 Interventions: Treatment components, evidence and provision of the interventions

>>	Pleasant Events Progra 12,19,20		Exercise and Support Intervention for People with Dementia and their Caregivers ^{17,18}	Community Occupational Therapy in Dementia (COTiD) ^{16,24,29}
	Original interventions	This program is based on the behavioral treatment of depression in dementia program. ¹⁹ The Dutch manual for the program was published by Dutch Institute for Primary Healthcare ²⁰ , and is supplemented by examples from Gitlin et al.'s program ¹²	This intervention is based on Reducing Disability in Alzheimer's Disease. ¹⁸ After a pilot study with the original program, this program was adapted. Attributes like a ball, weights, and elastic equipment made the exercises more attractive for people with dementia. The amount of sessions was limited to eight. This program emphasized the pleasant events component and the training for activating events, beliefs, and consequences (ABC) was included in the psycho-edu- cation for CGs. ¹⁷	This therapy is described in Graff et al. ^{16,24,45} Four sessions of diagnostics and goal defining to priorities meaningful activities. Next compensatory strategies to adapt activities for PWD and the use of environmental modifications are trained, the CG is trained in supervision skills.
	Evidence	Significant effects (post treatment) for PWD on - Depression (CDS p<.001; ES 1.7) and for CG Depression (HDRS p<.01) N (I/C): 23/20. ¹⁹ Significant effects (4 months) for PWD on - Activity engagement (p.029; 95% CI 0.02-0.41) - Frequency of agitated behavior (p.009; 95% CI5509) and for CG on - Mastery (p.013; 95% CI 0.08 - 0.60) - Confidene using activities (p.011; CI 0.41 - 2.94) N (I/C): 27/29. ¹²	Significant effects (3 months) for PWD on - Depression (CDS p .02; MED -1.03 (95% CI -0.17-1.19) - General health (SP-36 p <.001; MED 19.29 (95% CI 8.75-29.83). N (I/C): 76/77. ¹⁸ Significant effect (3, 6 months for PWD on - Attention span of PWD (p .04; ES 0.25) N (I/C): 57/54 ²⁶ Effects were not demon- strated in a RCT with the adjusted program ^{25,27}	Significant effects (3 months) for PWD on - Daily functioning (AMPS process: p <.0001; ES 2,7) - Depression (CDS: p<.0001; ES 0,7) - General health (GHQ: p<.0001; ES 0,7) - Quality of life (Dqol: p<.0001; ES 1,1) and for CGs - Competence (SCQ: p<.0001; ES 1,3) - Depression (CES-D: p<.0001; ES 1.3) - General health (GHQ: p .001; ES 1.1) - Quality of life (Dqol: p<.0001; ES 1.5) N(1/C): 68/67. ^{24,29} These effects were not demonstrated in other studies. ^{23,28}

	Pleasant Events Program 12,19,20	Exercise and Support Intervention for People with Dementia and their Caregivers ^{17,18}	Community Occupational Therapy in Dementia (COTiD) ^{16,24,29}
Organizational structure	The program was offered as part of the Academic Collaborative Centre for Dementia, which entails the collaboration of Rotterdam University, the Home Care Organization De Zellingen and the Van Kleef Institute (for generating and dis- seminating knowledge for home care professionals). The coaches were students graduating in healthcare studies and home-care workers from De Zellingen. They were supervised during the intervention. The project leader and case manager in the region recruited participants.	The program was offered within an RCT from the Department of Clinical Psychology, VU University, Amsterdam ¹⁷ The coaches were Master students from the Depart- ment of Clinical Psychology, VU University, who completed a special training program for geropsychology. They were supervised during the project. The project leader recruited participants for the program with the help of caregiver organizations, local Alzheimer cafes (public meetings for people with dementia, their caregivers, and others) and case managers throughout the Netherlands.	Occupational Therapy was regularly offered and delivered by certified occupational therapists, who followed an additional training for this program. The participants were referred by general practitioners and geriatricians as well as case managers throughout the Netherlands.

PWD = people living with dementia; CG = informal caregiver

RCT = Randomized Controlled Trial; p = p-value; ES = effect size; CI = Confidence Interval;; MED = Mean Estimated Difference; N (I/C): Study population Intervention/Control group;

AMPS = Asessment of Motor and Process Skills; CDS = Cornell Depression Scale; CES-D = Centre of Epidemic Studies – Depression; Dqol = Dementia Quality of Life instrument; GHQ = General Health Questionnaire; HDRS = Hamilton Depression Rating Scale; SCQ = Sense of Competence Questionnaire; SP-36= Sickness Profile-36

The aim of our study was to provide more insight into the working mechanisms of three dyadic, activating interventions as the participating people with dementia, caregivers, and guiding coaches perceived them during the intervention? We define "working mechanisms" as "specific processes within psychosocial interventions which appear to contribute to positive changes", according to Dugmore and colleagues.³¹

Participants and methods

Design

>>

We used a qualitative design with semi-structured interviews to gather in-depth information about intervention processes that contributed to positive change for

the participating dyads.^{34,35} We wanted to explore the perspectives of participating people with dementia and caregivers, as well as the perspective of the coaches who guided them during the intervention.

Ethical considerations

Ethical approval for interviews with the participating dyads was obtained as part of the approving process of the ethics committees for the Pleasant Events Program and the Exercise and Support Intervention [Medical Ethics Review Committee of the Erasmus Medical Center, Rotterdam, The Netherlands (number 2009-117), and the Medical Ethics Review Committee of the VU University Medical Center, Amsterdam, The Netherlands (number 2008/320)]. Ethical approval was obtained separately for the interviews with participants in the Occupational Therapy Intervention [Medical–Ethical Review Committee for Mental Health Care, Rotterdam, The Netherlands (number 11.123)]. The dyads consented to the study procedures, including participation in evaluative interviews. The approval for the Pleasant Events Program and Occupational Therapy Intervention included written informed consent from the caregiver who signed for the dyad. The approval for the Exercise and Support Intervention included written informed consent from both the people with dementia and the caregivers.

Recruitment and participants

For the Pleasant Events Program and the Exercise and Support Intervention, the project managers asked the participants (a total of 23 dyads) and the coaches if they were willing to participate in an interview. All but one dyad agreed to be interviewed. For the Occupational Therapy intervention, which was reimbursed under the Dutch Health Insurance Act, we contacted occupational therapists and asked them to identify clients willing to participate in an interview. We contacted 51 occupational therapists. Nine of them identified 16 dyads who were willing to participate in an interview. Thus, a convenience sample was composed of 38 cases, consisting of a person with dementia, a caregiver and a coach in each. The other occupational therapists had no referrals for these activating interventions for people with dementia (26), some indicated they were too busy to take part (3), and others thought it too burdening for the dyad(s) they coached (2), or the dyad(s) had quit early (1), and the reason is unknown (10).

Upon receiving names and contact details, the interviewer (NL or AEP) phoned the dyad, explaining again the goal of the interview and the study. If both the person with dementia and the caregiver were still willing to be interviewed, an appointment was made. Four dyads cancelled their interviews because of the burden or illness. Eventually 34 dyads (9 for the Pleasant Events Program, 11 for the Exercise Intervention, and 14 for the Occupational Therapy Intervention) were interviewed. We interviewed 19 coaches, who guided 28 dyads. The coaches consisted of professional occupational therapists, master students of geropsychology, 4th-year healthcare students, and homecare workers, all with the supervision of an experienced psychologist or occupational therapist. Table 2 provides more information about the participants.

34 Cases	Mean age (years),	Male/ Female	Relation	Onset of dementia: (Mean)	Care services used by dyads	Interviewed in cases
Persons with dementia	78 (62-93)	22 / 12 (65/35%)		1–5 Years (2.6)	Geriatrician /general physician 24 x Case manager 17 x Meeting centre or day care 8 x Home care 3 x Speech therapy 1 x Pastoral worker 9 x Informal care support/ volunteer/ Alzheimer café	27*
Caregivers	Partners: 75 (61-88) Children: 53 (45-65)	9 /25 (27/73%) -	28 Partners 6 Children			34
	Professional Experience:					
Coaches	0-30 Years					28 coaches in 34 cases**

Table 2 Participants

* Not all PWDs were able to participate in an interview

**Some coaches were involved in more than one case

Data collection

We aimed to interview both members of the dyad, either together or separately, whichever they preferred. Seven caregivers said that the partner or parent with dementia could no longer reflect on their experiences, and a joint interview would be too burdensome for them. Only the caregivers in these dyads were interviewed. In two interviews, a daughter was present in addition to the person with dementia and caregiver. All interviews started with an open question about their experiences with the intervention. The follow-up questions for the persons with dementia addressed the main topics or activities in the intervention and what they had learned from participating. Non-verbal signs such as frowning were interpreted as prompts to pose a question in another way or to investigate the answer further. Follow-up questions for the caregivers addressed the topics, activities, what had been learned, what had been changed and how or why. The interviews with the dyads varied from 50 to 120 minutes, with a mean of 75 minutes. The interviews took place at the homes of the dyads, except for one caregiver who was interviewed by phone. AEP conducted seven interviews with dyads in the Exercise and Support Intervention. NL conducted all the other interviews.

The interviews with the coaches took place after the interview with the dyad, and they also began with an open question about the coach's experience with the intervention with that particular dyad. Follow-up questions addressed the achieved results, how the coach tailored the intervention to the needs and preferences of the dyad, and what was not achieved. The interviews with the coaches lasted 20 to 30 minutes. These interviews took place at their workplaces (12), homes (4), a public place (2), or by phone (1).

Field notes were taken during all interviews. All face-to-face interviews were digital-audio recorded and transcribed verbatim. Notes were taken during the two interviews by phone and transcribed extensively immediately afterwards. All transcripts and field notes were imported into Atlas-ti-6/7.1 to facilitate qualitative analysis.³⁶

Data analysis

We performed a conventional content analysis, following an iterative and reflexive process based on the constant comparative method.^{37,38}

Stage 1

Three researchers (NL, JL, and AEP) coded the interviews for the first four cases independently. There was agreement about the selection of significant sections. The codes used for the treatment components were given in the intervention manuals, such as those for the chosen activities or exercises, instructions for persons with dementia, homework, needs inventory, adaptations, and open codes for the persons with dementia, like telling their story and being heard, and staying active; for caregivers, practical advice and reinforcement; for coaches, communication, revealing needs, and memory aids. We also coded the effects as the person with dementia–caregiver dyad perceived them; these were effects such as self-confidence, pleasure, new ideas, and more knowledge. Differences were discussed, and a preliminary code tree was composed. NL used this code tree to analyze seven

cases again. After 11 cases had been done, each case was summarized, and the working mechanisms of each case and the effects as perceived by the interviewees were described.

Stage 2.

The research team discussed these results based on these eleven cases. The codes were grouped into more abstract categories that covered what contributed to positive change for the dyads, such as the importance of activities, attention for both capacities and limitations, confidence in the coach, hope, and individualized adaptations. The code tree was changed as necessary. Next, ten new cases were coded and the 11 cases that had been coded were re-coded in accordance with the new code tree. After re-coding the 21 interviews, more categories were formulated, such as actually doing things together with the person with dementia, emphasis on potential possibilities, the coach's expertise in dementia, and the coach's creativity. We recognized three themes in these categories.

Stage 3

These themes and categories were deepened with the experiences of the last 13 cases.

Stage 4

Then we identified one encompassing theme with three subthemes.

Results

We identified *empowerment* as the core theme in the stories of our participants, e.g. focusing on remaining capacities, searching for strengths that could compensate for limitations, and increasing self-confidence without denying limitations. Thus a dyad was enabled to get a grip on their situation and find out what works for them by themselves. The people with dementia and caregivers appreciated the positive approach of focusing on capabilities instead of failures. Some dyads saw the intervention as a training course to learn how to cope with their situation by improving their capacities instead of a therapy focusing on problems. The emphasis was on "getting a grip on their situation" rather than on "being in need of help". Some coaches noticed that the dyads had already discussed their daily problems with other care professionals. Often these professionals acknowledged that their problems were a consequence of dementia, but they had not proposed ways of

dealing with the problems and compensating for limitations. Three working mechanisms emerged within the core theme of empowerment, which were present in all three interventions: 1. enabling activities without providing false hope; 2. exploring the most important personal activity needs of the person with dementia and the caregiver; and 3. a solution-focused approach to adapt, test, and practice activities. An individualized combination (rather than single components) of these working mechanisms, tailored to the needs and capacities of the dyad, contributed to positive change. Figure 1 shows working mechanisms and components.



Abbreviations: CG, informal caregiver; PWD, people living with dementia

Figure 1 Working mechanisms

Working mechanism 1. Enabling activities without providing false hope

This mechanism involves emphasizing capacities instead of limitations in a realistic way. This made the people with dementia and caregivers more aware of qualities that were still intact. The participants said they focused on maintaining activities, while acknowledging difficulties in the performance.

The therapist insisted that I should keep doing what I used to do. I felt inclined to pull back, but now I say to myself, if I stop, there will be nothing left. [A person with dementia]

Some participants reflected on the intervention as "an opportunity to learn", as if it was a training course to get better equipped for the situation.

Both people with dementia and caregivers often spoke of the coach's encouragement and positive approach. As a result, the dyads were more accepting of the intervention and wanted to "give it a try". All coaches emphasized the subtle and difficult balance between a focus on possibilities and staying active, but also recognized the reality of the dyad's situation, including limitations. They found it challenging to explain that maintaining daily and recreational activities may help to cope with the dementia process, although the activities could not relieve grief and feelings of loss. They needed to show enthusiasm and optimism, but had to avoid creating false hope. Especially the first contact was important for exploring the participants' interest in the intervention and attuning to their motivation. The dyads were more comfortable if the first contact took place in a home visit because they felt more at ease in their own homes.

Several respondents told us that the immediate aftermath of receiving the diagnosis was still affecting them at the time of the interview. In their perception, the physician had told them there was no therapy for them: "nothing can be done about it". For some participants, this message confirmed their own premises of dementia that "it's all over". They had lost their belief in their own capacities and had become apathetic.

After the diagnosis, my father [with dementia] slept the whole day, while he used to be busy all the time. The geriatrician suggested day care on a farm, but my father felt 'too good for this'. To gain insight into his activities, the occupational therapist asked him to write down what he did during the day.

The result was that in the morning he wrote down all kinds of activities to do, and the rest of the day he actually performed them. [Daughter]

When she (the occupational therapist) visited me again, I could show her my notes. [Father]

So, this approach, directed at what he was still able to do, did help him. [Daughter]

Working mechanism 2. Exploring the most important personal activity needs of people with dementia and caregivers

This mechanism is about accurate, detailed exploration of needs of both people with dementia and caregivers and assessing physical and cognitive limitations of the person with dementia, as well as identifying the remaining capacities and usable coping strategies of both. The dyads thought it important that their personal needs formed the core of the interventions, rather than being offered a pre-defined intervention program. The people with dementia recognized the genuine interest in themselves as individuals; they did not merely represent instances of a disease. The caregivers, too, said that attention for them as individuals was more helpful than being regarded only as caregivers.

I could show people who I am in a way. Well, I liked that very much. [A person with dementia] In fact, it was the first time I told the story of your dementia from my perspective. [His caregiver]

Feeling that the coach understood them was extremely important. Feeling understood meant that the coach listened well, was familiar with the diagnosis, understood the perceived difficulties, and maintained a positive attitude. Common was that the coaches inquired not only about the daily pattern of activities, but also about previous activities and important events in the lives of the dyad. This helped them uncover the dyad's perspective and gain the dyad's confidence.

The coach explored their needs and challenges with regard to activities. It was often difficult to puzzle out the dyad's real needs. Although some caregivers asked concrete questions such as "Can he or she (person with dementia) learn to use a mobile phone?", others asked general, sometimes unrealistic questions. The people with dementia said they wanted "to cope with memory problems" or wished "to be as the way they used to be before". Several caregivers wanted to know how to support the person with dementia, how to keep him or her active, or how to cope with feelings of burden. Often needs were formulated as wishes. The coaches explored those wishes and activities in detail with them, and tried to uncover what a dyad really missed.

It took some effort finding out the needs of this caregiver and the right way to approach him. I had to make him find out for himself instead of telling him. [Coach]

Nearly all the participants found actually doing activities or exercises with the coach helpful. Several persons with dementia could still reflect on this and had perceived it as a collaboration with the coach which made them feel more involved. However, one other person with dementia had experienced it negatively, as a way of being tested. Most of the caregivers were confident that the coach had a realistic picture of the capacities of the person with dementia. The coaches needed insight

into capacities, including declines in cognitive and physical functioning, of both people with dementia and caregivers. This could be obtained by observation and performing activities together. Furthermore, this allowed insight into the communication between the person with dementia and the caregiver. Visiting a dyad at home helped the coaches build confidence and gave them insight into the individual daily routines. This information, together with the clarified needs, enabled the coach and the dyad to set attainable goals that matched in the intervention.

Working mechanism 3. A solution-focused approach to adapt, test, and practice activities

This mechanism is about the practical adjustment of meaningful activities to personal capacities, and experimenting with new ways of overcoming problems. Adapted ways of performing activities were established through repetition as a new routine. We recognized a solution-focused approach by piloting adaptations in collaboration with the coach, the person with dementia, and the caregiver. Patience and persistence were needed to find an individual, tailored approach. The coaches needed creativity when searching for appropriate activities, tailored exercises, and adaptations to the individual needs of the persons with dementia and caregivers. The actual doing often created ideas for possible adaptations of activities or exercises, which sometimes had not been foreseen. The coaches used expressions such as *'it turned out that...'*, or after some attempts *'then suddenly we found out that...'*.

Her daughter wanted to encourage her because she was just sitting there the whole day. As a pianist, she had given concerts in earlier days. After a few visits, she was willing to sit with me at the piano, and fortunately she started to play (because I couldn't...). I asked her to write down in her diary: play a few times a week, and she wrote: 'Mind the piano!' Her neighbors told the daughter that they heard her playing. [Coach of a woman with dementia, living on her own with help from her daughter]

The coaches encouraged the dyad and involved them actively, they took on a coaching attitude. The dyads mentioned a variety of actions that had been helpful to them: simplifying activities, using memory aids, changing routines, experimenting with an adapted attitude and adapted instruction, validation of providing good care for the caregivers, and enhancing informal or professional support. Some received adaptations for physical limitations.

Most caregivers felt insecure about their way of providing care, and they were worried that it might be improper or suboptimal. They felt that being empowered to try out several options to manage daily problems was important.

It provided me with the mindset just to try; if it doesn't work today, maybe I can try again tomorrow. [A caregiver]

Several caregivers mentioned that they had already had made some changes themselves and that the intervention had not changed much, but it increased their self-confidence.

Discussion

We identified *empowerment* as the core theme that explained why the interventions were perceived as working according to people with dementia, caregivers, and their assisting coaches. More specifically, we found three working mechanisms for the beneficial impact of the interventions in the interviews:

- 1. enabling activities without providing false hope;
- 2. a detailed exploration of the most important activity needs; and
- 3. a solution-focused approach to adapt, test, and practice activities.

The first working mechanism, enabling activities without providing false hope taught people with dementia and caregivers to cope with their situation by improving their capacities instead of emphasizing problems. Several dyads reflected very positively on the intervention as a training course instead of needing support. The term "support" was associated with helplessness, whereas the term "training" was associated with maintaining skills.³⁹ Although people living with early dementia and their caregivers experience all kinds of difficulties, they often do not want to be seen as being in need of professional help, which they may perceive as negative and stigmatizing.⁴⁰ Emphasizing possibilities without providing false hope also means an opening to the future. Realistic beliefs help both people with dementia and caregivers to cope in a positive manner and to maintain better health and social relationships.⁴¹ Especially in the case of progressive diseases like dementia there is a thin line between realistic hope and false hope. It is always uncertain whether the benefits of an intervention aiming to maintain capacities will surpass the decline. Coaches with positive experiences with the interventions can express a

certain convincing attitude and enthusiasm.⁴² The second working mechanism is essential for professionals to provide realistic expectations.

The second working mechanism to empower people with dementia and caregivers is a detailed exploration of their most important needs for activities and their capacities and limitations. Knowledge about dementia, understanding its consequences in daily life, and knowing how to communicate with people with dementia and caregivers are necessary to make dyads feel understood and, hopefully, to share their actual needs. Accurate needs assessment, including revealing latent needs, will clarify if and how person with dementia–caregiver dyads need support to maintain daily activities. In addition, a coach should display a positive attitude and be able to assess mental and physical capacities and limitations related to the performance of activities.⁴³

The third working mechanism is a solution-focused approach to adapt, test, and practice activities, and it is directed at the activity needs of both the person with dementia and the caregiver. To enable activities it is important to find creative, but also realistic and pragmatic solutions for people with dementia and caregivers to continue to do activities they value but need adaptation as a result of the disease. Actually doing things and the performance of activities can elicit creative, realistic, and pragmatic solutions, that fit both the people with dementia and their caregivers.⁴⁴

The respondents always mentioned more than one mechanism, rather than single components, that had been important for beneficial results and emphasized the process. An individualized approach, tailored to the needs and capacities of the dyad, contributed to positive change. Many of the people with dementia and caregivers shared in the interviews that they felt more self-confident, and that some people with dementia maintained meaningful activities. The dyads found solutions, which they could use for a longer time.

Strengths and limitations

A strength of this study is that we succeeded in interviewing 27 persons with dementia. Many of them could still talk about the intervention or the coach 6 or even 9 months later. It also helped that the caregiver was present at nearly all the interviews and often provided examples of activities that prompted the person with dementia to tell us about his or her experiences. Some people with dementia were no longer able to reflect on the intervention, but they could still indicate what activities were important to them and why. Since we had a convenience sample of dyads and coaches who were willing to participate in an interview, this sample may not be representative of the total group of people with dementia, caregivers, and professionals. Both dyads and coaches wanted to share their experiences to contribute to improving care for people with dementia. The intervention gave most dyads satisfactory benefits. For those dyads the interventions appeared to fit their needs and preferences rather well. We interviewed some dyads who stopped early in the intervention, too. They mainly needed information, were not ready at that moment for a practical approach, did not have activity needs, or the needs assessment had been to general. Those dyads afforded us information about preconditions for these interventions and what was not working. Furthermore, the heterogeneity in terms of age, education, relationship, work experience, duration of the dementia, and both positive and negative experiences may be representative for people with dementia and caregivers in general. The coaches had various background characteristics as well.

Recommendations for practice

Our findings can provide guidance for competences professionals need when they would like to provide an activating intervention to people with dementia and their caregivers. It is important to enable people with dementia and caregivers to continue to do activities taking into account any declines in capacities. Professionals have to be skilled in assessing the needs of people with dementia and caregivers and their capacities and limitations, and in training how to adapt activities and practice. Good communication skills are needed to engage both people with dementia and caregivers in the intervention. Starting from individual needs with a focus on possibilities requires the coaches to have an open and creative attitude in a collaborative process with the dyad.

Creativity and innovative thinking, may need more attention in training and education. Professionals may consider offering these interventions to people with dementia and caregivers in a training session to get a grip on the situation, rather than just giving care service, aligned with what people with dementia and caregivers perceive as the valuable mechanisms of these interventions.

Recommendations for research

Further study is needed to find out how the professionals guiding these interventions can achieve empowerment for people with dementia and caregivers, and how they can learn to develop these competencies. Studies on the impact of dyadic, psychosocial, activating interventions should focus on enabling activities despite declines in capacities and reinforcing self-efficacy. Therefore, outcome measures should also include self-efficacy, with aspects such as hope and belief in one's own capacities, along with outcome measures for mood, burden, and quality of life.

References

- 1. Livingston G, Sommerlad A, Orgeta V, et al. Dementia prevention, intervention, and care. *Lancet*. 2017;390(10113):2673-2734.
- Smits CHM, de Lange J, Dröes RM, Meiland F, Vernooij-Dassen M, Pot AM. Effects of combined intervention programmes for people with dementia living at home and their caregivers : a systematic review. *International Journal of Geriatric Psychiatry*. 2007;22(12):1181-1193.
- 3. Van't Leven N, Prick AE, Groenewoud JG, Roelofs PD, de Lange J, Pot AM. Dyadic interventions for community-dwelling people with dementia and their family caregivers: a systematic review. *Interntional Psychogeriatrics*. 2013;25(10):1581-1603.
- Nygard L. Responses of persons with dementia to challenges in daily activities: a synthesis of findings from empirical studies. *American Journal of Occupational Therapy*. 2004;58(4):435-445.
- van der Roest HG, Meiland FJM, Maroccini R, Comijs HC, Jonker C, Dröes RM. Subjective needs of people with dementia: a review of the literature. *International Psychogeriatrics*. 2007;19:559-592.
- Öhman A, Nygard, L. Meanings and Motives for Engagement in Self-Chosen Daily Life Occupations Among Individuals With Alzheimer's disease. OTJR: Occupation, Participation and Health. 2005;25(3):89-97.
- 7. Stav WB, Hallenen T, Lane J, Arbesman M. Systematic review of occupational engagement and health outcomes among community-dwelling older adults. *American Journal of Occupational Therapy*. 2012;66(3):301-310.
- 8. Vikstrom S, Josephsson S, Stigsdotter-Neely A, Nygard L. Engagement in activities: Experiences of persons with dementia and their caregiving spouses. *Dementia: The International Journal of Social Research and Practice*. 2008;7(2):251-270.
- 9. Van Bruggen S, Gussekloo J, Bode C, Touwen DP, Engberts DP, Blom JW. Problems experienced by informal caregivers with older care recipients with and without cognitive impairment. *Home Health Care Services Quarterly*. 2016;35(1):11-24.
- 10. Peeters J, Van Beek, APA, Meerveld, JHCM, Spreeuwenberg, PMM, Francke, AL. Informal caregivers of persons with dementia, their use of and needs for specific professional support: a survey of the National Dementia Programme. *BMC Nursing*. 2010;9(9).
- 11. Vikström S, Borell L, Stigsdotter-Neely A, Josephsson S. Caregivers' Self-Initiated Support Toward Their Partners With Dementia When Performing an Everyday Occupation Together at Home. OTJR: Occupation, Participation and Health. 2005;25(34):149-159.
- Gitlin LN, Winter L, Burke J, Chemett N, Dennis MP, Hauck WW. Tailored activities to manage neuropsychiatric behaviors in persons with dementia and reduce caregiver burden: A randomized pilot study. American Journal of Geriatric Psychiatry. 2008;16(3):229-239.

- 13. Gitlin LN, Winter L, Corcoran M, Dennis MP, Schinfeld S, Hauck WW. Effects of the home environmental skill-building program on the caregiver-care recipient dyad: 6-month outcomes from the Philadelphia REACH Initiative. *Gerontologist*. 2003;43(4):532-546.
- 14. Gitlin LN, Winter L, Dennis MP, Hodgson N, Hauck WW. A biobehavioral home-based intervention and the well-being of patients with dementia and their caregivers: the COPE randomized trial. *JAMA*. 2010;304(9):983-991.
- 15. Gitlin LN, Winter L, Dennis MP, Hodgson N, Hauck WW. Targeting and managing behavioral symptoms in individuals with dementia: A randomized trial of a nonpharmacological intervention. *Journal of the American Geriatrics Society*. 2010;58(8):1465-1474.
- 16. Graff M, Melick Mv, Thijssen M, Verstraten P, Zajec J. Ergotherapie bij ouderen met dementie en hun mantelzorgers: Het EDOMAH-programma [Occupational Therapy for Older Persons with Dementia and their informal Caregivers: COTiD-programme]. Houten: Bohn Stafleu van Loghum, 2010.
- Prick AE, de Lange J, Scherder E, Pot AM. Home-based exercise and support programme for people with dementia and their caregivers: study protocol of a randomised controlled trial. BMC Public Health. 2011;11:894.
- 18. Teri L, Gibbons LE, McCurry SM, et al. Exercise plus behavioral management in patients with Alzheimer disease: a randomized controlled trial. *JAMA*. 2003;290(15):2015-2022.
- 19. Teri L, Logsdon RG, Uomoto J, McCurry SM. Behavioral treatment of depression in dementia patients: a controlled clinical trial. *Journals of Gerontology Series B Psychological Sciences and Social Sciences*.1997;52(4):P159-166.
- Verkaik R, Francke, A, Lahr, M, Zuidervaart, M. Huisbezoeken; Voorkom Somberheid bij Geheugenproblemen, handboek voor de hulpverlener (plezierige activiteitenmethode) [Home visits: Preventing depressive mood in memory problems, manual for the professional: Pleasant activities]. Utrecht: NIVEL; 2005. Available in Dutch from https://www.nivel.nl/en/node/891.
- 21. Mountain GA, Craig CL. What should be in a self-management programme for people with early dementia? *Aging & Mental Health.* 2012;16(5):576-583.
- O'Sullivan G, Hocking C. Translating action research into practice: seeking occupational justice for people with dementia. OTJR: Occupation, Participation and Health. 2013;33(3):168-176.
- 23. Döpp CM, Graff MJ, Teerenstra S, Olde Rikkert MG, Nijhuis-van der Sanden MW, Vernooij-Dassen MJ. Effectiveness of a training package for implementing a community-based occupational therapy program in dementia: A cluster randomized controlled trial. *Clinical Rehabilitation*. 2014.
- Craff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM. Community based occupational therapy for patients with dementia and their care givers: Randomised controlled trial. British Medical Journal. 2006;333(7580):1196-1199.
- 25. Prick AE, de Lange J, Scherder E, Twisk J, Pot AM. The effects of a multicomponent dyadic

intervention on the mood, behavior, and physical health of people with dementia: a randomized controlled trial. *Clinical Interventions in Aging*. 2016;11:383-395.

- 26. Prick AE, de Lange J, Scherder E, Twisk J, Pot AM. The Effects of a Multicomponent Dyadic Intervention With Physical Exercise on the Cognitive Functioning of People With Dementia: A Randomized Controlled Trial. *Journal of Aging and Physical Activity*. 2017;25(4):539-552.
- 27. Prick AE, de Lange J, Twisk J, Pot AM. The effects of a multi-component dyadic intervention on the psychological distress of family caregivers providing care to people with dementia: a randomized controlled trial. *International Psychogeriatrics*. 2015;27(12):2031-2044.
- 28. Voigt-Radloff S, Graff M, Leonhart R, et al. A multicentre RCT on community occupational therapy in Alzheimer's disease: 10 sessions are not better than one consultation. *British Medical Journal open*. 2011.
- Craff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM.
 Effects of community occupational therapy on quality of life, mood, and health status in dementia patients and their caregivers: A randomized controlled trial. *Journals of Gerontology* - Series A Biological Sciences and Medical Sciences. 2007;62(9):1002-1009.
- 30. Brodaty H, Arasaratnam C. Meta-analysis of nonpharmacological interventions for neuropsychiatric symptoms of dementia. *American Journal of Psychiatry*. 2012;169(9):946-953.
- 31. Dugmore O, Orrell M, Spector A. Qualitative studies of psychosocial interventions for dementia: a systematic review. *Aging & Mental Health*. 2015;19(11):955-967, p.962.
- 32. Olazaran J, Reisberg B, Clare L, et al. Nonpharmacological therapies in Alzheimer's disease: a systematic review of efficacy. *Dementia and Geriatric Cognitive Disorders*. 2010;30(2):161-178.
- 33. Pinquart M, Sorensen S. Helping caregivers of persons with dementia: which interventions work and how large are their effects? *International Psychogeriatrics*. 2006;18(4):577-595.
- 34. Hsieh HF, Shannon SE. Three approaches to qualitative content analysis. *Qualitative Health Research*. 2005;15(9):1277-1288.
- 35. Kvale S, Brinkmann S. InterViews: learning the craft of qualitative research interviewing. second ed. California: SAGE Publications, Inc, 2009.
- 36. Friese S. Qualitative data Analysis with ATLAS.ti. London: SAGE Publications, 2012.
- 37. Boeije H. Analysis in Qualitative Research. London: SAGE Publications Ltd, 2010.
- Bogdan RC, Biklen SK. Qualitative research for education: an introduction to theory and methods. Boston: Allyn & Bacon, 1997.
- Haberstroh J, Neumeyer K, Krause K, Franzmann J, Pantel J. TANDEM: Communication training for informal caregivers of people with dementia. *Aging & Mental Health*. 2011;15(3):405-413.
- 40. Boots LM, Wolfs CA, Verhey FR, Kempen GI, de Vugt ME. Qualitative study on needs and wishes of early-stage dementia caregivers: the paradox between needing and accepting help. International Psychogeriatrics. 2015;27(6):927-936.

- 41. Wolverson Radbourne EL, Clarke C, Moniz-Cook E. Remaining hopeful in early-stage dementia: a qualitative study. *Aging & Mental Health*. 2010;14(4):450-460.
- 42. Clemson L, Donaldson A, Hill K. Implementing person-environment approaches to prevent falls: a qualitative inquiry in applying the Westmead approach to occupational therapy home visits. *Australian Occupational Therapy Journal*. 2014;61:325-334.
- 43. Zwijsen SA, van der Ploeg E, Hertogh CM. Understanding the world of dementia. How do people with dementia experience the world? *International Psychogeriatrics*. 2016;28(7):1067-1077.
- 44. Kuiper C. The Eventmaker, The hybrid art of performing professionals, work-setting rehabilitation. Den Haag: Lemma, 2007.
- 45. Graff MJ, Vernooij-Dassen MJ, Zajec J, Olde Rikkert MG, Hoefnagels WH, Dekker J. How can occupational therapy improve the daily performance and communication of an older patient with dementia and his primary caregiver? A case study. *Dementia: The International Journal of Social Research and Practice.* 2006;5(4):503-532.

PART II

The appropriateness of criteria for referral to dyadic, activating interventions



CHAPTER 5 -

Indicators to estimate the appropriateness of activating interventions for people living with dementia and for their informal caregivers

Published as:

Van't Leven N, Ploeg ES van der, de Lange J, Pot AM. Indicators to estimate the appropriateness of activating interventions for people living with dementia and for their informal caregivers. *Aging & Mental Health*, 2018; 22(11), 1416-1423. doi:10.108 0/13607863.2017.1358353. Epub 2017 Aug 28.

Abstract

Objectives

Dyadic activating interventions support both people with dementia and their informal caregivers to maintain activities. For a person-centered approach referrers need insight in how specific interventions might meet individual needs, characteristics and preferences of a dyad. This study aimed to develop a set of indicators for three psycho-social dyadic, activating interventions.

Method

We used the 'RAND Appropriateness Method' directed at agreement on indicators within a panel of experts. Qualitative research had identified 31 relevant conceptual indicators. A panel of 12 experts in dementia care rated the extent to which these indicators are recognizable for them in their clinical practice. Indicators with median ratings in the top third segment of the nine-point-scale were considered recognizable.

Results

18/31 conceptual indicators (58%) were found recognizable in 75%-90% of the panelists' clients. Although consensus on the recognizability of some indicators about the need or preference for physical and social activities was lacking, the respondents nevertheless recommended including these in regular assessments. Other indicators were judged too difficult to recognize in clinical practice.

Conclusion

The selected indicators offer guidance to referrers on what intervention(s) to choose, and discuss the appropriateness in a shared decision making process, thus contributing to a person-centred approach.

Introduction

The need for community-based care of people living with dementia is anticipated to increase in the coming decades.¹ Many will be supported by informal caregivers to manage everyday life challenges. Tailored care and support for either is necessary because dementia affects their daily lives in different ways, depending not only on the type, stage and symptoms of the dementia, health and co morbidity but also life course, personality and coping, preferences and social situation.²⁻⁶ As a person-centred approach has been found important, formal care-services should include both the person with dementia and the informal caregiver.^{7,8} Integrated care pathway approaches for dementia in several Western-European countries include psychosocial and allied health interventions for both people with dementia and their caregivers to maintain occupational performance.^{9,10} For one of the major problems in dementia is cognitive decline, and therewith the difficulty to maintain usual activities.^{5,11,12} Several authors emphasize that continuous performance of familiar activities gives positive outcomes such as improved self-esteem, satisfaction and continuity in the life course.^{13,14} Caregivers, on the other hand, have difficulty maintaining their own activities because their care giving consumes much time and energy.^{4,15} Despite the centrality of deficits in social and occupational functioning, the assessment of dementia is too often limited to the medical domain.¹² Furthermore there is a lack of targeted interventions that are needs-based, without a negative or stigmatizing character, especially directed to acceptance, managing changing roles and selfmanagement.^{12,16-18} Therefore several interventions with a focus on what is possible whilst activating and enhancing strengths of the person with dementiacaregiver dyad have been developed.¹⁹ Effect studies show heterogeneous outcomes, but positive outcomes for functional domains, such as ADL-dependency for people living with dementia and competence for caregivers.²⁰ Still, people receiving a diagnosis of dementia often are not advised about supportive interventions.^{16,19} Possible reasons are that these interventions are not provided in the dyad's geographical area or if available, dyads are not referred to those interventions.^{12,16-18} Referrers such as general practitioners, geriatricians and case managers should take the initiative to inform dyads about these interventions. To be able to do so they need insight into goals of interventions and how these might meet a dyad's needs.²¹

Earlier research about personalized dementia care has already indicated features that influence effects of psychosocial interventions, such as depression, loneliness, the caregiver's role perception, perceived level of control, gender, marital status
and living situation, but these are not related to the referring process.²² These features are still too global, however, to estimate the appropriateness of interventions in relation to personal needs, characteristics, and preferences of person with dementia-caregiver dyads, and are mainly related to the caregiver, not to the person with dementia. Hence, referrers need more detailed information to determine the best person-centred approach for a specific dyad and match individual needs, characteristics, and preferences of a dyad with an intervention.

This study focused on: the Pleasant Events Program²³, the Exercise and Support Intervention for People with Dementia Living at Home and Their Caregivers²⁴ and Occupational Therapy According to the Community Occupational Therapy in Dementia Guideline (COTID)²⁵. These interventions make use of various activities, offer psycho-education, and provide emotional care that help the dyad adapt to the effects of dementia in their daily lives. The person with dementia and the caregiver practise these activities together. The interventions have a basic structure to start with and are adjusted to the dyad's wishes and needs. The Pleasant Events Program provided support in adapting and planning pleasant activities, e.g. learning to use skype with children and grandchildren using a stepwise manual, occasionally taking public transport to a different part of the city when going shopping instead of going the usual shops, watching National Geographic on television or taking a walk in the neighborhood. The Exercise and Support Intervention contained exercises for flexibility, balance, strength and/or endurance exercises using a ball, weights and elastic bands and combined this physical exercise with support for planning pleasant activities, as well as psycho-education and communication skill-training. Occupational Therapy consisted of the improvement of self-care over a broad spectrum, such as learning to use the remote control for the television with orientation marks, going on a bicycle-tour that is worry-free for the informal caregiver at home by taking a well-known route and using a mobile phone, helping to keep a daily structure and schedule with an individually adapted agenda. All three are short-term interventions consisting of 6-10 home visits.

A qualitative study revealed five factors important for the fit of three dyadic, activating interventions to the needs, characteristics and preferences of dyads.²⁶ These factors may be helpful for referrers, but are still broad. Therefore we developed a set of indicators building on the factors found in the qualitative study.²⁶ The research aim was to identify which of those indicators referring professionals in their clinical practice recognize in people with dementia and their caregivers.

Methods

Design

We used the RAND Appropriateness Method (RAM), which was developed to assess guality of care with the help of indicators.^{27,28} RAM defines appropriate care as 'the expected health benefit exceeds the expected negative consequences by a sufficiently wide margin that the procedure is worth doing, exclusive of cost'.²⁷ Although randomized control trials are seen as the golden standard for evidencebased care, they are often not available or do not meet the level of detail needed for a wide range of patients. Directed at agreement of a panel of experts, RAM combines expert opinion and empirical evidence. RAM has evidence of predictive validity of indicators.²⁹ Application of the method starts with drawing up a highly structured list of clinical indicators abstracted from existing scientific evidence.^{27,30} In two rounds, a number of experts rate those indicators for their appropriateness on a scale from 1-9. In the first round the experts rate the indicators in a survey. The individual ratings serve as input for a group discussion in which the content and the scope of the indicators are clarified. Both steps increase the validity of the final ratings in the second round, when the experts rate again the indicators individually. Then, appropriate indicators are selected on the basis of these second-round ratings. Appropriateness is based on agreement within the panel and scores in the top-third segment (7-9).²⁷ Indicators for which there was no agreement or which were assigned low ratings are discarded. The process results in a set of valid indicators (Figure 1).



Figure 1 Steps of study method (based on RAND Appropriateness Method)

Participants

We aimed to compose a panel of 15 experts, representing different health care organizations in our region, the greater Rotterdam urban area. In the Netherlands geriatricians of a memory clinic as well as general practitioners disclose the diagnosis of dementia. Next they serve as gatekeepers in the healthcare system because they refer to other services, including psychosocial and allied health services, and they know when prescribing or referring is appropriate. Usually a case manager is assigned to a person with dementia-caregiver dyad, who provides support and coordinates care with the geriatrician or general practitioner. Careservices for people with dementia and their caregivers like Alzheimer cafes, support-groups and day care are widely available in The Netherlands. Therefore we invited geriatricians, general practitioners as well as case managers with at least several years' experience with community dementia care. We wanted to include experts of the 16 health care organizations in the region. Via our professional network candidate panel members within those organizations were identified. They were sent an email with information about the study and an invitation to participate. Four of them agreed to participate themselves, others forwarded the invitation to colleagues, and one did not react. After having been send reminders by email and telephone, 14 experts of 13 health care organizations were willing to participate. Unfortunately for one organization the proposed data were not compatible with their planning.

Procedures

Step 1: Operationalization of factors into indicators (preparation)

This study builds on the results of a qualitative study exploring the fit of three dyadic, activating interventions to needs, characteristics, and preferences of people with dementia and their caregivers.²⁶ For the present study, the research team (NL, JL, EP, AMP) operationalized the five factors in observable indicators based on the qualitative descriptions elicited from interviews. Each indicator describes a need, characteristic or preference of the person with dementia and/ or the caregiver. A number of indicators concern characteristics such as the phase of dementia, awareness of the impact of the disease on daily life, and openness to a change in routine. Other indicators concern lifestyle characteristics such as doing sports and making outings. Yet other indicators describe needs for maintaining activities, needs for daily pastimes or structure, physical activity or self-sufficiency, and needs for adjustment to physical limitations and preferences for doing activities apart or together. The research team thoroughly discussed clarity

and consistency of these indicators and then developed 31 conceptual indicators (Appendix, round 1).

Step 2: Expert-panel rating of indicators in an online survey (data collection and analysis)

The experts received a link to an online survey which also included a description of the interventions and an abstract of the findings of the qualitative study. They were asked to rate the indicators in the survey on a scale from 1-9 for the question: 'For what percentage of the clients (people with dementia and their caregivers) in your clinical practice do you recognize the presence or absence of this indicator?' The rating scale is given in Table 1. The experts were additionally asked to suggest rewordings or other clarifications of items. The research team collected comments and scores of the online survey from all respondents and computed the median for each indicator (Appendix, round 1).

Table 1 Rating scale

Score*	1	2	3	4	5	6	7	8	9
Percentage of clients	0%	10%	25%	40%	50%	60%	75%	90%	100%
In words	none of the clients	some	a small part	some less than half	half	some more than half	most	almost all	all clients

^oA score of 1 means that it is not at all possible to recognize the presence or absence of this indicator in any client, signifying that this indicator is not accurate to recognize a certain need, characteristic or preference. A score of 9 means that it is possible to recognize the presence or absence of this indicator in all clients in their clinical practice, thus being a highly recognizable indicator 27.

Step 3: Group meeting (data collection and analysis)

The experts were encouraged to clarify their ratings in a group meeting. This step is designed to establish whether discrepant ratings are due to real disagreement in clinical judgment or to different interpretations of items. One expert who had agreed to participate and rated the survey could not attend the group meeting due to illness.

At the start of the meeting the panelists received a summary of the scores they had assigned in the survey and the median score of all experts for each indicator. Each indicator was discussed for recognizability in practice. The moderator (JL) first asked respondents if the indicator was clear and then invited them to clarify their scores in the online survey, especially those that diverged from the median. The arguments for a lower or a higher score than the median often led to a discussion. When necessary, indicators were clarified by the researcher (NL) using examples from the original qualitative study and sometimes reworded upon agreement within the panel. The discussion was audio-taped, and notes were made. The audio-tape was transcribed to collect relevant comments on the scoring of the indicators.

Step 4: Expert-panel rating indicators, second round (data collection and analysis) Upon discussion of sets of 4-6 conceptual indicators the experts individually rated each indicator again. For each indicator, we computed the median of these scores. Our panel finally consisted of an even number of panelists. Therefore, the median could fall between two points; for example six scores under or up to 6 and six scores 7 and higher, thus at 6.5. In this case the indicator is still recognizable for a majority of clients, and we counted the median as a 7, and included the median in the higher appropriateness segment (7-9).²⁷

Next, we calculated agreement/disagreement. For a panel of 12 panelists RAM defines agreement as: 'a maximum of three panelists rating outside a three-point segment around the median', and disagreement as: 'four or more panelists rating outside the three-point segment around the median'.^{27,30} Furthermore RAM uses two definitions of agreement: a strict and a relaxed one. We used the relaxed definition, which means: the median is located in the three-point segment (7-9), but ratings can be scored outside the segment, e.g. the median of 7 falls in the top third segment (7-9), but ratings are 6,7,8. We choose to use this definition, because with ratings of 6 and 7 the indicator is still recognizable in a majority of the clients, thus relevant for estimating the appropriateness (Appendix 1, round 2). Based on these second scores, we could discern recognizable indicators.

Results

Participants

The panel consisted of seven case managers (all women) and five physicians (three geriatricians from a general hospital, an academic hospital, and a nursing home, respectively, and two general practitioners; two men and three women). Their mean age was 50 years, and they had had on average 15 years' experience with dementia care.

Textual adaptations to the wording of the indicators

Eight of the 31 indicators were reformulated. For example the indicator 'the dyad wants to maintain the current situation' was changed into: 'the dyad wants to *strongly* maintain their current *way of living*'. A further 13 indicators were adapted by adding 'is/is not' or an adjective, like in the indicator 'the person with dementia *is/ is not* accustomed to spend time alone', and 'the person with dementia likes to have *enjoyable* shared activities with the caregiver'. Descriptions of the indicators, scores in rounds one and two, and all textual adaptations are given in the Appendix.

Appropriate and discarded indicators per factor

Factor I. Timing and openness for change (eight indicators).

The experts agreed on the recognizability of three indicators for circa 75% of their clients (I.3, I.5, I.6) (Table 2). These indicators concerned accepting change to cope with dementia. Five indicators were discarded because of disagreement between panelists (I.8) and low scores on recognizability (I.1, I.2, I.4, I.7). The indicators about being sufficiently informed were discarded because the words 'sufficient' or 'enough' were judged not clear enough. The indicators concerning the capacities of the person with dementia to establish new routines through repetition and energy of the caregiver to cope with a new approach were still found too subjective and difficult to check (Table 2). The experts were not able to estimate a person with dementia's capacity 'to establish new routines through repetition, and adapting manners to perform an activity or use memory-aids'. For example they were not able to estimate the feasibility of advices for memory-aids and indicated they did not have time for memory training. Therefore some of them discouraged the use of memory-aids for their clients. The experts disagreed about recognizing energy and burden of a caregiver. Some panelists were reluctant to trust what caregivers told about burden, feeling that caregivers tend to overrate themselves.

Factor II. Need for activities (four indicators).

All four indicators were recognizable in daily practice for circa 75%-90% of the clients.

Factor III. Lifestyle (five indicators).

The experts rated three indicators as recognizable in ±75% of their clients, namely those about an active lifestyle, used to sport and making outings (III.1, III.4, III.5). Two indicators were discarded because of disagreement (III.3) and a low score (III.2) on the recognizability of a preference for physical activity (Table 2). The experts

commented that assessing a preference for physical activity or sports was currently not included in the intake interview or questionnaire.

Factor IV. Apart or Together (six conceptual indicators).

The experts scored four indicators as recognizable for ±75%-90% of their clients (IV.2, IV.3, IV.5, IV.6). These indicators were about the need to spend time together. Two indicators were discarded because of low scores on recognizability, including the caregiver's preference to do enjoyable shared activities with the person with dementia (IV.1, IV.4) (Table 2).

Factor V. Meaning of activities (eight conceptual indicators).

Four indicators of this theme were recognizable for circa 75% - 90% of their clients, including the need for self-sufficiency and safety (V.4, V.6, V.7, V.8). Four indicators, concerning activity as a means for pastime, daily routine, and social contacts, were discarded because of disagreement (V.1, V.2, V.3, V.5). There was disagreement about the recognizability of the indicators related to the meaning or goal of the activities for people with dementia, such as passing time, physical activity, having social contact and positive experiences. Some panelists told that their needs assessment was influenced through their own knowledge about the importance of activities and social contacts for people with dementia and caregivers. One indicator about preference for physical activity was scored low and there was disagreement on one about the meaning of an activity (Table 2). The experts argued that in the current situation these indicators were not recognizable, but should be easy to check if included in intake interview or questionnaire. In addition the experts mentioned that they habitually assessed the appropriateness of support services available in their region.

Furthermore the panelists commented that seeing a dyad several times and meeting clients at home helped recognizing the indicators. Safety in and around the house was a regular topic for the case managers, but was seldom addressed by general practitioners and geriatricians in their consultations.

Overall the panelists rated 18 of the 31 conceptual indicators as recognizable in 75%-90% of the clients, with the relaxed definition of agreement according to RAM.

Table 2 Recognizable and discarded indicators

Factor I. Timing and openness for change	Appropriate	Discarded	Score
I.1 The dyad is sufficiently informed about the consequences of dementia for daily life		x	L
1.2 CG has an understanding of the consequences of dementia for the daily activities of the person with dementia		x	L
I.3 The dyad has a pro-active attitude, they want to anticipate on future situations	x		
I.4 The dyad wants to actively counteract decline as much as possible		x	L
I.5 The dyad is not focused on limitations, but on possibilities	x		
I.6 The dyad wants to strongly maintain their current way of living	x		
I.7 PwD still has the capacity to establish new routines through repetition (e.g. use a memory-aid)		x	L
I.8 CG is able to put energy into coping with a new approach, is not overburdened		x	D
Factor II. Need for activities			
II.1 PwD has a need for a meaningful occupational routine	x		
II.2 CG has a need for advice about how to cope with the behavior of the PwD	x		
II.3 CG has a need for support in how to assist or instruct the PwD to perform activities	x		
II.4 CG has a need for more insight into the capacities of the PwD, what he or she is able to do	x		
Factor III. Lifestyle			
III.1 The dyad or one of them (PwD or CG) has or had an active lifestyle	x		
III.2 PwD and/or CG likes physical activity		x	L
III.3 PwD and/or CG likes doing sports, in an institutional (group) setting or as a routine (running, bicycling)		x	D
III.4 PwD and/or CG is used to doing sport	x		
III.5 PwD and/or CG likes outings like shopping or making a visit	x		
Factor IV. Apart or together			
IV.1 PwD is/is not accustomed to spend time alone		x	L
IV.2 PwD depends a lot/limited on CG during the day	x		
IV.3 PwD likes to have enjoyable shared activities with CG	x		
IV.4 CG has a need for enjoyable shared activities with PwD		x	L
IV.5 CG has a strong/limited need for his or her own activities	x		
IV.6 CG has a need for more time for his or her own life	x		

> [Factor V. Meaning of activities	Appropriate	Discarded	Score
	V.1 PwD has a strong/limited need for something to do for passing time		x	D
	V.2 PwD has a strong/limited need for physical activity		x	D
	V.3 PwD has a strong/limited need for social contacts		x	D
	V.4 PwD has a strong/limited need for self-sufficiency	x		
	V.5 PwD has a strong/limited need for positive experiences		x	D
	V.6 TPwD will benefit from adaptations or assistive devices for physical limitations	x		
	V.7 CG has a need for advice about safety at home	x		
	V.8 CG has a need for advice about safety outside	x		

PwD: person with dementia; CG: caregiver; L= low scores (median in 1-3 or 4-6); D= disagreement

Discussion

>

The indicators identified in this study can help referrers estimate the appropriateness of activating interventions for people with dementia and their caregivers. The indicators related to needs and preferences for self-sufficiency, adjustment to physical limitations, and activities either apart or together are useful to assess the appropriateness of the Occupational Therapy intervention. Indicators related to the Pleasant Events Program and the Exercise and Support Intervention were not recognizable sufficiently, according to the panel. Thus, the valid indicators do not allow discerning which of the three available interventions would be most appropriate. The panellists found some indicators not recognizable because these are not part of a needs assessment in usual care. The panelists stated that indicators about a preference for physical activity, doing sports and about a need for activities that afforded social contacts were useful and should be recognizable if included in a standard needs assessment. When added, the list of indicators could permit assessing the appropriateness of one of the three interventions in particular. Other discarded indicators dealt with the need or preference of a person with dementia to spend time alone as well as the need of a caregiver to have enjoyable shared activities with the person with dementia. Finally, the indicators about capacities of a person with dementia to establish new routines through repetition, necessary to use adapted ways of performing activities, and about energy of the caregiver to cope with a new approach, necessary to experience with different approaches in these activating interventions were not recognizable for referring professionals. Professionals involved in the interventions can explore those needs, preferences and capacities during the start of the intervention.

Needs assessment of both people with dementia and caregivers is gaining ground, but is still in its infancy, although it might contribute to a person-centred approach once it has been more developed. Our panellists found it challenging to make latent needs of a dyad explicit, as also has been found in earlier research.⁵ Furthermore they recognized a trend in their needs assessment towards a more service-directed approach instead of a need-driven approach; they anticipated on frequently used and easy available support-services. Working in a routine fashion did not contribute to personalized support for people with dementia and caregivers.

It appeared that not only the quality of needs-assessment but also referral to activating interventions in regular care settings can be improved. Interventions may not have been available or the panelists were not aware of interventions being available.^{12,16-18} This was illustrated by the panellists who discouraged the use of memory-aids for people with dementia. Referral to skills training, which is part of these activating interventions would be a valuable option in this case. Because our panel consisted of highly expert clinicians who were open for these interventions, we expect that the use of these interventions and inter-professional collaboration in this field is still more limited in care as usual. Implementation of activating interventions enhancing strengths of the dyad has not been achieved yet, despite existing evidence, albeit heterogeneous.^{20,31}

Strengths and weaknesses of the study

The indicators are based on the results of a former qualitative study, and they are rooted in the reported data of people with dementia, caregivers and professionals, related to their lived experiences, using their words and expressions.²⁶ Next, the participating experts were open for discussion, and had long-standing experience of caring for people living with dementia. They represented a wealth of clinical experience in different care settings and care institutions.

Needs, characteristics and especially preferences are culturally sensitive. The respondents in the qualitative study were mostly native Dutch people. Cultural attitudes towards dementia and the caregiving role have implications for how people with dementia and CGs can best be supported.^{32,33} Hence, these indicators might be different for people with dementia and CG from other ethnic populations in the Netherlands.

Next, indicators cannot simply be transferred between countries.³⁴ The healthcare system and care pathways for dementia affect the availability of interventions and the role of referring professionals who are in the lead for supporting people with dementia and CG after the diagnosis of dementia is disclosed. Furthermore we performed this study in one, urban, area, which possibly limits the representative-ness for other regions.

Conclusion

The aim of this study was to develop a set of indicators with which referring professionals can estimate the appropriateness of dyadic activating interventions for people with dementia and for their caregivers. Eighteen of the 31 proposed indicators (58%) were found recognizable in 75%-90% of the clients, both people with dementia and the CG. These 18 indicators evaluate (aspects of) openness to change; need to maintain activities; former occupational performance and lifesty-le; need for activities apart or together; and needs for self-sufficiency, assistive devices for physical limitations and safety in and around the house.

Implications for research

Further study is needed to evaluate if the valid indicators found in this study, including indicators for physical and social activity, are appropriate to identify the dyads that will benefit the most from these interventions, either apart or together, as well as the practical use of these indicators for referring professionals. If so, evaluation of the use for people with dementia and caregivers from different cultural backgrounds is needed.

Implications for practice

We recommend clinicians to include these indicators in their regular assessment, and add questions about the need for activities too. These indicators can be helpful for assessing needs, characteristics and preferences of a dyad, thus contributing to a person-centred approach to provide tailored advice, especially related to dyadic activating interventions.^{7,8} The indicators offer geriatricians, general practitioners and case managers guidance to discuss the use of these interventions with a dyad in a shared decision-making process. Inter-professional collaboration in integrated care pathways may stimulate the use of activating interventions.³⁵ Professionals involved in performing the intervention programs, like occupational therapists and others, can use these indicators to inform referrers for what kind of couples and which needs these interventions might be appropriate.

References

- 1. Prince M, Wimo A, M. G, et al. The Global Impact of Dementia, An analysis of prevalence, incidence, cost and trends. London: Alzheimer's Disease International, 2015.
- 2. Ablitt A, Jones GV, Muers J. Living with dementia: a systematic review of the influence of relationship factors. *Aging & Mental Health*. 2009;13(4):497-511.
- 3. de Vugt ME, Stevens F, Aalten P, et al. Do caregiver management strategies influence patient behaviour in dementia? *International Journal of Geriatric Psychiatry*. 2004;19(1):85-92.
- 4. Peeters JM, Van Beek AP, Meerveld JH, Spreeuwenberg PM, Francke AL. Informal caregivers of persons with dementia, their use of and needs for specific professional support: a survey of the National Dementia Programme. *BMC Nursing*. 2010;9:9.
- van der Roest HG, Meiland FJM, Maroccini R, Comijs HC, Jonker C, Dröes RM. Subjective needs of people with dementia: a review of the literature. *International Psychogeriatrics*. 2007;19:559-592.
- Van Mierlo LD, Van der Roest HG, Meiland FJ, Dröes RM. Personalized dementia care: proven effectiveness of psychosocial interventions in subgroups. *Ageing Research Reviews*. 2010;9(2):163-183.
- Brooker D, Latham I. Person-Centred Dementia Care, Making Services Better with the VIPS Framework. Second Edition: ed. London: Jessica Kingsley Publishers, 2016.
- 8. Kitwood TM. Dementia Reconsidered: The Person Comes First Buckingham UK: Open University Press, 1997.
- 9. Mahler M, Meerveld JH. Care pathway Dementia [Zorgstandaard Dementie]. Utrecht: Vilans, 2013.
- 10. Samsi K, Manthorpe J. Care pathways for dementia: current perspectives. *Clinical Interventions in Aging*. 2014;9:2055-2063.
- Nygard L. Responses of persons with dementia to challenges in daily activities: a synthesis of findings from empirical studies. *American Journal of Occupational Therapy*. 2004;58(4):435-445.
- O'Sullivan G, Hocking C. Translating action research into practice; seeking occupational justice for people with dementia. OTJR: Occupation, Participation and Health. 2013;33(3):168-176.
- 13. Öhman A, Nygard, L. Meanings and Motives for Engagement in Self-Chosen Daily Life Occupations Among Individuals With Alzheimer's disease. OTJR: Occupation, Participation and Health. 2005;25(3):89-97.
- 14. Vikstrom S, Josephsson S, Stigsdotter-Neely A, Nygard L. Engagement in activities: Experiences of persons with dementia and their caregiving spouses. *Dementia: The International Journal of Social Research and Practice*. 2008;7(2):251-270.

- 15. Vikström S, Borell L, Stigsdotter-Neely A, Josephsson S. Caregivers' Self-Initiated Support Toward Their Partners With Dementia When Performing an Everyday Occupation Together at Home. *OTJR: Occupation, Participation and Health.* 2005;25(34):149-159.
- Aminzadeh F, Molnar FJ, Dalziel WB, Ayotte D. A review of barriers and enablers to diagnosis and management of persons with dementia in primary care. *Canadian Geriatrics Journal*: 2012;15(3):85-94.
- 17. Boots LM, Wolfs CA, Verhey FR, Kempen GI, de Vugt ME. Qualitative study on needs and wishes of early-stage dementia caregivers: the paradox between needing and accepting help. *International Psychogeriatrics*. 2015;27(6):927-936.
- Martin F, Turner A, Wallace LM, Choudhry K, Bradbury N. Perceived barriers to self-management for people with dementia in the early stages. *Dementia: The International Journal of Social Research and Practice*. 2013;12(4):481-493.
- Moniz-Cook E, Vernooij-Dassen M, Woods B, Orrell M. Psychosocial interventions in dementia care research: the INTERDEM manifesto. Aging & Mental Health. 2011;15(3):283-290.
- 20. Van't Leven N, Prick AE, Groenewoud JG, Roelofs PD, de Lange J, Pot AM. Dyadic interventions for community-dwelling people with dementia and their family caregivers: a systematic review. *International Psychogeriatrics*. 2013;25(10):1581-1603.
- Van't Leven N, Graff MJ, Kaijen M, de Swart BJ, Olde Rikkert MG, Vernooij-Dassen MJ. Barriers to and facilitators for the use of an evidence-based occupational therapy guideline for older people with dementia and their carers. *International Journal of Geriatric Psychiatry*. 2012;27(7):742-748.
- 22. Van Mierlo LD, Meiland FJ, Van Hout HP, Dröes RM. Toward an evidence-based implementation model and checklist for personalized dementia care in the community. *International Psychogeriatrics*. 2016;28(5):801-813.
- 23. Verkaik R, Francke A, Lahr M, Zuidervaart M. Huisbezoeken; Voorkom Somberheid bij Geheugenproblemen, handboek voor de hulpverlener (plezierige activiteitenmethode). [Home visits: Preventing depressive mood in memory problems, manual for the professional: Pleasant activities]. retrieved from https://www.nivel.nl/nl/publicatie/huisbezoeken-voorkom-somberheid-bij-geheugenproblemen-handboek-voor-hulpverlener-en. Utrecht: Nivel;2005.
- 24. Prick AE, de Lange J, Scherder E, Pot AM. Home-based exercise and support programme for people with dementia and their caregivers: study protocol of a randomised controlled trial. *BMC Public Health*. 2011;11:894.
- 25. Graff M, Melick Mv, Thijssen M, Verstraten P, Zajec J. Ergotherapie bij ouderen met dementie en hun mantelzorgers: Het EDOMAH-programma. [Occupational Therapy for Older Persons with Dementia and their informal Caregivers: COTiD-programme]. Houten: Bohn Stafleu van Loghum, 2010.

- 26. Van't Leven N, de Lange J, Prick AE, Pot AM. How do activating interventions fit the personal needs, characteristics and preferences of people with dementia living in the community and their informal caregivers? *Dementia: The International Journal of Social Research and Practice.* 2016.
- 27. Fitch K, Bernstein SJ, Aguilar MD, Burnand B, LaCalle JR, Lazaro P, ... Kahan JP. The RAND/ UCLA Appropriateness Method User's Manual. RAND CORP MONICA CA., 2001.
- 28. Shekelle P. The appropriateness method. *Medical decision making: an international journal of the Society for Medical Decision Making*. 2004;24(2):228-231.
- Campbell SM, Braspenning J, Hutchinson A, Marshall MN. Research methods used in developing and applying quality indicators in primary care. *British Medical Journal*. 2003;326(7393):816-819.
- 30. Basger BJ, Chen TF, Moles RJ. Validation of prescribing appropriateness criteria for older Australians using the RAND/UCLA appropriateness method. *British Medical Journal open*. 2012;2(5).
- 31. Brodaty H, Arasaratnam C. Meta-analysis of nonpharmacological interventions for neuropsychiatric symptoms of dementia. *American Journal of Psychiatry*. 2012;169(9):946-953.
- 32. Bunting M, Jenkins C. Transcultural nursing strategies for carers of people with dementia. Nursing Older People. 2016;28(3):21-25.
- 33. Lawrence V, Murray J, Samsi K, Banerjee S. Attitudes and support needs of Black Caribbean, south Asian and White British carers of people with dementia in the UK. British Journal of Psychiatry. 2008;193(3):240-246.
- 34. Van der Ploeg ES, Depla MFIA, van Hout HPJ, Nijpels G, Rigter H, Mackenbach JP. Care for vulnerable older persons: need, utilization and appropriateness (PhD thesis). Rotterdam, Erasmus University Rotterdam, 2009.
- 35. Döpp CM, Graff MJ, Olde Rikkert MG, Nijhuis van der Sanden MW, Vernooij-Dassen MJ. Determinants for the effectiveness of implementing an occupational therapy intervention in routine dementia care. *Implemention Sciences*. 2013;8:131.

Scores by experts for conceptual indicators to estimate the appropriateness of activating interventions Appendix.

	Scores round 1: Online survey									Scores round 2: After group meeting										
	Indicators	score						-		Indicators Changed so	ore							ā	₅	
		1 2	r M	4 5	و	~	ۍ ۵	< > ~	4/D/ //0/1	ri Minor Major	2 3 4	4	و	~	00 00	₹`> ¢				
	l. Timing									. Timing										
80	The dyad has clarity about diagnosis dementia			1	2	7	9	>	; 90%	The dyad is sufficiently informed about the x consequences of dementia for daily life		1 5	7	7	7	ö	60%	ъ	m	
	The dyad is aware of the consequences of dementia for their daily life	1	-	1 3	4		5		٥	CG has an understanding of the consequences of x dementia for the daily activities of the person with dementia	2	16	г	2		ö	50%			
	The dyad has a pro-active attitude, they want to anticipate on future situations		4	2 2		2	-		٥	The dyad has a pro-active attitude, they want to anticipate themselves on future situations x		2 2	2	'n	-	 >	75%			
	The dyad wants to actively counteract decline			۵ ۲	4	н		0'	2:50%	The dyad wants to actively counteract decline x as much as possible	1	m	ы	Μ		ö	60%			
	The dyad is not focused on limitations, but on possibilities in the current situation (like a course)		. 2	2 4	1 2	н	-		٥	The dyad is not focused on limitations, but on x possibilities	н	ŝ	7	ĥ	г	ÿ	75%			
	The dyad wants to maintain the current situation	1	-	1 3	-	Μ	7		۵	The dyad wants to <i>strongly</i> maintain their x current <i>way of living</i>		2	ς Υ	m	4	÷	75%			
	PwD still has the capacity to cut out with routines (e.g. use a memory-aid)	н	, ,	4 2	5	н	-		۵	PwD still has the capacity to establish new x routines through repetition (e.g. use a memory-aid)	2 4	4		н		ö	40%			
	CG has enough energy to cope with a new approach, is not overburdened			2 4	-	ы			٥	CG is able to put energy into coping with a new x approach, is not overburdened		1		ъ			۵			

PwD: person with dementia; CG: caregiver Median in bold A= Agreement; V=Valid : Q=questionable : I=Invalid: (percentage) = recognizable in amount of patients D= Disagreement MI= Maintained Indicators; DI= Discarded Indicators

Scores round 1: Online survey									Scores round 2: After group meeting										
Indicators	score								Indicators	hanged scor	a							ā	Ā
	1 2	ہ س	4	9	~	00	~	A/D/ //Q/I		∼ ⊢ Minor Major	m	4	С	~	∞	6	A/D/ V/Q/I		
II. Need for activities									ll. Need for activities										
4 PwD has a need for structure and daily routine		н	2		Μ	ы	-	%06 :/	PwD has a need for a <i>meaningful occupational</i> routine	×	7	н		N	4	-	V: 75%	0	4
CG has a need for advice about how to cope with the behavior of the PwD			2		Q	7	-	/: 75%	CG has a need for advice how to cope with the behavior of the PwD				-	5	'n	1	V: 90%		
CG has a need for support in how to assist or instruct the PwD to perform activities			г	m.	9	7	-	/: 75%	CG has a need for support in how to assist or instruct the PwD to perform activities				7	9	4	-	V: 75%		
CG needs more insight into the capacities of the PwD, what he or she is able to do			2	- 1	ы	4	-	/: 75%	CG has a need for more insight into the capa- cities of the PwD, what he or she is able to do				7	m	'n	2	V: 90%		
III. Lifestyle									III. Lifestyle										
5 The dyad or one of them (PwD or CG) has or had an active lifestyle	Ч		2	5	9	ы	/	/: 75%	The dyad or one of them (PwD or CG) has or had an active lifestyle				5	D	m	-	V: 75%	7	ŝ
PwD and/or CG deem physically activity important	2		4	N	Μ	н	0	2: 60%	PwD and/or CG likes physical activity	×		-	4	~ •	-	0	Q: 60%		
PwD and/or CG likes doing sports	4	-	1		2	н		٥	PwD and/or CG likes doing sports, in an institutional (group) setting or as a routine (trunning, bicycling)	×	2		2 1	m 	4		۵		
PwD and/or CG is used to sport	m		m	5	7			۵	PwD and/or CG is used to doing sport		ы			4	4	-	V: 75%		
PwD and/or CG likes outings, like shopping or making a visit		н	7	-	'n	Μ	-	/: 75%	PwD and/or CG likes outings like shopping or making a visit				-	∞	m	-	V: 75%		
M.D. sources with domostics. (C. sources)																			

PwD: person with dementia; CG: caregiver Median in bold

A= Agreement; V= Valid ==: Q=questionable ==: I=Invalid: (percentage) = recognizable in amount of patients D= Disagreement MI= Maintained Indicators; DI= Discarded Indicators

Scores round 1: Online survey								Scores round 2: After group meeting					
Indicators	score							Indicators Changed score					Ę
	1 2 3	8	ъ	9	7 8	⊀ ≻ ٥	/0/ //0/	R I Minor Major	4 5 6 7 8	6	A/ D / V/Q/I	:	:
IV. Apart or together								IV. Apart or together					
6 PwD is accustomed to be on his own	4 2	4			-	<u></u>	40%	PwD is/is not accustomed to spend time alone x 2 :	5 1 1		Q: 40%	7	+
PwD depends on CG during the day					3 1	>	%06:	PwD depends a lot/ limited on CG during the day x	83		V: 90%		
PwD likes to have shared activities with CG	г			2	5 2	>	: 75%	PwD likes to have <i>enjoyable</i> shared activities with CG x	1 8 3		V: 75%		
CG has a need for shared activities with PwD		2	4	F	4 1	0'	: 60%	CG has a need for <i>enjoyable</i> shared activities with PwD x	4 2 4 1		Q: 60%		
CG has a need for own activities	г	Г	Ч	7	2 5	>	: 75%	CG has a strong/limited need for his or her own activities	1 1 2 6	_	V: 90%		
CG has a need for more time for his or her own life	-	. 1	7	-	2		۵	CG has a need for more time for his or her own life	e M M	-	V: 90%		
V. Meaning of activities								V. Meaning of activities					
8 PwD has a need for something to do for passing time		г	m	7	~ ~		<u>م</u>	PwD has a strong/limited need for something to do for x	1 2 4 2 2		۵	4	+
PwD has a need for physical activity	2 1	- 2	m	7	2		٥	PwD has a strong/limited need for physical activity x	2 2 3 3		٥		
PwD has a need for social contacts	ч	Ч	Μ		~ ~		٥	PwD has a strong/limited need for social contacts x	1 3 1 4 3		٥		
PwD has a need for self-sufficiency	5	2		4	2 2		٥	PwD has a strong/limited need for self-sufficiency x	1 3 5 3		V: 75%		
PwD has a need for positive experiences	1 4 1	-	7	-	2		۵	PwD has a strong/limited need for positive experiences x 4 :	1 3 1 2		٥		
PwD has a need for adaptations or assistive devices for physical limitation	1	3 1	Ч	N	1 3		۵	PwD will benefit from adaptations or assistive devices for x physical limitations	2 4 6		V: 90%		
CG has a need for advice about safety at home			2	°,	4	>	: 75%	CG has a need for advice about safety at home	1 1 1 6 2	1	V: 75%		
CG has a need for advice about safety outside				m m	4	>	: 75%	CG has a need for advice about safety outside	1 6 5		V: 75%		

PwD: person with dementia; CG: caregiver Median in bold A= Agreement; V=Valid ==: Q=questionable ==: I=Invalid: (percentage) = recognizable in amount of patients D= Disagreement MI= Maintained Indicators; DI= Discarded Indicators

CHAPTER 6 -

Exploring the usefulness of indicators for referring people with dementia and their informal caregivers to activating interventions: a qualitative analysis of needs assessments.

Published as:

Van't Leven N, de Lange J, Groenewoud JH, Ploeg ES van der, Pot AM. Exploring the usefulness of indicators for referring people with dementia and their informal caregivers to activating interventions: a qualitative analysis of needs assessments. *BMC Geriatrics*, 2019; Article number: 230, 1-7. doi:10.1186/s12877-019-1221-0. Epub 2019 Aug 23.

Background

People with dementia and their informal caregivers frequently report difficulties in maintaining their usual activities. We had previously developed a set of indicators to estimate whether dyadic, activating interventions can meet these needs for activity. This study investigates how people with dementia and informal caregivers talk about the indicators in interviews for needs assessments, and how professionals identify activity needs and preferences. Our research goal was to explore the usefulness of the indicators for assessing the activity needs of community-dwelling dyads. Such assessments are needed for appropriate referral to activating interventions.

Methods

A dementia case manager assessed the needs of community-dwelling people with dementia and their informal caregivers; we carried out secondary analyses on the dataset resulting from the audio-tapes and transcripts. We applied qualitative, deductive content analysis because we wanted to identify both explicit and implicit needs and preferences. We used the indicators that we had developed in previous research as codes.

Results

Both people with dementia and informal caregivers do explicitly mention needs, preferences, and characteristics related to the indicators in the needs assessments. Possible implicit needs and preferences were frequently identified in their stories.

Conclusions

Needs-driven care requires high-quality needs assessments. Both people with dementia and their informal caregivers need encouragement to express their latent needs and preferences. In addition, latent needs and preferences have to be further explored in needs assessments to find out the real meaning. The out-comes of this study highlight the significance of structured needs assessments for mapping the activity needs of people with dementia and their informal caregivers. Many people with dementia and informal caregivers reported activity needs, which suggests that activating interventions may be appropriate. The indicators can help professionals identify activity needs so that they can discuss matching activating interventions with the dyad.

Background

People with dementia encounter all kinds of difficulties in maintaining their activities of daily living and social functioning. They need support and care to manage their everyday life.¹ Informal caregivers often provide most of the care and support for community-dwelling people with dementia.² They may have difficulties helping the people with dementia with daily activities and maintaining their own activities at the same time due to stress and a lack of time.^{2,3} The people with dementia and their informal caregivers often have difficulty adapting the daily routine to their changing capacities.⁴⁻⁶

Dyadic activating interventions aim to increase a dyad's skills to continue meaningful activities and cope with diminishing capacities.⁷⁻¹¹ The overall evidence for their efficacy is heterogeneous.^{12,13} Nevertheless, dyads who participated in these interventions, felt empowered by having been offered solutions for their most important activity needs. Such solutions came from in-depth assessments of their capacities and limitations.¹⁴ After disclosure of the diagnosis, people with dementia receive little advice about dyadic activating interventions that focus on maintaining daily functioning and social roles.^{15,16}

Dyadic activating interventions should match dyads' needs, characteristics, and preferences.^{17,18} Individual needs and preferences for support vary largely, depending on the personal setting, health and comorbidity, and the coping style of both people with dementia and their informal caregiver.¹⁹⁻²¹ In a qualitative study, we found that dyads who wish to remain active and are open to alternative ways of going about their daily activities, might profit from these interventions.²² Identifying activity needs, personal characteristics, and preferences is a prerequisite for estimating the appropriateness of interventions for a dyad.¹⁷ Because needs and preferences may be latent or expressed implicitly, probing questions can help reveal them.^{23,24}

To help professionals explore the activity needs, characteristics, and preferences of dyads, we developed a set of indicators for referral on the basis of the findings of the qualitative study. The set of 31 indicators was divided into five themes: 'Need for activities', 'Timing and openness', 'Lifestyle', 'Doing things apart or together' and 'Meaning of activities' (Table 1).

Table 1 Presence of indicators in needs-assessment

Indicators	Present
1. Need for activities	
 The PWD has a need for a meaningful occupational routine The CG has a need for advice how to cope with the behavior of the PWD The CG has a need for support in how to assist or instruct the PWD to perform activities The CG has a need for more insight into the capacities of the PWD, what he or she is able to do 2. Timing and openness for change. The dyad is informed sufficiently about the consequences of dementia for daily life 	14 9 6 1
 2 The CG has an understanding of the consequences of dementia for the daily activities of the PWD 3 The dyad has a pro-active attitude, they want to anticipate themselves on future situations 4 The dyad wants to counteract decline actively as much as possible 5 The dyad is not focused on limitations, but on possibilities 6 The dyad wants to strongly maintain their current way of living 7 The PWD still has the capacity to cut out with routines (e.g. use a memory-aid) 8 The CG is able to put energy into coping with a new approach, is not overburdened 	2 7 - 9 3 - 1
3. Lifestyle	
 The dyad or one of them (PWD or CG) has or had an active lifestyle The PWD and/or CG likes physical activity The PWD and/or CG likes doing sports, in an institutional (group) setting or as a routine (running, bicycling) The PWD and/or CG is used to sport The PWD and/or CG likes outings like shopping or making a visit 	8 15 6 5 9
4. Apart or together	
 The PWD is/is not accustomed to spend time alone The PWD depends a lot/limited on CG during the day The PWD likes to have enjoyable shared activities with CG The CG has a need for enjoyable shared activities with PWD The CG has a strong/limited need for his or her own activities The CG has a need for more time for his or her own life 	- 7 3 6 8 10
5. Meaning of activities	
 The PWD has a strong/limited need for something to do for passing time, See 1.1. The PWD has a strong/limited need for physical activity , See 3.2. The PWD has a strong/limited need for social contacts The PWD has a strong/limited need for self-sufficiency The PWD has a strong/limited need for positive experiences The PWD will benefit from adaptations or assistive devices for physical limitations The CG has a need for advice about safety at home The CG has a need for advice about safety outside 	14 15 5 7 - 8 2 2 2

PWD: person with dementia; CG: caregiver

The theme 'Need for activities' appeared to be the central theme, and the other themes provide guidance for determining the type of activities that may be most appropriate. The indicators of 'Timing and Openness' are related to acceptance and readiness for interventions that make use of adaptations of activities and environment and encourage changing daily routines. The 'Lifestyle' indicators facilitate choosing the type of activities that motivate the dyad. The indicators of 'Doing things apart or together' relate to preferences and routines of a dyad in their relationship and to the changing interdependence between the person with dementia and informal caregiver. The indicators concerning the 'Meaning of activities' may help in identifying the goals and objectives that are important to the dyad, such as having a fixed routine in the day, just passing time doing something at hand, being physically active, maintaining social contacts, maintaining independency, addressing safety inside and outside the house, or addressing physical limitations that hinder activities. A panel of clinicians recognised most of the indicators in their clinical practices.²⁵ The panel confirmed that needs assessment is a multi-layered process in which needs and preferences often have to be coaxed out of the dyad's stories.²⁵ To explore the usability of the indicators in clinical practice, the current study investigated how people with dementia and their informal caregivers talked about the indicators in needs assessments and how professionals identified these needs and preferences. Our research goal was to explore the usefulness of indicators in assessing the activity needs of communitydwelling people with dementia and their informal caregivers for appropriate referral to activating interventions.

Methods

Study design

We used a qualitative approach to identify needs, preferences, and characteristics as described in the indicators.²⁶ Latent needs and preferences can be interpreted from the stories told in the interviews.²⁷ For this study, we used data collected for the VitaDem project in our secondary analysis. The integral needs-driven approach developed in the 3-year VitaDem project (2015-2018) was intended to enhance functional independence and social inclusion of community-dwelling people with dementia and their informal caregivers.²⁸⁻³⁰ The purpose of the needs assessment in the VitaDem project was to reveal individual needs and the dyad's shared needs of self-sufficiency, vitality, and social inclusion. Needs to maintain daily activities were part of the semi-structured needs-assessment interviews, by dementia case managers. The level of education of the dementia case managers was comparable to higher professional education, completed with clinical experience (1-25 years). For this project they were trained to explore the stories of people with dementia and caregivers in depth and to avoid giving solutions before exploration. The people with dementia and caregivers were interviewed separately in their homes. A multi-disciplinary case conference followed for discussion of the principal needs and wishes of the dyad and brainstorming about the most appropriate tailored interventions, after clarification of these needs and wishes.

Ethical considerations

The Medical Ethics Review Committee of the Erasmus Medical Center, the Netherlands gave their ethical approval for the VitaDem project (MEC-2015-028). The people with dementia and informal caregivers consented to audio recording of the interviews and anonymous use of the data for research.

Data and study population

Our analysis included transcripts of the two interviews for each dyad and the first part of the case conference, in which the case manager answered questions to clarify dyads' needs. Complete data with the needs assessments of 20 dyads were available.

The case managers recruited dyads from their client databases. The inclusion criteria were the people with dementia had to have a diagnosis of dementia, both the person with dementia and their informal caregiver had to be 65 years old or older, and both had to be living in the case manager's practice area. The exclusion criteria were inability of the candidates to express themselves in Dutch or being on a waiting list for institutionalisation (Table 2).

	Mean age (range)	Men/Women	MMSE Mean (range)	Dyad-relation
Persons with dementia	81 years (68-89)	14 men/6 Women	21.5 (10-29)	19 spouses
Caregivers	76 years (48-84)	5 Men/15 Women		i mother-daughter

Table 2 Study population

MMSE: Mini Mental State Examination

There was an even distribution in education level, employment history and income in the study population. The needs-assessment interviews with persons with dementia lasted averagely 34 (12–80) minutes. The interviews with their informal caregivers lasted averagely 41 (18–75) minutes. The clarification part of the case conferences lasted averagely 15 (10–25) minutes. The interviews and case conferences took place between April 2015 and March 2017. The interviews and multi-disciplinary case conferences were transcribed verbally, anonymised, and imported into Atlas-ti 7.2 for qualitative analysis.

Data analysis

We used deductive content analysis to explore how the indicators appeared in the needs assessments.³¹ We carefully read three transcripts per dyad (two interviews and the clarification in the case conference), and we listened to the audio recordings of the interviews to familiarise ourselves with the stories. We sought text fragments that represented a need, characteristic, or preference related to the indicators and used the indicators as codes. The text fragment could present an explicitly mentioned present or absent need, characteristic, or preference, and it could refer to a latent need or implicit preference. Three researchers (NL, JL, and a trainee) independently coded the transcripts for the first dyad, and they discussed differences in linking text fragments to indicators. Then NL and the trainee coded the transcripts for 10 dyads. NL coded the remaining transcripts. We summarised the text fragments for indicators into a condensed description, illustrated with quotes. After analysing the needs assessments for 16 dyads, we discussed the preliminary results for their plausibility and consistency with the researchers involved in the VitaDem project (NL, JL, HG, and a researcher). To make the trustworthiness more rigorous, three researchers (JL, HG, and a researcher) coded the transcripts for one dyad independently to check the consistency with NL's coding. The three researchers coded nine of the ten indicators that NL had coded in that the same transcript, and we concluded that our coding was consistent to a great extent. We counted the presence of the indicators as the last step.

Results

People with dementia and their informal caregivers do mention needs, preferences, and characteristics related to the indicators in the needs-assessment interviews (Table 1). Most presentations of the indicators related to the former 'Lifestyle' were explicit and clear. The indicators related to the other themes, however, were more often implicitly present in the clients' stories. These needs and preferences should have had further exploration in the interviews to find out the real meaning. In total, we identified 27 of the 31 indicators.

The 'Need for activities' theme

The indicator describing the person's with dementia need to maintain activities was present in the needs assessments of 14 dyads. Some people with dementia said quite clearly that they wanted to keep doing what they were used to doing, and some mentioned specific activities, such as household activities or a hobby. A latent need for activity was recognised when a person with dementia said that he or she was bored or missed a daily structure. Other people with dementia said that everything was going fine; they did not know of anything more they could wish for. However, their excitement when talking about past activities might suggest a latent need for current activities. Caregivers had a need for advice about how to help the people with dementia. Some were irritated by the person's with dementia behaviour; for example, about inactivity or a person with dementia gardening for hours without a pause for a meal. These examples may indicate a need for help in learning how to assist the person with dementia in activities, although the case managers did not explore this topic further.

The 'Timing and openness' theme

Several dyads said that they gathered information about dementia, some has already adapted their daily routines, and some said that they took future changes into account. Rejecting domestic help because one prefers to do it one's own way can represent a strong wish to stick to one's current ways.

The 'Lifestyle' theme

The people with dementia and informal caregivers clearly detailed their former lifestyles: how they spent their days, what their usual activities were, and which outings or sports they preferred.

The 'Doing things apart or together' theme

Many informal caregivers clearly stated their need to maintain their own activities. Several dyads also mentioned the need for enjoyable shared activities. Some informal caregivers said they felt a lack of time for shared activities because the person with dementia attended day-care and had other appointments with care services.

The 'Meaning of activities' theme

Several people with dementia and informal caregivers explained clearly what the activities meant to them, and why the activities were important to them. However, the nature of the activities (e.g., social contact, being self-sufficient, or safety) was frequently unclear and difficult to identify from the stories in the needs assessments. One person with dementia said that he did not miss the gardening after he gave up his allotment garden, but when encouraged to tell more, he said he missed the mate he collaborated with. Another person with dementia said he wanted to continue driving; probing questions made it clear that driving a car meant for him that he could keep his role of helping his wife get the groceries.

Some indicators overlapped or were difficult to distinguish from each other. The indicators about the caregiver's need for advice about how to cope with behaviour (Table 1-1.2) and how to assist the person with dementia in activities overlapped (Table 1-1.3). The indicators concerning the meaning of activities for passing time (Table 1-5.1) and engaging physical activity (Table1-5.2) were difficult to distinguish from indicators belonging to the themes of 'Need for activities' (Table 1-1.1) and 'Active lifestyle' (Table 1-3.2).

Discussion

This study shed light on how community-dwelling people with dementia and their informal caregivers expressed their needs and preferences for maintain meaningful activities, and how case managers identified these needs and preferences. Our study confirms that dyads often have needs to maintain activities. However, they need encouragement to express and to explore those needs. We frequently identified activity needs and preferences that were implicit in their stories, which should have been further explored. Preferences related to the theme of 'Lifestyle' were expressed clearly in the needs assessments, but in the other four themes 'Need for activities', 'Timing and openness', 'Apart or together', and 'Meaning of the activity', we identified latent needs and preferences. The set of indicators with guiding questions is useful for searching actively and systematically for activity needs, including latent needs and their meaning.

Influence of the study setting: limitations and strengths

The transcripts of the needs assessments used for this study may not represent usual care. Improving tailored care and multi-disciplinary collaboration were important features in the VitaDem project. As a result, the needs assessment got more emphasis than it would in usual care. The dementia case managers were trained to conduct the needs-assessment interviews with a focus on maintaining daily life and ageing in place.

A strength of this study is the inclusion of needs assessment with people with dementia. Interviewing people with dementia for needs assessment requires a careful approach. Confidence between the person with dementia and the interviewer and the use of short sentences are important in interviewing people with dementia, and questions need to be specific. Questions about needs may be too abstract for them, so prompts and examples were often needed to formulate their experiences, needs, and preferences. Empathy and carefully following their thoughts were essential competencies for the quality of the needs assessments.

Implications for practice

In a person-centred approach, a professional is sensitive to tracing unmet needs, including activity needs ³². The professionals' expertise in client-centred interview skills is a prerequisite for revealing needs and preferences because people with dementia and their informal caregivers often express only the needs for which they know a solution exists. They need encouragement to explore latent needs.^{6,24,25} Our study indicates that explicit training in interview techniques is necessary for professionals in dementia care. However, when unmet needs are made explicit, a higher level of care that adds to health-related quality of life can be achieved.³² To meet a need adequately, it is also important to explore what that need or wish means to the person. The indicators concerning the 'Meaning of activities' may help identify the goals and objectives of a dyad, such as having a fixed routine in the day, just passing time doing something at hand, being physically active, maintaining social contacts, maintaining independency, addressing safety inside and outside the house, or addressing physical limitations that hinder activities. Knowing the meaning of the activities provides clues to finding alternatives if the preferred activity is no longer possible.

Activating interventions aim to increase a dyad's skills to adapt activities and make changes in the environment or daily routines, and they require readiness.³³ There-

fore, the set of indicators include questions about timing and openness for these activating interventions. In an early phase of dementia, people with dementia and their informal caregivers are very aware of social and occupational deficits. Nevertheless, assessment is often limited to medical and personal care domains in usual care.⁴ Dementia-care specialists plead for integral needs-based approaches for dementia, which are becoming more and more developed and implemented in clinical practice.³⁴⁻³⁶ This needs-based approach is also recommended in care pathways for dementia in several West European countries.^{37,38} With regard to activity needs, the potential benefit of activating interventions needs to be taken into account, especially in an early phase of dementia.

Implications for research

The set of indicators can be a starting point for studying implementation in clinical practice and their utility for professionals. Do they have an added value in exploring activity needs, a dyad's being ready for change, and estimating the appropriateness of activating interventions? If so, this approach to developing indicators can be applicable for other needs domains and interventions too.

Conclusions

Needs-based care can contribute to a higher level of care and quality of life for people with dementia and their informal caregivers, but it requires good-quality needs assessment. More attention to exploring needs and preferences seems necessary for professionals in dementia care. The indicators can help professionals assess activity needs so that they can better discuss the appropriateness of activating interventions with a dyad.

References

- 1. Livingston G, Sommerlad A, Orgeta V, et al. Dementia prevention, intervention, and care. *Lancet*. 2017;390(10113):2673-2734.
- 2. Peeters J, Van Beek A, Meerveld J, Spreeuwenberg P, Francke A. Informal caregivers of persons with dementia, their use of and needs for specific professional support: a survey of the National Dementia Programme. *BMC Nursing*. 2010;9(9).
- 3. Vikström S, Borell L, Stigsdotter-Neely A, Josephsson S. Caregivers' Self-Initiated Support Toward Their Partners With Dementia When Performing an Everyday Occupation Together at Home. OTJR: Occupation, Participation and Health. 2005;25(34):149-159.
- O'Sullivan G, Hocking C. Translating action research into practice: seeking occupational justice for people with dementia. OTJR: Occupation, Participation and Health. 2013;33(3):168-176.
- 5. Van Bruggen S, Gussekloo J, Bode C, Touwen DP, Engberts DP, Blom JW. Problems experienced by informal caregivers with older care recipients with and without cognitive impairment. *Home Health Care Services Quarterly*. 2016;35(1):11-24.
- 6. van der Roest HG, Meiland FJ, Comijs HC, et al. What do community-dwelling people with dementia need? A survey of those who are known to care and welfare services. *International Psychogeriatrics*. 2009;21(5):949-965.
- Gitlin LN, Winter L, Burke J, Chemett N, Dennis MP, Hauck WW. Tailored activities to manage neuropsychiatric behaviors in persons with dementia and reduce caregiver burden: A randomized pilot study. *American Journal of Geriatric Psychiatry*. 2008;16(3):229-239.
- Graff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM.
 Community based occupational therapy for patients with dementia and their care givers:
 Randomised controlled trial. British Medical Journal. 2006;333(7580):1196-1199.
- 9. Prick AE, de Lange J, Scherder E, Pot AM. Home-based exercise and support programme for people with dementia and their caregivers: study protocol of a randomised controlled trial. *BMC Public Health.* 2011;11:894.
- 10. Teri L, Gibbons LE, McCurry SM, et al. Exercise plus behavioral management in patients with Alzheimer disease: a randomized controlled trial. *JAMA*. 2003;290(15):2015-2022.
- 11. Verkaik R, Francke A, Lahr M, Zuidervaart M. Huisbezoeken; Voorkom Somberheid bij Geheugenproblemen, handboek voor de hulpverlener (plezierige activiteitenmethode). [Home visits: Preventing depressive mood in memory problems, manual for the professional: Pleasant activities]. retrieved from www.nivel.nl/publicaties. Utrecht: Nivel;2005.
- Olazaran J, Reisberg B, Clare L, et al. Nonpharmacological therapies in Alzheimer's disease: a systematic review of efficacy. *Dementia and Geriatric Cognitive Disorders*. 2010;30(2):161-178.

- 13. Van't Leven N, Prick AE, Groenewoud JG, Roelofs PD, de Lange J, Pot AM. Dyadic interventions for community-dwelling people with dementia and their family caregivers: a systematic review. *International Psychogeriatrics*. 2013;25(10):1581-1603.
- 14. Van't Leven N, de Lange J, Van der Ploeg E, Pot AM. Working mechanisms of dyadic, psychosocial, activating interventions for people with dementia and informal caregivers: a qualitative study. *Clinical Interventions in Aging*. 2018.
- Aminzadeh F, Molnar FJ, Dalziel WB, Ayotte D. A review of barriers and enablers to diagnosis and management of persons with dementia in primary care. *Canadian Geriatrics Journal*: 2012;15(3):85-94.
- Moniz-Cook E, Vernooij-Dassen M, Woods B, Orrell M. Psychosocial interventions in dementia care research: the INTERDEM manifesto. *Aging & Mental Health*. 2011;15(3):283-290.
- 17. Brooker D, Latham I. Person-Centred Dementia Care, Making Services Better with the VIPS Framework. Second Edition: ed. London: Jessica Kingsley Publishers, 2016.
- Kitwood TM. Dementia Reconsidered: The Person Comes First Buckingham UK: Open University Press, 1997.
- 19. Ablitt A, Jones GV, Muers J. Living with dementia: a systematic review of the influence of relationship factors. *Aging & Mental Health*. 2009;13(4):497-511.
- Borsje P, Hems MA, Lucassen PL, Bor H, Koopmans RT, PotAM. Psychological distress in informal caregivers of patients with dementia in primary care: course and determinants. *Family Practice*. 2016;33(4):374-381.
- 21. de Vugt ME, Stevens F, Aalten P, et al. Do caregiver management strategies influences patient behaviour in dementia? *International Journal of Geriatric Psychiatry* 2004;19:85-92.
- 22. Van't Leven N, de Lange J, Prick AE, Pot AM. How do activating interventions fit the personal needs, characteristics and preferences of people with dementia living in the community and their informal caregivers? *Dementia: The International Journal of Social Research and Practice.* 00 2019; 18(1): 157-177.
- 23. Boots LM, Wolfs CA, Verhey FR, Kempen GI, de Vugt ME. Qualitative study on needs and wishes of early-stage dementia caregivers: the paradox between needing and accepting help. *International Psychogeriatrics*. 2015;27(6):927-936.
- 24. van der Roest HG, Meiland FJM, Maroccini R, Comijs HC, Jonker C, Dröes RM. Subjective needs of people with dementia: a review of the literature. *International Psychogeriatrics*. 2007;19:559-592.
- 25. Van't Leven N, Van der Ploeg E, de Lange J, Pot AM. Indicators to estimate the appropriateness of activating interventions for people living with dementia and for their informal caregivers. *Aging & Mental Health*. 2017:1-8.
- 26. Hsieh HF, Shannon SE. Three approaches to qualitative content analysis. *Qualitative Health Research*. 2005;15(9):1277-1288.

- 27. Murray M. Narrative psychology and narrative analysis. In: Smith JA, ed. Qualitative Psychology, A practical guide to research methods second ed. London: Sage, 2008:111-132.
- 28. Groenewoud JH, Senhaji H, Van't Leven N, de Lange J. Barriers and facilitators to improve the social network of home-dwelling people living with dementia and their family caregivers. *Alzheimer's & Dementia*. 2017;13(7 (Suppl.)):1456.
- 29. Groenewoud JH, Van't Leven N, Senhaji H, Döpp C, de Lange J. Vitaal blijven en meedoen voor mensen met dementie en hun mantelzorgers; de VitaDem aanpak [Vitality and participation for people living with dementia and their informal caregivers: the Vitadem approach]. Rotterdam: Kenniscentrum Zorginnovatie, Hogeschool Rotterdam [Research Centre Innovations in Care, Rotterdam University of Applied Sciences] 2018.
- 30. Senhaji H, de Lange J, Groenewoud JH, Van't Leven N. The indicated needs of Dutch people with dementia to enable 'ageing in place'. *Alzheimer's & Dementia*. 2017;13(7Sup-pl):495-496.
- 31. Elo S, Kyngas H. The qualitative content analysis process. *Journal of Advanced Nursing*. 2007;62(1):107-115.
- 32. Scholzel-Dorenbos CJ, Meeuwsen EJ, Olde Rikkert MG. Integrating unmet needs into dementia health-related quality of life research and care: Introduction of the Hierarchy Model of Needs in Dementia. *Aging & Mental Health*. 2010;14(1):113-119.
- Gitlin LN, Rose K. Factors associated with caregiver readiness to use nonpharmacologic strategies to manage dementia-related behavioral symptoms. *International Journal of Geriatric Psychiatry*. 2014;29(1):93-102.
- Eichler T, Thyrian JR, Dreier A, et al. Dementia care management: going new ways in ambulant dementia care within a GP-based randomized controlled intervention trial. *International Psychogeriatrics*. 2014;26(2):247-256.
- 35. Reuben DB, Evertson LC, Wenger NS, et al. The University of California at Los Angeles Alzheimer's and Dementia Care program for comprehensive, coordinated, patient-centered care: preliminary data. *Journal of the American Geriatrics Society*. 2013;61(12):2214-2218.
- 36. Samus QM, Johnston D, Black BS, et al. A multidimensional home-based care coordination intervention for elders with memory disorders: the maximizing independence at home (MIND) pilot randomized trial. American Journal of Geriatric Psychiatry. 2014;22(4):398-414.
- 37. Mahler M, Meerveld JH. Care pathway Dementia [Zorgstandaard Dementie]. Utrecht: Vilans, 2013.
- 38. Samsi K, Manthorpe J. Care pathways for dementia: current perspectives. *Clinical Interventions in Aging*. 2014;9:2055-2063.

Appendix Indicators in the needs assessments and examples

Indicators	Present in NA (N=20)	Examples and quotes
1. Need for activities		
1** The PWD has a need for a meaningful occupational routine.	14	People said that they particularly missed some activities and wanted to continue them. Others needed something to do without mentioning anything specific, but wanted a more satisfying way to spend their time. PWD: 'I sit here so often with idle hands, and that's just not my nature.'
2° The CG has a need for advice how to cope with the behavior of the PWD.	9	Some CGs asked openly for advice. The need for advice was also implicit in statements about expecting too much of the PWDs, asking them to do things they could not do, or blaming them for falling short. CG: 'You [PWD] tell me that you liked your meal at the day-care centre. So why can't you remember what it was?' CM: 'She [PWD] is inactive and avoids activities. It's hard for her husband (CG), but he does not know how to activate her.'
3° The CG has a need for support in how to assist or instruct the PWD to perform activities.	6	CGs talked about needing advice for practical things like using a mobile phone or remote control, but also more generally, for how to activate the PWD, how to instruct. CG: 'When I was ill, I sometimes thought it would be so nice if someone brought me a cup of tea and a biscuit a few of those little things, you know. But I had to do it all myself.' PWD [her husband]: 'Yes, but I did. It isn't that I didn't want to, I just forgot. All those small things I didn't notice.' Other CG: 'She [PWD] is fanatic about gardening. She can start at n a.m., and last week I had to get her from the garden at 8.30 in the evening and I said, 'Come on, we're going get ourselves something to eat.' Other CG: 'She [PWD] is fanatic about gardening. She can start at n a.m., and last week I had to get her from the garden at 8.30 in the evening and I said, 'Come on, we're going get ourselves something to eat.'
4° The CG has a need for more insight into the capacities of the PWD, what he or she is able to do.	1	This indicator was difficult to discern from 1.2. CM: 'He [PWD] has a room for himself, for his hobbies, which is a mess. His wife wants him to clean it up, but he doesn't know where to begin.'
2. Timing and openness for change		
1 The dyad is informed sufficiently about the consequences of dementia for daily life.	4	CGs and some PWDs said that they had read about dementia and visited Alzheimer cafés. PWD: 'I feel that I forget things now. I have to watch out not to forget Yes, I work on it, I write it down.'
2 The CG has an understanding of the consequences of dementia for the daily activities of the PWD.	2	Some CGs recognized the consequences because they had other family members with dementia.
3** The dyad has a pro-active attitude, they want to anticipate themselves on futuresituations.	7	Examples were visiting the Alzheimer cafés or meetings for informal caregivers, searching for information (in books, and on internet), and adapting things in the house to minimalise the risk of falling. The dyad took future changes into account. CG: 'We think about what the next step can be. What will be the next thing for her [PWD] to hand in? What do we have to know? Where can we get it? Be prepared!'

Indicators	Present in NA (N=20)	Examples and quotes
2. Timing and openness for change		
4 The dyad wants to counteract decline actively as much as possible.	-	
5** The dyad is not focused on limitations, but on possibilities.	9	Dyads adapted their routines to cope with limitations. Examples were adapting activities, taking a course, using services for groceries, domestic help, and ordering articles online. The dyads also mentioned societal arrangements, e.g. for transport and personal alarms. CG: 1 bought an iPad for him. My daughter installed the dementia app and added photos. I have to help him turn it on and find what he wants' PWD adds: 'You just need toeh put your finger on it, and then the photo of one of my boys, or a photo of her is there [wife and children].'
6** The dyad wants to strongly maintain their current way of living.	3	Some dyads did not want to change their routines or could not. They continued their ways as before, despite the consequences of dementia. CM: 'I think she anxiously sticks to this daily program; this is what she can manage.' CG: 'I cancelled my domestic help. She never does it the way I want, and then I pay her a pile of money, while I do it myself better and quicker.'
7 The PWD still has the capacity to cut out with routines (e.g. use a memory-aid).	-	
8 The CG is able to put energy into coping with a new pproach, is not overburdened.	1	Some CGs said they could take decisions; others were overwhelmed and had little energy for starting an intervention. CG: 'I can take decisions firmly and all that I hope to continue this for the future. I fear the moment that I have an accident or something. Then it's game over, for him too [PWD].'
3. Lifestyle		
1 ^{**} The dyad or one of them (PWD or CG) has or had an active lifestyle.	8	The PWDs and CGs talked about their daily activities, now and in the past, painting a picture of an active lifestyle. They mentioned hobbies such as knitting, painting, making music, reading, and physical activity. Others talked about going out 'getting some fresh air every day', meeting their friends, going to the cinema or theatre. Some dyads would baby-sit their grandchildren, some were volunteers in the church or elsewhere.
2** The PWD and/or CG likes physical activity.	2	The PWDs cycled and went for walks. Some mentioned the importance of physical activity for their fitness and for delaying decline; others just liked a walk. Some dyads had in-house rooms for physical fitness
3° The PWD and/or CG likes doing sports, in an institutional (group) setting or as a routine (running, bicycling).	2	The participants mentioned regularly going to a fitness centre or a cycling club. They also liked regular swimming, golf, and/or yoga
4° The PWD and/or CG is used to sport.		Some PWDs and CGs mentioned various sports, such as football, tennis, running, triathlons, handball, and skating. It was also evident that others were not accustomed to sports. CG: 'I have never been a sportsman or anything like that.'

-	~	
-	-	

Indicators	Present in NA (N=20)	Examples and quotes
3. Lifestyle		
5** The PWD and/or CG likes outings like shopping or making a visit.	9	The activities mentioned varied: shopping, going to museums, dancing, going out for dinner, and playing bingo or cards. PWD: 'Just sitting around in the house is not my cup of tea .'
4. Apart or together		
1 The PWD is/is not accustomed to spend time alone.	-	
2 The PWD depends a lot/limited on CG during the day.	7	CGs had to manage the PWDs during the day, instruct them, and answer their questions what to do continuously. CGs assisted the PWDs with domestic tasks and, later, personal care as well. CM: 'He [PWD] becomes restless if he does not see or hear her.'
3* The PWD likes to have enjoyable shared activities with CG.	3	Only a few mentioned this: CM: 'They are happy with each other. They support each other a lot.' CG: 'Back in those days most of our activities were separate, but now we do a lot more together. I think my wife needs that because of the illness.'
4* The CG has a need for enjoyable shared activities with PWD.	6	CGs explained what they were accustomed to doing together because of shared interests or practical reasons (domestic tasks). Some missed time to spend together due to appointments. CG: 'Cinema, concerts, taking a walk with you [PWD] the prospect of some nice things that we can do together. I can look forward to that.'
5** The CG has a strong/limited need for his or her own activities.	8	CGs mentioned short activities such as leisurely going to a shop or doing a crossword puzzle – and also their hobbies that take time. CM: He [CG] is a tenor in a choir. He can manage the repetitions in the evening [leaving his wife alone at home], but a concert can take a whole day. He foresees that that will be difficult, and he has to make up his mind.'
6** The CG has a need for more time for his or her own life.	10	The CGs said that they did want to let the PWD be alone for a longer time. They felt rushed too often. Some said they simply needed 'peace'. Some CGs cancelled their volunteer work because they could not combine it with caring for their partners. CG: 'I fly to the shop in a hurry, and I fly back again.'
5. Meaning of activities		
1 The PWD has a strong/limited need for something to do for passing time.	14	See part 2.1
2 The PWD has a strong/limited need for physical activity.	15	See parts 3.2 and 3.3
3** The PWD has a strong/limited need for social contacts.	5	The PWDs talked about the market, the mall, the sports club, or a club for playing cards because they could meet their friends there and have a chat. PWD: I used to have mates whom I met regularly at the garden lots.'

Indicators	Present in NA (N=20)	Examples and quotes
5. Meaning of activities		
 4^{**} The PWD has a strong/limited need for self-sufficiency. 5 The PWD has a strong/limited need for positive experiences. 	7	The PWDs said that they wanted to do activities by themselves. This was important for their self-esteem. Some PWDs felt it was import for them to be able to do their share in domestic and household activities. PWD: 'I wanted to make the bed, but the sheet it did not work. It makes me sad. My husband has said it a thousand times: 'I can do it for you', but I don't want that. My answer is: 'I'll do it myself.'
6** The PWD will benefit from adaptations or assistive devices for physical limitations.	8	There were limitations in hand strength, balance, walking, and stairs. Some already had adaptations or assistive devices; others would benefit from them too.
7°° The CG has a need for advice about safety at home.	2	Some CGs did not want to leave the PWD alone at home, because of things that had happened, but also because they feared there were other dangerous situations. CG: Nothing serious has happened – yet. It isn't that he will burn the house down or anything, but suppose something serious did happen to him.'
8** The CG has a need for advice about safety outside.	2	CGs were concerned about PWDs not recognising the route and getting lost, and they worried about risky behaviour in traffic. CG: 'I definitely do not begrudge her [PWD] her cycling, but she is not aware of priority rules, for example.'

PWD: person with dementia; CG: caregiver; CM: case manager; NA: needs assessment

** very good recognizable in the majority of clients, and recommended by expert panel (25)

* fairly good recognizable in the majority of clients, and recommended by expert panel (25)

CHAPTER 7 -

General discussion
This chapter presents the outcomes of our research on dyadic, activating interventions for people with dementia and their informal caregivers, who live together in the community. Dyadic, activating interventions support the maintenance of meaningful activities for people with dementia and their caregivers that dementia may prevent them from participating in. The goal of this thesis was first, to study the impact of these interventions on people with dementia and their caregivers, and second, to explore the appropriateness of criteria for referring them to these interventions. In this final chapter, the main findings of both parts of the thesis are presented, along with a reflection on the findings, methodological considerations, suggestions for future research and implications for practice.

Main findings of the thesis

Part I Impact of dyadic, activating interventions for people with dementia and their informal caregivers

Our systematic review of the impact of dyadic, psychosocial interventions showed significant positive effects on the mood, behavior, daily activities and quality of life of people with dementia, as well as on the mood, sense of burden, competence and quality of life of caregivers (Chapter 2). We found particularly promising results when reviewing components of interventions closely related to the targeted functional domains. Components designed to increase the practice of meaningful activities did improve levels of activity and functional dependence. Components designed to increase the caregiver's supporting skills improved their sense of competence. And components designed to improve sleep at night had a positive impact on sleep. Despite these positive and obvious associations, the results across all studies were heterogeneous. Interventions proven effective in one trial were often not replicated in other studies studying similar interventions. The needs of people with dementia and their caregivers were not considered as part of the inclusion process by any of the studies. Scientific studies are frequently based on a 'one size fits all' approach. We therefore hypothesized, that one of the reasons for the heterogeneity of effects may be a poor match between intervention components and personal needs, characteristics and preferences of people with dementia and their caregivers.

In our next two studies, we investigated the potential of tailored support through dyadic, activating interventions (Chapters 3 and 4). We included three interven-

tions that can be considered as 'dyadic, activating interventions'. Dyadic, activating interventions aim to increase the skills of people with dementia and their informal caregivers in both maintaining daily activities and coping with a decline in general functioning due to the illness. The Pleasant Events Program focused on maintaining enjoyable activities, the Exercise and Support Intervention on maintaining physical fitness and enjoyable activities, and Occupational Therapy (Community Ocupational Therapy in Dementia – COTiD) on maintaining independence in meaningful activities.¹-⁴

We studied each intervention's fit with the needs, characteristics and preferences of participating dyads (Chapter 3). Interviews with participants and their coaches revealed five factors that described the most important needs, characteristics and preferences that lead to a better match with the intervention. Four factors describe actual needs, characteristics and preferences of dyads, whereas the fifth factor explores the meaning of the activities for a dyad:

- 1 Timing and openness to change: dyads who were aware of the impact dementia had on their daily life and who were open to making changes in their routine perceived the interventions as appropriate. Often, this occurred in an early stage of dementia.
- 2 Need for activity: dyads who experienced a lack of daily activities, but expressed a desire to maintain these activities perceived the interventions as appropriate.
- 3 Lifestyle: dyads with an active lifestyle and a preference for physical activity perceived the interventions as appropriate.
- 4 Doing activities apart-or-together: dyads having a need for joint activities and shared experiences felt supported by the interventions.
- 5 Meaning of the activity: before any appropriate matches can be made, it seemed highly relevant to explore the reason why the people with dementia and their caregivers participated in that particular activity. A need for activity can stem from the loss of daily pastimes, but also from experiencing pleasure or satisfaction or enjoying independence or social contacts while performing certain activities. For example, the ability to make coffee can mean a variety of things for different people: independence, daily routine, caring for or sharing with their partner, friends and family.

Next, we studied the working mechanisms of the interventions as perceived by participating dyads and professionals (Chapter 4). Three working mechanisms were

identified, each with several components. Interventions were perceived as positive, when the coach:

- 1 Enabled participation in activities for the person with dementia and the caregiver without providing false hope. Components:
 - Focusing on the dyad staying active.
 - Emphasizing their abilities, while not ignoring any limitations.
 - Attuning to their motivation.
 - Focusing on opportunities to learn new habits or adapt their routines.
- 2 Explored the most important personal activity needs of the person with dementia and the caregiver. Components:
 - Paying attention to the needs of both the person with dementia and the caregiver.
 - Being competent in the field of dementia.
 - Observing dyads practicing activities to explore capacities and limitations.
 - Setting attainable goals and choosing doable activities.
- 3 Utilized a solution-focused approach to adapt, test, and practice activities. Components:
 - Piloting ideas, tips and solutions.
 - Stimulating dyads' persistence and repetition of activities.
 - Creatively adapting activities.
 - Practicing activities together.
 - Using coaching communication.

The core theme was empowerment. The interventions seemed to work when they empowered people with dementia and their caregivers by increasing their self-confidence, competence and hope which enabled them to maintain activities.

Part II The Appropriateness of criteria for referral to dyadic, activating interventions

Part II of this thesis was directed at applying the findings of Part I to assist clinical professionals when referring people with dementia and their caregivers to dyadic, activating interventions taking into account their needs, characteristics and preferences. The five factors identified in chapter 3 were operationalized into draft-criteria. According to an expert panel of referring professionals, the presence or absence of just over half the criteria (18 of 31) could be recognized in the large majority of their clients (Chapter 5). In the panel discussion, experts repeatedly mentioned the complexity of conducting a comprehensive needs assessment,

especially when trying to identify people's latent needs. Some reported a lack of attention to a dyad's need for social contact or their preference for physical exercise. They also shared that they were used to limiting their assessments to services they knew were readily available.

The draft-criteria were regularly identified in real-life needs assessments conducted by dementia case managers, but they were usually not expressed in an explicit manner (Chapter 6). Most dyads expressed the need to maintain activities. However, the importance of and the meaning of the activity was not frequently discussed. The dyads clearly needed encouragement to share and explore those needs, but no further exploration took place. Similar to the previous study, we found that case managers often limited their exploration of needs to those catered for by standard care services. A service-oriented focus instead of a person-centered approach seemed to be common practice.

Reflection on the findings

Within person-centered care, it is important to address meaningful activities.⁵⁻⁹ Dyadic, activating interventions can support dyads to maintain activities. Neurological or physical impairment, as well as personal, social, and environmental contexts can influence activity and participation.¹⁰ The dyadic, activating interventions that we studied, are focused on enablement, taking impairment, activity and participation into account.¹¹ This enabling approach is oriented to the optimal use of remaining capacities and person-centered, a core recommendation in the Global Action Plan on the Public Health Response to Dementia of the World Health Organisation.¹² In this thesis we developed criteria that may help referring professionals in estimating the appropriateness of dyadic, activating interventions for a dyad. The criteria that are recognizable, as demonstrated in chapters 5 and 6, can be used as a conversation tool to both assess the activity needs of a dyad and evaluate the appropriateness of the interventions. Our research revealed two barriers for the use of these criteria: the suboptimal quality of the needs assessments (Chapters 3, 5, 6), and a limited awareness of activating interventions to support dyads (Chapters 4, 5). We will discuss these barriers in more detail below.

Quality of needs assessment

Our studies identified several limitations in current needs assessments (Chapter 3, 5, 6). The needs, characteristics and preferences of people with dementia and

informal caregivers are not systematically checked, and the assessment is often limited to medical and personal care domains. We also noticed a lack of in-depth exploration of needs and their meaning for that specific dyad. Previous research shows that dyads only mention needs when they know a service exists.¹³ They are hesitant to mention needs and preferences when they are unsure if there is a potential intervention.¹⁴ A comprehensive needs assessment can explore personal needs in more detail. Our research showed that questions about preferences and wishes encourage people with dementia and their caregivers to talk about their daily life and how they cope with limitations (Chapters 3, 4, 6). By doing so, latent needs might emerge, particularly those related to their daily functioning. In addition, exploring needs stimulated people with dementia and caregivers to come up with solutions themselves. It can be challenging to talk about practical topics in detail, because the stories of people with dementia and their caregivers are often focused on loss and grief.¹⁵ However, people with dementia and caregivers in our research mentioned that a professionals' acknowledgement of their needs and feelings was important to them, even without a direct solution (Chapter 4).

Finally, identifying the meaning of an activity for people with dementia and caregivers is important as part of a comprehensive needs assessment, as it helps professionals to both better understand the dyad's needs and find an appropriate intervention (Chapters 3, 4, 5, 6). Riding a bicycle can be meaningful as a mode of transportation, but also as a need for a healthy lifestyle, recreation, or social contact. An appropriate intervention should match the meaning of the activity for that individual. As a result, different interventions might be suggested to people with, superficially, a similar activity need.

Awareness and knowledge of activating interventions and their impact

Our research showed that referring professionals were neither sufficiently aware of nor informed about activating interventions for people with dementia and their caregivers. Hence, they did not inform people with dementia sufficiently about the possible positive outcomes of training and an enabling physical and social environment, which might compensate for declines in functioning. As a result, people with dementia often thought that nothing more could be done, further increasing their feeling of helplessness following a diagnosis that is often devastating in itself (Chapter 3).

People with dementia and their caregivers can feel more positive when they are aware of small steps and changes that can help them cope.¹⁶ For example, one

person with dementia in our study, mentioned that the intervention had encouraged him to stay active and to push himself to maintain his daily activities. A caregiver shared that she learned to motivate her partner in a gentler way, rather than using a phrase that frustrated him ('you have to'). Other studies also emphasize the importance of positive feelings as a coping mechanism for people with dementia.¹⁷⁻²⁰ When referring professionals are more aware of dyadic, activating interventions, they can discuss an enabling and capacity-building approach and its possible benefits with a dyad.

Methodological considerations

Evidence for impact of interventions

Our systematic review and other reviews showed evidence for the beneficial impact of dyadic, activating interventions.²¹⁻²⁴ These reviews also point to inconsistency in evidence for individual interventions. This is also the case for the three interventions central to this thesis. Some scientific evidence supported them when this study began, but the positive findings of the 'Exercise and Support Intervention' and 'COTiD' were not replicated in later studies.²⁵⁻³³ Looking back, this may be because the studied interventions did not specifically match the needs, characteristics and preferences of the participants involved.^{14,34-36} There is still a gap in knowledge, especially regarding which intervention works best for whom and when.^{22,37} This research addresses some of those issues.

Criteria and indicators

In Part II of this thesis, criteria for the assessment of activity needs were developed using the RAND Appropriateness Method (Chapter 5). The RAND methodology requires an extensive literature review on the effectiveness of the relevant intervention as a basis for developing indicators.³⁸ However, there was a limited amount of evidence available supporting our research. We still choose to use the RAND methodology as it offered a structured and objective procedure for evaluating the recognizability of the draft-criteria. We formulated the draft-criteria following the findings of our qualitative study (Chapter 3). The RAND expert panel then rated them for recognizability in their daily practice. RAND normally uses the term 'indicators,' but because the underlying evidence for our study was less rigorous, the term 'criteria' was more adequate.

Triangulation in data collection and analysis

Input from several stakeholders is a strength in research. We interviewed people with dementia, their caregivers and the coaches involved in the studies described in chapters 3, 4 and 6. For the studies described in chapters 5 and 6 we included referring professionals, like case managers, geriatricians and general practitioners. This enabled us to incorporate a variety of perspectives. The unit of analysis in our studies was a case, where each case consisted of a person with dementia, their informal caregivers and any supporting professionals.³⁹ In each case, the narratives of the person with dementia, the caregiver and the coach complemented each other, resulting in 'rich data'.³⁹ The inclusion of multiple sources of information contributed to increased validity, credibility and trustworthiness of our findings.^{40,41}

Roles of the primary investigator

In qualitative research, the researcher's expectations and assumptions may influence the validity of the findings.⁴¹ The professional experience of the primary investigator as an occupational therapist working with people with dementia and their caregivers played a significant role in this thesis. It contributed to both building a confidential relationship ('rapport') with respondents and acknowled-ging the perspectives of both dyads and coaches (role-taking).⁴² However, the primary investigator's experience as an occupational therapist in the field of dementia also poses a potential bias, as personal expectations and assumptions may have influenced the data interpretation process.⁴⁰ Reflexive objectivity can improve the validity of a study's outcomes.⁴³ In our studies, we strived for 'reflexive objectivity' through collaboration with researchers from several backgrounds, including nursing, psychology, medicine, and public health sciences.^{41,43}

Recommendations for research

In this thesis, we developed a set of criteria to match dyadic, activating interventions with the needs, characteristics and preferences of people with dementia and their caregivers. The criteria can be used as a tool for both assessing activity needs of a dyad and evaluating the appropriateness of dyadic, activating interventions. Research regarding the validity of the criteria as a referring tool in clinical practice is needed. Furthermore, future research could focus on different dyadic, activating interventions, and add diversity of participants for example different cultural backgrounds. With regard to the design of future studies on dyadic, activating interventions for people with dementia and their caregivers, we recommend to explore a dyad's needs and preferences before including them in the study. Furthermore, specific outcome measures should be included related to the target areas of the intervention in addition to generic outcomes. For example, if an intervention is focused on physical fitness, one of the outcome measures should be directly related to physical fitness. If an intervention is focused on an increase in activities or social contacts, then the outcomes should at least include the measurement of those domains. We also recommend the inclusion of a measure regarding empowerment, like 'competence' or 'self-confidence', since this may help explain any impact that could be found. In addition to the usual outcome measures, like mood, behavioral changes and sense of burden, these outcome measures can give more comprehensive insight into the impact of dyadic, activating interventions and their working mechanisms.⁴⁴⁻⁴⁶

Implications for practice

The implications for practice provide suggestions for referring professionals, providers of dyadic, activating interventions, and teachers who focus on dementia care.

Referring professionals

Quality of the needs-assessment

Referring professionals, like clinicians and case managers, may benefit from learning additional skills to perform comprehensive needs assessments. In the Netherlands, geriatricians, general practitioners and dementia case managers hold a central position for referral to care and support after the initial dementia diagnosis. A comprehensive assessment would include all domains of daily life, like cognitive and physical impairment, activity, the social and environmental context and participation. Client-centered interview techniques and sensitivity to unmet needs and preferences in coping with declines in capacities and functioning are important. Questions like, 'What activities do you prefer?', or, 'What activities do you like to do together', followed by questions about the meaning and limitations the dyad experiences may help identify activity needs, preferences and latent needs. Such a needs assessment may assist the dyad in coming up with their own solutions.

Awareness and knowledge of dyadic, activating interventions

Referring professionals should become more aware of dyadic, activating interventions' positive impact in that they can make a big difference to people with dementia and their caregivers. For example, they may better cope with daily problems, even if they only made minor changes at first. Next, they should actively search the available dyadic, activating interventions and strength-based approaches in their region.⁴⁷

Providers and professionals of dyadic, activating interventions

Availability of dyadic, activating interventions

Service providers in the field of dementia care should increase the amount of dyadic, activating interventions offered, preferably evidence-based and focusing on different domains like physical activity, pleasant activities, music, social contacts and ADL-activities. This will make it easier to refer people with dementia and their caregivers to interventions that fit their needs. Wider availability will also add increased public awareness of the remaining capacities and interests of people with dementia. Professionals who provide dyadic, activating interventions should actively inform referring professionals about the goals and impacts of their intervention.

Competencies of professionals guiding the interventions

Professionals guiding dyadic, activating interventions must be able to set realistic goals with a dyad. For every dyad, they should explore where adaptations are needed. For an enabling approach using strength-based interventions, a professional must focus on clients' abilities while respecting their limitations. A professional will also benefit from extensive knowledge of cognitive and behavioral processes that accompany dementia to recognize capacities and limitations.⁴⁸ A rehabilitative approach is central to the paramedical health disciplines. Along with the other disciplines, they can shape this enabling approach in dementia care services. Finally, a creative and open mind will help professionals brainstorm solutions tailored to the people with dementia and the caregivers.

Education on dementia care

The following recommendations concern both undergraduate study programs and professional education in dementia-care.

- Professionals must be trained in a dyadic approach to both people with dementia and informal caregivers, taking their personal needs, characteristics and preferences into account when considering psychosocial interventions.^{5,49}
- The relationship between meaningful activities and well-being requires more attention in professional education. It should be emphasized that meaningful activities may be small activities such as daily habits and self-care or bigger activities and hobbies.⁵⁰
- The working mechanisms for an enabling and activating approach can be part of training programs, such as a focus on capacities, motivation, attainable goals, practicing activities and strength-based strategies for people with dementia and their caregivers.^{16,49}
- Finally, for a person-centered approach, interprofessional collaboration is needed, a challenge for all professionals involved in dementia care. This requires a professional to be empathetic and have knowledge of the services offered by other care providers. To date, both referring professionals and those guiding the interventions are still too focused on how the client can benefit from what they themselves do, rather than focusing on what is best for that particular dyad. It is important that students from different study programs collaborate on projects that come from 'real-life' clinical practice. They should be familiar with other disciplines from the start of their education and learn how to collaborate and complement each other.

Conclusion

Based on the identified factors for determining a fit and the perceived working mechanisms, criteria have been developed to guide clinicians and case managers in assessing activity needs and evaluating the potential match of an activating intervention with a dyad's needs, characteristics and preferences. A more comprehensive assessment of activity needs and greater awareness of the potential of dyadic, activating interventions may help people with dementia and their caregivers. This will contribute to more attention for continuing activities and feelings of self-confidence, competence and hope.

References

- Gitlin LN, Winter L, Burke J, Chemett N, Dennis MP, Hauck WW. Tailored activities to manage neuropsychiatric behaviors in persons with dementia and reduce caregiver burden: A randomized pilot study. American Journal of Geriatric Psychiatry. 2008;16(3):229-239.
- Graff M, Melick Mv, Thijssen M, Verstraten P, Zajec J. Ergotherapie bij ouderen met dementie en hun mantelzorgers: Het EDOMAH-programma [Occupational Therapy for Older Persons with Dementia and their informal Caregivers: COTiD-programme]. Houten: Bohn Stafleu van Loghum, 2010.
- 3. Prick AE, de Lange J, Scherder E, Pot AM. Home-based exercise and support programme for people with dementia and their caregivers: study protocol of a randomised controlled trial. *BMC Public Health.* 2011;11:894.
- 4. Verkaik R, Francke A, Lahr M, Zuidervaart M. Huisbezoeken; Voorkom Somberheid bij Geheugenproblemen, handboek voor de hulpverlener (plezierige activiteitenmethode). [Home visits: Preventing depressive mood in memory problems, manual for the professional: Pleasant activities]. retrieved from https://www.nivel.nl/nl/publicatie/huisbezoeken-voorkom-somberheid-bij-geheugenproblemen-handboek-voor-hulpverlener-en. Utrecht: Nivel, 2005.
- Aminzadeh F, Molnar FJ, Dalziel WB, Ayotte D. A review of barriers and enablers to diagnosis and management of persons with dementia in primary care. *Canadian Geriatrics Journal* 2012;15(3):85-94.
- 6. Dröes RM, Chattat R, Diaz A, et al. Social health and dementia: a European consensus on the operationalization of the concept and directions for research and practice. *Aging & Mental Health*. 2017;21(1):4-17.
- Kitwood TM. Dementia Reconsidered: The Person Comes First Buckingham UK: Open University Press, 1997.
- 8. de Lange J. Omgaan met dementie [Dealing with dementia](PhD thesis) Utrecht: Trimbos-instituut [Netherlands Institute of Mental Health and Addiction], 2004.
- O'Sullivan G, Hocking C. Translating action research into practice: seeking occupational justice for people with dementia. OTJR: Occupation, Participation, and Health. 2013;33(3):168-176.
- 10. Townsend EA, Polatajko HJ. Enabling Occupation II: advancing an occupational therapy vision for health, well-being, & justice through occupation. Ottawa: Canadian Association of Occupational Therapists, 2007.
- 11. Clare L. An enablement approach to dementia care and prevention. In *Proceedings of the International Federation on Aging Global Summit*; Copenhagen, 2016.
- 12. WHO. Global Action Plan on the Public Health Response to Dementia of the World Health Organization 2017-2025. Geneva: World Health Organization, 2017

- 13. van der Roest HG, Meiland FJM, Maroccini R, Comijs HC, Jonker C, Dröes RM. Subjective needs of people with dementia: a review of the literature. *International Psychogeriatrics*. 2007;19:559-592.
- Donkers HW, Van der Veen DJ, Teerenstra S, Vernooij-Dassen MJ, Nijhuis-Vander Sanden MWG, Graff MJL. Evaluating the social fitness Programme for older people with cognitive problems and their caregivers: lessons learned from a failed trial. *BMC Geriatrics*. 2018;18(1):237.
- Hasselkus BR, Murray BJ. Everyday occupation, well-being, and identity: the experience of caregivers in families with dementia. *American Journal of Occupational Therapy*. 2007;61(1):9-20.
- 16. Yu DS, Li PW, Zhang F, Cheng S, Ng TK, Judge KS. The effects of a dyadic strength-based empowerment program on the health outcomes of people with mild cognitive impairment and their family caregivers: a randomized controlled trial. *Clinical Interventions in Aging*. 2019;14:1705-1717.
- 17. Edvardsson D, Fetherstonhaugh D, Nay R. Promoting a continuation of self and normality: person-centred care as described by people with dementia, their family members and aged care staff. *Journal of Clinical Nursing*. 2010;19(17-18):2611-2618.
- O'Sullivan G, Hocking C, Spence D. Dementia: The need for attitudinal change. Dementia: The International Journal of Social Research and Practice. 2014;13(4):483-497.
- Wolverson EL, Clarke C, Moniz-Cook ED. Living positively with dementia: a systematic review and synthesis of the qualitative literature. *Aging & Mental Health*. 2016;20(7):676-699.
- 20. Wolverson Radbourne EL, Clarke C, Moniz-Cook E. Remaining hopeful in early-stage dementia: a qualitative study. *Aging & Mental Health*. 2010;14(4):450-460.
- 21. Brodaty H, Arasaratnam C. Meta-analysis of nonpharmacological interventions for neuropsychiatric symptoms of dementia. *American Journal of Psychiatry*. 2012;169(9):946-953.
- 22. Coon DW, Thompson L, Steffen A, Sorocco K, Gallagher-Thompson D. Anger and depression management: psychoeducational skill training interventions for women caregivers of a relative with dementia. *Gerontologist*. 2003;43(5):678-689.
- Olazaran J, Reisberg B, Clare L, et al. Nonpharmacological therapies in Alzheimer's disease: a systematic review of efficacy. *Dementia and Geriatric Cognitive Disorders*. 2010;30(2):161-178.
- Smits CHM, de Lange J, Dröes R-M, Meiland F, Vernooij-Dassen M, Pot AM. Effects of combined intervention programmes for people with dementia living at home and their caregivers: A systematic review. *International Journal of Geriatric Psychiatry* 2007;22(12):1181-1193.

- 25. Döpp CM, Graff MJ, Teerenstra S, Olde Rikkert MG, Nijhuis-van der Sanden MW, Vernooij-Dassen MJ. Effectiveness of a training package for implementing a community-based occupational therapy program in dementia: a cluster randomized controlled trial. *Clinical Rehabilitation*. 2015;29(10):974-986.
- Graff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM. Community based occupational therapy for patients with dementia and their care givers: Randomised controlled trial. British Medical Journal. 2006;333(7580):1196-1199.
- 27. Graff MJL, Vernooij-Dassen MJM, Thijssen M, Dekker J, Hoefnagels WHL, Olde Rikkert MGM. Effects of community occupational therapy on quality of life, mood, and health status in dementia patients and their caregivers: A randomized controlled trial. *Journals of Gerontology* - Series A Biological Sciences and Medical Sciences. 2007;62(9):1002-1009.
- 28. Prick AE, de Lange J, Scherder E, Twisk J, Pot AM. The effects of a multicomponent dyadic intervention on the mood, behavior, and physical health of people with dementia: a randomized controlled trial. *Clinical Interventions in Aging*. 2016;11:383-395.
- 29. Prick AE, de Lange J, Scherder E, Twisk J, Pot AM. The Effects of a Multicomponent Dyadic Intervention With Physical Exercise on the Cognitive Functioning of People With Dementia: A Randomized Controlled Trial. *Journal of Aging and Physical Activity*. 2017;25(4):539-552.
- 30. Prick AE, de Lange J, Twisk J, Pot AM. The effects of a multi-component dyadic intervention on the psychological distress of family caregivers providing care to people with dementia: a randomized controlled trial. *International Psychogeriatrics*. 2015;27(12):2031-2044.
- 31. Teri L, Gibbons LE, McCurry SM, et al. Exercise plus behavioral management in patients with Alzheimer disease: a randomized controlled trial. *JAMA*. 2003;290(15):2015-2022.
- 32. Teri L, Logsdon RG, Uomoto J, McCurry SM. Behavioral treatment of depression in dementia patients: a controlled clinical trial. *Journals of Gerontology Series B Psychological Sciences and Social Sciences*. 1997;52(4):P159-166.
- 33. Voigt-Radloff S, Graff M, Leonhart R, et al. A multicentre RCT on community occupational therapy in Alzheimer's disease: 10 sessions are not better than one consultation. *British Medical Journal open*. 2011.
- 34. Döpp CM, Graff MJ, Olde Rikkert MG, Nijhuis van der Sanden MW, Vernooij-Dassen MJ. Determinants for the effectiveness of implementing an occupational therapy intervention in routine dementia care. *Implementation Science*. 2013;8:131.
- 35. Prick AE, de Lange J, van 't Leven N, Pot AM. Process evaluation of a multicomponent dyadic intervention study with exercise and support for people with dementia and their family caregivers. *Trials.* 2014;15(1):401.
- 36. Voigt-Radloff S, Graff M, Leonhart R, Hull M, Olde Rikkert M, Vernooij-Dassen M. Why did an effective Dutch complex psycho-social intervention for people with dementia not work in the German healthcare context? Lessons learnt from a process evaluation alongside a multicentre RCT. British Medical Journal open. 2011;1(1):e000094.

- Gitlin LN. Good news for dementia care: caregiver interventions reduce behavioral symptoms in people with dementia and family distress. *American Journal of Psychiatry*. 2012;169(9):894-897.
- Fitch K, Bernstein SJ, Aguilar MD, et al. The RAND/UCLA Appropriateness Method User's Manual. Santa Monica, CA, 2001.
- 39. Evers JC. Elaborating on Thick Analysis: About thoroughness and Creativity in Qualitative Analysis. Forum: Qualitative Social Research 2016;17(1).
- 40. Corbin J, Strauss A. Basics of Qualitative Research, techniques and procedures for developing Grounded Theory. 4th ed. Califormia: SAGE Publications, 2015.
- 41. Levitt HM, Bamberg M, Creswell JW, Frost DM, Josselson R, Suarez-Orozco C. Journal article reporting standards for qualitative primary, qualitative meta-analytic, and mixed methods research in psychology: The APA Publications and Communications Board task force report. *American Psychologist.* 2018;73(1):26-46.
- 42. Maso I, Smaling A. Kwalitatief onderzoek: praktijk en theorie [Qualitative research: practice and theory]. Amsterdam: Boom, 1998.
- 43. Kvale S, Brinkmann S. Interviews, Learniong the Craft of Qualitative research interviewing. 2th ed. Califormia: SAGE Publications, 2009.
- 44. Bontje P. Unfolding familiarity. re-occupying daily life among older persons with physical disabilities, in Japan. (PhD thesis) Stockholm: Department of Neurobiology, Care Sciences and Society, Karolinska Institutet, 2015.
- 45. Cutchin MP, Aldrich RM, Baillard AL, Coppola S. Action theories for occupational Science: The contributions of Dewey and Bourdieu. *Journal of Occupational Science*. 2008;15(3):157-165.
- 46. Kuiper C. The Eventmaker, The hybrid art of performing professionals, work-setting rehabilitation. Den Haag: Lemma, 2007.
- Bennett S, Laver K, Voigt-Radloff S, Letts L, Clemson L, Graff M, Wiseman J, Gitlin L. Occupational therapy for people with dementia and their family carers provided at home: a systematic review and meta-analysis. *British Medical Journal open*. 2019;9(11):e026308. doi: 10.1136/bmjopen-2018-026308.
- Zwijsen SA, van der Ploeg E, Hertogh CM. Understanding the world of dementia. How do people with dementia experience the world? *International Psychogeriatrics*. 2016;28(7):1067-1077.
- 49. Dröes RM, Schols J, Scheltens P. Meer kwaliteit van leven; integratieve, persoonsgerichte dementiezorg [More quality of life; integrative, person-centred dementia care]. Leusden: Diagnosis Uitgevers, 2015.
- 50. Giebel CM, Sutcliffe C. Initiating activities of daily living contributes to well-being in people with dementia and their carers. *International Journal of Geriatric Psychiatry*. 2018;33(1):e94-e102.





Chapter 1 describes the introduction of this thesis' topic. People with dementia have trouble maintaining their usual, daily life. A lack of daytime activities at home and outdoors and the need for company is frequently reported. Informal caregivers report difficulties supporting the person affected by dementia with his or her activities and engaging in shared activities. Maintaining their own activities is also difficult, as care work consumes much time and energy. Dyadic, activating interventions place emphasis on maintaining meaningful activities and using the remaining capacities of the affected person to support both people with dementia and their caregivers. Three of those dyadic, activating interventions are central in our studies: the 'Pleasant Events Program', the 'Exercise and Support Intervention' and 'Occupational Therapy'(according to the Community Occupational Therapy in Dementia-guideline – COTiD). Past scientific effect-studies for these interventions have shown positive results in mood, daily activities and the general health of people with dementia, as well as on the mood and competence of their caregivers, and the quality of life of both parties.

Despite the difficulties encountered by people with dementia and their caregivers in daily activities, dyadic, activating interventions are scarcely offered. In current dementia care, the needs assessment is often limited to medical and personal care domains, while social and occupational domains are not systematically checked. More knowledge of dyadic, activating interventions and how to match them with person-specific needs, characteristics and preferences is needed. Criteria for evaluating the appropriateness of dyadic, activating interventions might help referring professionals in the referral process. The overall aim of this thesis has been (1) to study the impact of dyadic, activating interventions on people with dementia and their caregivers, and (2) to develop and evaluate criteria for referring a dyad to these interventions.

Part I The impact of dyadic, activating interventions for people with dementia and their informal caregivers

Chapter 2 reports the evidence of dyadic, psychosocial interventions for people with dementia and their family caregivers and the relationship with treatment components. A systematic review included 23 randomized controlled trials of moderate to high quality. These trials concerned 20 different dyadic psychosocial programs for people with dementia and caregivers between January 2005 and January 2012. 19 of these programs found significant positive effects on mood,

behavior, daily activities and the quality of life of people with dementia, and on the mood, sense of burden, competence and quality of life of caregivers. We found promising results for interventions with treatment components that were closely related to the targeted functional domains. Components designed to increase the practice of meaningful activities did improve levels of activity and functional dependence. Components designed to increase the caregiver's supporting skills improved their sense of competence. And components designed to improve sleep at night had a positive impact on sleep. However, while some of these interventions were proven effective in one trial, studies on similar interventions in other settings found no effect. We hypothesized that more attention must be devoted to matching targeted functional domains and the treatment components of an intervention with the needs of the person with dementia and the informal caregiver.

Chapter 3 describes a qualitative study to explore factors that participants perceive to be important for the match of dyadic, activating interventions. To improve knowledge about which intervention best suits specific needs, characteristics and preferences, we conducted semi-structured interviews with people with dementia (27), their caregivers (34) and coaches (19) after their participation in one of the following interventions: the 'Pleasant Events Program', the 'Exercise and Support Intervention' or 'COTID'. Five factors were identified influencing the intervention's appropriateness with the needs, characteristics and preferences of a dyad.

- 1 Timing and openness to change: dyads who were aware of the impact of dementia on their daily life and who were open to making daily routine changes, perceived the interventions as appropriate. Often, they were in an early stage of dementia.
- 2 Need for activity: dyads who experienced a lack of daily activities, but expressed a desire to maintain these activities, felt supported by the interventions.
- 3 Lifestyle: dyads with an active lifestyle and a preference for physical activity perceived the interventions as appropriate.
- 4 Doing activities apart-or-together: dyads having a need for joint activities and shared experiences felt supported by the interventions.
- 5 Meaning of the activity: the interventions had to match the reason why the people with dementia and their caregivers participated in that activity. Their need for activity can stem from the loss of daily pastimes, pleasure or satisfaction with certain activities, independence, or social contacts. For example, the ability to make coffee can mean different things for different people:

independence, daily structure, caring for and sharing with their partner, friends and family.

Attention to these five factors when choosing an intervention can contribute to a more person-centered approach, allowing the intervention to better fit the personal needs, characteristics and preferences of people with dementia and their caregivers.

Chapter 4 comprises a qualitative study about the working mechanisms of the three dyadic, activating interventions, as perceived by people with dementia, their caregivers and coaches. More knowledge of the working mechanisms of activating interventions might help to increase their impact. We analyzed the semi-structured interviews with dyads and coaches to find out what the working mechanisms had been for them. Three working mechanisms were identified, each with several components. Interventions were perceived as positive, when the coach:

- 1 Enabled the person with dementia and the caregiver without providing false hope. Components:
 - Focusing on the dyad staying active.
 - Emphasizing their abilities, while not ignoring any limitations.
 - Attuning to their motivation.
 - Focusing on opportunities to learn new habits or adapt their routines.
- 2 Explored the most important personal activity needs of the person with dementia and the caregiver. Components:
 - Paying attention to the needs of both the person with dementia and the caregiver.
 - Being competent in the field of dementia as a coach.
 - Observing dyads practicing activities to explore capacities and limitations.
 - Setting attainable goals and choosing doable activities.
- 3 Utilized a solution-focused approach to adapt, test, and practice activities. Components:
 - Piloting ideas, tips and solutions.
 - Stimulating dyads' persistence and repetition of activities.
 - Showing creativity to adapt activities.
 - Training activities.
 - Using coaching communication

Empowerment emerged as a core theme. Participants appreciated the focus on finding remaining capacities and strengths to compensate for limitations. They perceived that an individualized, strength-based approach contributed to positive changes, like more attention for continuing activities and feelings of confidence, competence, and hope. Dyads were able to take control of their situation and find out what worked for them. Coaches needed to be skilled in assessing a dyad's needs, capacities and limitations for taking part in activities, in communicating about activities, and in teaching dyads a solution-focused approach.

Part II The appropriateness of criteria for referral to dyadic, activating interventions

The five factors described in chapter 3 may help referrers match an intervention with the needs, characteristics and preferences of dyads. To make this knowledge more accessible and useful in clinical practice, we operationalized these factors into 31 draft-criteria, based on the qualitative descriptions in the interviews. Each criterion described a need, characteristic or preference of the person with dementia or the caregiver.

Chapter 5 presents a study about the recognizability of these draft-criteria for referring professionals. A panel of 12 physicians and case managers, all experts in dementia care, rated how recognizable these criteria were in their clinical practice, using the 'RAND Appropriateness Method'. The panelists scored 18 criteria as recognizable in most of their clients. For criteria about a preference for physical and social activities, a consensus was lacking, because referrers were not accustomed to assessing those criteria. Panelists also reported difficulty with making the latent needs of a dyad explicit, and the habit to limit their assessment to services within their region. The 18 recognizable criteria offer guidance to referrers to assess activity needs and discuss the appropriateness of an activating intervention with a dyad.

Chapter 6 focuses on the usefulness of the criteria for needs assessment and referral to dyadic, activating interventions. As well as recognizability, we wanted to explore the practicality of the criteria for assessing the activity needs of a person with dementia and their caregiver in clinical practice. We performed secondary analyses of interviews for needs assessments with 20 dyads by a

dementia case manager. Most people with dementia and caregivers reported activity needs. We frequently identified activity-needs, characteristics and preferences that were implicit in the stories of dyads. Preferences related to lifestyle were always expressed explicitly, but preferences for the other factors were often implicit. These needs should have been explored more profoundly in the assessment, to investigate the appropriateness of activating interventions.

Chapter 7 contains the general discussion. The criteria developed in this thesis may help referrers to assess activity needs, characteristics and preferences of dyads, and estimate the appropriateness of a dyadic, activating intervention, in line with a person-centered approach. The criteria are recognizable for referrers in their clinical practice and are present in most real-life needs assessments. Our studies reveal two barriers to using the criteria for selecting the most appropriate dyadic, activating intervention: the quality of the needs assessment and limited knowledge of activating interventions. We notice a lack of in-depth exploration of personal needs, and, where activity needs are identified, the exploration of the meaning of those activities for a dyad is often insufficient.

Our studies demonstrate that a service-oriented focus is most prevalent in needs assessments rather than a person-centered approach. Furthermore, referring professionals have limited knowledge about and are not sufficiently aware of the impact of dyadic, activating interventions.

The criteria developed in this thesis can be used as a conversation tool for assessing activity needs of a dyad, and evaluating the appropriateness of dyadic, activating interventions. We propose the following recommendations for research and practice:

- Research regarding the validity of the criteria as a referring tool in clinical practice is needed.
- Future studies exploring dyadic, activating interventions, should explore a dyad's needs and preferences before including them in a study. Possible effects are contingent on whether they fulfil the needs and preferences of the participants. They should also include outcome measures directly related to the target areas of the activating intervention and empowerment.
- Referring professionals, like clinicians and case managers, should move from a service-oriented focus to a person-centered approach. A detailed assessment of a dyad's personal needs includes their activity needs and exploring what the various activities mean to them. Greater awareness is needed of the impact of

strength-based and enabling approaches, such as dyadic, activating interventions.

- Dementia care providers should include more dyadic, activating interventions in their services.
- Professionals need to be trained in a person-centered, dyadic approach.
 Education for referring professionals should both address needs assessment and convey the relation between daily activities for people with dementia and their caregivers and their well-being.
- Education in dementia care should teach the skills for implementing an activating approach for people with dementia and their caregivers, instead of taking over activities in caregiving.
- Teaching creative thinking techniques will help brainstorm tailored solutions for the person with dementia and the caregiver.
- Interprofessional education should support interprofessional collaboration, an important condition for implementing a person-centered approach.



Samenvatting - Summary in Dutch

Hoofdstuk 1 beschrijft de aanleiding voor dit proefschrift. Mensen met dementie hebben moeite om hun dagelijks leven en hun bezigheden voort te zetten zoals ze gewend waren. Zij hebben zowel binnenshuis als buitenshuis vaak weinig om handen en kunnen een gebrek aan sociale contacten ervaren. Tegelijkertijd ervaren mantelzorgers regelmatig problemen bij het ondersteunen van de persoon met dementie in activiteiten, en bij het ondernemen van gezamenlijke activiteiten. Ook vinden zij het vaak moeilijk om hun eigen activiteiten voort te zetten, omdat de zorg veel tijd en energie vraagt. Activerende interventies, gericht op hen als koppel, zijn van belang om zowel de persoon met dementie als de mantelzorger te ondersteunen bij het continueren van betekenisvolle activiteiten, en het benutten van de nog aanwezige capaciteiten van de persoon met dementie.

Drie van dergelijke op koppels gerichte, activerende interventies staan centraal in onze studies: de 'Plezierige Activiteiten Methode', de 'Thuisinterventie met Beweging en Ondersteuning' en 'Ergotherapie (Ergotherapie voor mensen met Dementie en hun Mantelzorgers aan Huis (Edomah)- programma)'. Effectstudies naar deze interventies lieten positieve resultaten zien voor mensen met dementie op hun stemming, algemene gezondheid en het uitvoeren van dagelijkse activiteiten, en voor mantelzorgers op hun stemming en gevoel van competentie, en voor beiden op de kwaliteit van leven.

Ondanks deze positieve effecten worden deze op koppels gerichte, activerende interventies nog maar beperkt aangeboden. Ook ontbreekt kennis over wie het beste is gebaat bij welke interventie. Vanuit een persoonsgerichte benadering is het belangrijk dat interventies aansluiten op persoonlijke behoeften, voorkeuren en kenmerken van de persoon met dementie en de mantelzorger. In de huidige dementiezorg worden vooral medische en zorgbehoeften geïnventariseerd. Het sociale domein, waar in een vroeg stadium van dementie veel behoeften aan het behouden van betekenisvolle activiteiten kunnen liggen, wordt niet systematisch in de behoefte inventarisatie meegenomen. Criteria om behoeften, voorkeuren en kenmerken gericht op betekenisvolle activiteiten te inventariseren ontbreken, terwijl professionals (artsen en casemanagers) juist kunnen helpen door te verwijzen naar geschikte activerende interventies voor mensen met dementie en hun mantelzorgers. Het doel van deze thesis was (1) de impact van activerende interventies voor mensen met dementie en hun mantelzorgers te bestuderen, en (2) criteria om naar op koppels gerichte, activerende interventies te verwijzen te ontwikkelen en te evalueren.

Deel I Activerende interventies voor mensen met dementie en mantelzorgers

Hoofdstuk 2 beschrijft de effectiviteit van activerende interventies voor mensen met dementie en mantelzorgers, en de relatie met de behandelcomponenten van de interventies. In een systematische review zijn 23 studies van gemiddelde tot hoge kwaliteit over 20 op koppels gerichte, psychosociale interventies voor mensen met dementie en mantelzorgers opgenomen. De studies zijn tussen januari 2005 en januari 2012 gepubliceerd. Voor 19 interventies werden significante, positieve effecten gevonden op een of meerdere van de volgende uitkomstmaten: stemming, gedrag, dagelijkse activiteiten en kwaliteit van leven voor mensen met dementie, evenals voor stemming, zorglast, competentie en kwaliteit van leven van mantelzorgers. Soms was een interventie effectief in één studie, maar kon een andere studie geen effect aantonen. Het wetenschappelijk bewijs voor de effectiviteit van verschillende op koppels gerichte, psychosociale interventies is daarom heterogeen. Uit de literatuurreview kwam ook naar voren dat interventies met behandelcomponenten die betekenisvolle activiteiten ondersteunen veelbelovend zijn voor effecten op het domein 'uitvoeren van activiteiten'. Dat gold ook voor interventies met behandelcomponenten waarbij de mantelzorger praktische vaardigheden oefende voor het omgaan met de persoon met dementie voor effecten op competenties van de mantelzorger.

Hoofdstuk 3 beschrijft een kwalitatieve studie naar factoren die volgens de deelnemers van belang waren voor de aansluiting van de drie op koppels gerichte, activerende interventies op hun individuele behoeften, kenmerken en voorkeuren. We hebben semigestructureerde interviews gehouden met mensen met dementie (27), hun mantelzorgers (34) en de begeleidende coaches (19) na deelname aan een van de interventies: de 'Plezierige Activiteiten Methode', de 'Thuisinterventie met Beweging en Ondersteuning' en 'Ergotherapie (Edomah)'. De analyse leidde tot vijf factoren, die de aansluiting van de interventies op behoeften, kenmerken en voorkeuren, beïnvloedden.

1 Tijdigheid en openstaan voor verandering. Mensen met dementie en mantelzorgers die zich bewust waren van de consequenties van de dementie in hun dagelijks leven, en ervoor openstonden om hun dagelijkse routines aan te passen, vonden de interventies geschikt. Vaak waren zij in een beginnend stadium van dementie. Mensen met dementie en mantelzorgers die hun routines niet konden of wilden aanpassen op dat moment, hadden de interventies niet als geschikt ervaren.

- 2 Behoefte aan activiteiten. Mensen met dementie en mantelzorgers die hun dagelijkse bezigheden wilden behouden, en daarbij problemen tegenkwamen, vonden de interventies positief en ondersteunend. De interventies sloten niet aan bij mensen met dementie en mantelzorgers voor wie andere onderwerpen dan activiteiten op dat moment het belangrijkst waren.
- 3 Leefstijl. Mensen met dementie en mantelzorgers met een actieve leefstijl en een voorkeur voor fysieke inspanning vonden deze interventies geschikt.
- 4 Het samen of juist apart doen van activiteiten. Mensen met dementie en mantelzorgers bij wie de interventies aansloten bij de behoefte aan gezamenlijke activiteiten en gedeelde ervaringen voelden zich ondersteund.
- 5 Betekenis van de activiteit. De interventies werden als geschikt ervaren als zij aansloten bij de betekenis van de activiteit voor de persoon met dementie en de mantelzorger. Die betekenis kon zijn: iets te doen hebben, plezier in bepaalde bezigheden hebben, ervaren van zelfredzaamheid of het onderhouden van sociale contacten. Koffiezetten kan bijvoorbeeld verschillende betekenissen hebben voor verschillende mensen: het behouden zelfstandigheid of dagelijkse structuur, het verzorgen van een kopje koffie voor de ander, of het ontvangen van visite. Naast betekenis werden bij deze factor ook fysieke beperkingen als belemmeringen genoemd.

Aandacht voor deze vijf factoren bij het kiezen van een interventie kan bijdragen aan een meer persoonsgerichte benadering, waarbij de interventie zo goed mogelijk aansluit op persoonlijke behoeften, kenmerken en voorkeuren van mensen met dementie en mantelzorgers.

Hoofdstuk 4 bevat een kwalitatieve studie naar de werkzame mechanismen van de drie op koppels gerichte, activerende interventies, volgens de deelnemers en de begeleiders. Meer inzicht in de werkzame mechanismen kan helpen om de impact van de interventies te vergroten. In de semigestructureerde interviews met mensen met dementie, mantelzorgers en de coaches vroegen wij hen wat volgens hen werkzame mechanismen van de interventies waren. De analyse hiervan leidde tot drie werkzame mechanismen, elk bestaand uit meerdere componenten.

De interventies werden als positief ervaren als de coach:

 de persoon met dementie en de mantelzorger ondersteunde en stimuleerde actief te blijven, zonder valse hoop te wekken.
 Componenten van dit mechanisme waren:

- Focussen op 'actief blijven';
- Benadrukken van de aanwezige capaciteiten, maar ook de beperkingen erkennen;
- Aansluiten bij hun motivatie;
- Focussen op mogelijkheden voor mensen met dementie en mantelzorgers om nieuwe gewoonten te leren of hun routines aan te passen.
- 2 de meest belangrijke individuele behoeften aan activiteiten van de persoon met dementie en mantelzorger onderzocht. Componenten van dit mechanisme waren:
 - Aandacht schenken aan de behoeften van zowel de persoon met dementie als de mantelzorger;
 - Competent zijn als coach op dementiegebied;
 - Het observeren van de koppels bij het uitvoeren van activiteiten en het in kaart brengen van capaciteiten en beperkingen;
 - Haalbare doelen stellen en het kiezen van haalbare activiteiten.
- 3 een oplossingsgerichte benadering toepaste om activiteiten aan te passen en te oefenen. Componenten van dit mechanisme waren:
 - Ideeën, tips en oplossingen uitproberen;
 - Het stimuleren van het doorzettingsvermogen van koppels en van het herhalen van activiteiten;
 - Creativiteit gebruiken om activiteiten aan te passen;
 - Activiteiten gezamenlijk oefenen;
 - Coachende communicatie gebruiken.

Empowerment was het centrale thema. De deelnemers waardeerden de focus op de nog aanwezige capaciteiten en hun sterke kanten om hun beperkingen te compenseren. Ze vonden dat de individueel aangepaste, op sterke kanten gebaseerde benadering had bijgedragen aan positieve veranderingen, zoals meer aandacht voor het blijven doen van activiteiten, meer zelfvertrouwen, een groter gevoel van competentie, en hoop. Mensen met dementie en mantelzorgers ervoeren meer grip op hun situatie, en leerden uit te zoeken wat voor hen werkte. De coaches moesten vaardig zijn in het inventariseren van behoeften, capaciteiten en beperkingen van mensen met dementie en mantelzorgers voor het uitvoeren van activiteiten, daar helder over kunnen communiceren, en in staat zijn om hen een oplossingsgerichte aanpak te leren.

Deel II Criteria om naar op koppels gerichte, activerende interventies te verwijzen

De vijf factoren, die in hoofdstuk 3 zijn beschreven, kunnen verwijzers helpen bij het kiezen van een interventie die aansluit op behoeften, kenmerken en voorkeuren van mensen met dementie en mantelzorgers. Om deze kennis meer toegankelijk te maken en bruikbaar in de klinische praktijk, hebben we deze factoren geoperationaliseerd in 31 concept-criteria, gebaseerd op de omschrijvingen in de interviews. Ieder criterium beschrijft een behoefte, een kenmerk of een voorkeur van de persoon met dementie of de mantelzorger.

Hoofdstuk 5 beschrijft een studie naar de herkenbaarheid van deze conceptcriteria voor verwijzende professionals. Een panel van 12 artsen en casemanagers, allemaal experts in de zorg voor mensen met dementie, scoorden de mate waarin deze criteria voor hen herkenbaar waren in hun klinische praktijk. We gebruikten de 'RAND Appropriateness Method'. De panelleden scoorden een groot deel van de criteria (18) als herkenbaar bij de meeste van hun cliënten. Consensus ontbrak voor de criteria over een voorkeur voor lichamelijke en sociale activiteiten, omdat de verwijzers niet gewend waren om deze voorkeuren te inventariseren. De panelleden bespraken ook de moeite die ze hadden om latente behoeften expliciet te maken, en hun gewoonte om de behoefte inventarisatie te beperken tot de beschikbare mogelijkheden in hun regio. De 18 herkenbare criteria bieden verwijzers richting om behoeften aan activiteiten van de persoon met dementie en de mantelzorger te inventariseren en de geschiktheid van een activerende interventie met beiden te bespreken.

Hoofdstuk 6 richt zich vervolgens op de praktische toepassing/bruikbaarheid van de criteria voor een behoefte inventarisatie en de verwijzing naar op koppels gerichte, activerende interventies. We voerden een secundaire analyse uit op interviews van casemanagers gericht op de inventarisatie van de behoeften van 20 cliëntparen. De meerderheid van de mensen met dementie en mantelzorgers noemde behoeften op het gebied van activiteiten. Voorkeuren met betrekking tot de factor Leefstijl waren altijd expliciet terug te vinden, maar de criteria behorend bij de andere factoren (Tijdigheid en openstaan voor verandering, Behoefte aan activiteiten, Activiteiten samen of juist apart en Betekenis van de activiteit) bleven vaak impliciet. Deze behoeften zouden uitgebreider geïnventariseerd moeten worden, om de geschiktheid van op koppels gerichte, activerende interventies te kunnen inschatten.

Hoofdstuk 7 bevat de algemene discussie. De ontwikkelde herkenbare criteria kunnen verwijzers helpen om behoeften, kenmerken en voorkeuren van mensen met dementie en van hun mantelzorgers met betrekking tot activiteiten te inventariseren, en vervolgens de geschiktheid van een activerende interventie in te schatten, passend bij een persoonsgerichte benadering. Onze studies tonen twee belemmeringen voor het gebruik van de criteria: de kwaliteit van de behoefte inventarisatie en de beperkte kennis over deze interventies bij verwijzers. We constateren dat behoefte inventarisaties niet erg diepgaand zijn. Als behoeften omtrent activiteiten werden aangegeven, werd de betekenis ervan voor de persoon met dementie en de mantelzorger vaak niet (genoeg) nagevraagd. Onze studies tonen aan dat behoefte inventarisaties eerder aanbodgericht dan persoonsgericht waren. Daarnaast hebben verwijzers beperkte kennis van op koppels gerichte, activerende interventies, en zijn ze onvoldoende op de hoogte van de impact ervan.

We hebben de volgende aanbevelingen voor verder onderzoek, en voor de praktijk:

- Voor het verder ontwikkelen van een verwijsinstrument voor de klinische praktijk, is verder onderzoek naar de validiteit van de criteria nodig.
- Toekomstige studies naar op koppels gerichte, activerende interventies zouden de behoeften en voorkeuren van mogelijke deelnemers en de aansluiting van specifieke interventies daarop moeten inventariseren, alvorens hen te includeren in de studie. Mogelijke effecten zijn immers afhankelijk van het al dan niet aansluiten bij die behoeften van de deelnemers. Ook zouden studies naar de effectiviteit van deze interventies uitkomstmaten moeten opnemen die empowerment en de domeinen waarop de interventie zich richt meten.
- De behoefte inventarisatie van verwijzende professionals, zoals artsen en casemanagers, zou moeten bewegen van aanbod-georiënteerd naar persoonsgericht. Een gedetailleerde inventarisatie van de persoonlijke behoeften van mensen met dementie en hun mantelzorgers moet ook behoeften aan activiteiten en de betekenis ervan bevatten. Verwijzers moeten zich meer bewust zijn van de impact van benaderingen die uitgaan van mogelijkheden van mensen met dementie, zoals deze activerende interventies.
- Aanbieders van zorg voor mensen met dementie zouden hun diensten moeten uitbreiden met meer op koppels gerichte, activerende interventies.
- Verwijzende professionals moeten worden getraind in een persoonsgerichte, op koppels gerichte benadering. Daarin moet aandacht zijn voor de kwaliteit van de behoefte inventarisatie en voor de relatie tussen (dagelijkse) activiteiten voor mensen met dementie en hun mantelzorgers en hun welbevinden.

- Vaardigheden voor een activerende benadering zouden in het onderwijs voor uitvoerende professionals meer aandacht moeten krijgen, om te snel overnemen van activiteiten te voorkomen.
- Technieken voor creatief denken in het onderwijs kunnen bijdragen aan het vinden van oplossingen-op-maat voor de persoon met dementie en mantelzorger.
- Ten slotte zou in het onderwijs aandacht moeten zijn voor interprofessionele samenwerking bij dementie, om verder te kunnen kijken dan het aanbod van de eigen discipline, een belangrijke voorwaarde voor een persoonsgerichte benadering.

Dankwoord – Acknowledgements in Dutch

Na mijn masterstudie wilde ik graag in onderzoek betrokken zijn. Het geeft verdieping, inspireert en biedt samenwerking met bevlogen collega's. Aan promoveren had ik aanvankelijk nooit gedacht, maar het bleek dé manier waarop ik onderzoek doen en werk kon combineren. Het resultaat is dit proefschrift.

Jacomine was onmisbaar bij dit onderzoek, van het allereerste plan tot de uitvoering ervan. Ik bedank je voor je niet-aflatende ondersteuning. Je bent onvoorstelbaar zorgvuldig en was altijd voorbereid in ons overleg. Ik kreeg mijn stukken altijd van feedback voorzien weer terug, ondanks jouw deadlines voor subsidieaanvragen, onderzoeksrapportages of colleges. Je legt de lat hoog en bent vernieuwend, inspirerend, en heel consistent in de kwalitatieve onderzoeksmethodologie. Op voorspraak van Jacomine sloot Anne Margriet bij dit promotietraject aan. Daarmee stelde je je open voor mij, afkomstig uit het praktijkgerichte hoger beroepsonderwijs, en ook voor de kwalitatieve onderzoeksmethodologie. Je plaatste altijd het stukje onderzoek waar wij mee bezig waren in het grotere geheel van wetenschappelijk onderzoek naar de zorg voor mensen met dementie en hun naasten. En op jouw voorstel sloot ook Eva halverwege aan als copromotor. Jij vervolgde dit avontuur met ons. Je voegde jouw (methodische insteek) insteek toe en betekende veel voor de leesbaarheid van de tekst in brede zin: de precisie in wat we hadden gedaan en gevonden, moest ook uit de tekst blijken. Ik bedank jullie alle drie voor al jullie inbreng en moeite, én voor de humor, die zo'n promotietraject zoveel doenlijker maakt. Ik heb genoten van de scherpzinnige feedback, vooral tijdens mondelinge overleggen.

Met dit proefschrift heb ik naast het doen van wetenschappelijk onderzoek ook mijn ervaringen als ergotherapeut kunnen verdiepen. Als jonge ergotherapeute vond ik in de jaren 80 werk in een psychogeriatrisch verpleeghuis. Dat was voor mij een nieuwe wereld na mijn ervaring in de volwassenrevalidatie. Ik vroeg me al snel af waarom taken van bewoners werden overgenomen, terwijl zij die met wat stimulans nog zelf konden uitvoeren. En waarom was de omgeving zo weinig ondersteunend voor de bewoners, en zo weinig huiselijk, terwijl zij er 'woonden'? In mijn contacten met bewoners merkte ik dat zij door meer zelf te doen, zichzelf ook meer konden uiten en misschien meer zichzelf konden zijn. In het verpleeghuis vond ik aansluiting bij een aantal collega's, en we gingen op zoek naar meer mogelijkheden om de bewoners te ondersteunen in hun dagelijkse activiteiten. Op de dagbehandeling leerde ik ook over mantelzorgers: hun loyaliteit, vragen en moeilijkheden.

Ouderen en mensen met dementie zijn altijd de rode draad in mijn werk gebleven,

als docent aan de Opleiding Ergotherapie, als onderzoeker bij het Kenniscentrum Zorginnovatie van de Hogeschool Rotterdam, en als ergotherapeut in een vrijgevestigde praktijk. De focus was op hoe hun dagelijkse activiteiten en gewoonten bij kunnen dragen aan het behouden van een betekenisvol leven. Het geeft dan ook veel voldoening dat ik juist op dit onderwerp onderzoek kon doen en kon promoveren. Ik ben Jacomine, Anne Margriet en Eva dankbaar dat zij dit onderwerp met mij deelden. Ik bedank de Hogeschool Rotterdam, en dan met name de Opleiding Ergotherapie en het Kenniscentrum Zorginnovatie, dat zij mij hierin wilden faciliteren.

Mensen met dementie, hun mantelzorgers, de coaches en ergotherapeuten en de deelnemers aan het panel van verwijzers hebben allen bijgedragen aan onze studies. Ze deelden open hun ervaringen en staken allen hun energie en tijd in de interviews en bijeenkomsten, vanuit de gedrevenheid om de begeleiding en zorg voor mensen met dementie en mantelzorgers beter te maken. Ik bedank Anna-Eva Prick en Connie Klingeman, die als projectleider van twee interventieprogramma's respondenten voor mij hebben geworven. Ook dank ik Anna-Eva Prick, Hanny Groenewoud en Pepijn Roelofs die als mede-auteurs bijdroegen aan de verschillende studies.

Het werk aan dit proefschrift gebeurde naast mijn baan bij de Opleiding Ergotherapie van de Hogeschool Rotterdam, ingebed in de onderzoekslijnen van het Kenniscentrum Zorginnovatie. De combinatie kostte wel eens moeite, maar de immer positieve houding op beide werkplekken heeft me zeker geholpen. Daarvoor bedank ik alle collega's. Mijn paranimfen Joan Verhoef en Hanny Groenewoud vertegenwoordigen beide plekken.

In mijn loopbaan hebben veel mensen mij een stukje geholpen om te komen waar ik nu ben. Van hen noem ik Veroon Snater, docent tijdens mijn opleiding ergotherapie, die mij in die praktijkgerichte opleiding op het spoor van theorie en onderzoek heeft gezet, en Chris Kuiper die mij op de Hogeschool Rotterdam het zetje gaf om de masterstudie te doen. Op mijn beurt hoop ik ook iets hiervan over te brengen aan onze studenten.

Ik bedank de verschillende collega's, vrienden en mijn gezin die teksten van feedback hebben voorzien. Dit promotietraject was zonder een begripvolle privéomgeving niet mogelijk geweest. Mijn familie en schoonfamilie bieden me een warme omgeving en vrienden zorgen voor relativering en afleiding. Ik ben gezegend met een sprankelend gezin, en ik hoop dat Bert en ik nog veel met elkaar, en met Emiel en Maria, en Amber kunnen delen.



Over de auteur - About the author in Dutch
Netta van 't Leven, geboren op 16 mei 1959, groeide op in Den Haag, en volgde de middelbare school aan het Christelijk Lyceum Delft. Na een tussenjaar, heeft ze van 1978-1982 Ergotherapie gestudeerd aan de Revalidatie Academie Hoensbroeck (nu Zuyd Hogeschool). Zij volgde van 1999-2001 de European Masters Study of Science in Occupational Therapy, een gezamenlijk initiatief van Karolinska University Stockholm, School of Occupational Therapy and Physiotherapy, Naestved, Hogeschool van Amsterdam, en University of Brighton.

Na de studie Ergotherapie ging ze als ergotherapeut tijdelijk aan het werk in het Nederlands Zeehospitium Kijkduin (nu Basalt). Daarna werkte zij zes jaar bij Psychogeriatrisch verpleeghuis Dorestad, onderdeel van Psychiatrisch Centrum Bloemendaal, in Den Haag (nu Parnassia), waar zij voor het eerst met mensen met dementie te maken kreeg. Later verbreedde zij haar werkervaring bij Stichting Verpleeghuis De Bieslandhof in Delft (nu Pieter van Foreest).

Vanaf 1994 werkt zij als docent Ergotherapie aan de Hogeschool Rotterdam. Na het afronden van haar masterstudie combineerde zij de functie van hogeschooldocent met die van onderzoeker bij projecten van het Kenniscentrum Zorginnovatie van Hogeschool Rotterdam, onder andere Academische Werkplaats Dementie en VitaDem, en bij de afdeling IQ healthcare van het Radboudumc. Van 2008 tot 2015 was zij praktijkhouder van een vrijgevestigde ergotherapiepraktijk 'Ergotherapie voor Ouderen', te Schiedam, waar zij ongeveer een dag per week als praktiserend ergotherapeut werkte.

In 2010 startte zij dit (part-time) promotietraject m.b.v. een promotievoucher van de Hogeschool Rotterdam. De studies voor deze thesis werden uitgevoerd bij het Kenniscentrum Zorginnovatie van Hogeschool Rotterdam, in samenwerking met de afdeling Klinische, neuro- en ontwikkelingspsychologie van de Vrije Universiteit, Amsterdam.



Publications and presentations

Publications other than included in this thesis

Peer reviewed

- PrickAJC, de Lange J, Van't Leven N, PotAM. Process evaluation of a multicomponent dyadic intervention study with exercise and support for people with dementia and their family caregivers. *Trials*. 2014;15:401.
- Van't Leven N, Graff MLJ, Kaijen M, de Swart BJM, Olde Rikkert MGM, Vernooij-Dassen MJM. Barriers to and facilitators for the use of an evidence-based occupational therapy guideline for older people with dementia and their carers. *International Journal Geriatric Psychiatry*, 2012;27(7):742-748.
- Van't Leven N, Graff MJL, Kaijen M, de Swart BJM, Olde-Rikkert MGM, Vernooij-Dassen MJM. Implementeren van het ergotherapie-programma voor ouderen met dementie en hun mantelzorgers aan huis; belemmerende en bevorderende factoren. Wetenschappelijk Tijdschrift voor Ergotherapie, 2012;2: 5-14.
- Van't Leven N, de Lange J. Quality time- in ondersteuning van mantelzorgers van thuiswonende ouderen met dementie: een kwalitatief onderzoek. *Wetenschappelijk tijdschrift voor Ergotherapie*, 2008;1,8-13.
- Van't Leven N, Jonsson H. Doing and being in the atmosphere of the doing. Environmental influences on occupational performance in a nursing home. *Scandinavian Journal of Occupational Therapy*, 2002;9:148-155.

Other

- Hoogland A, Van't Leven N, Groenewoud JH, de Lange J. Een goed leven met dementie?! Utrecht: Vilans, Kenniscentrum Zorginnovatie Hogeschool Rotterdam, 2018. https://www.vilans. nl/vilans/media/documents/publicaties/een-goed-leven-met-dementie.pdf
- Groenewoud JH, Van't Leven N, Senhaji H, Döpp C, de Lange J. *Handleiding Vitadem* Rotterdam: Kenniscentrum Zorginnovatie, Hogeschool Rotterdam, 2018. www.hr.nl/handleidingvitadem.
- Van't Leven N. Leren toepassen van strategieën door een oudere met dementie en de mantelzorger. Hoofdstuk 19 in: Ergovaardig, deel 2, Uitvoeren van interventies. Redactie: van Dijk K, Jakobs M, Laban-Sinke I, Nas A. Amsterdam: Boom uitgevers, 2017.
- Van't Leven N. Ergotherapeutische zorg aan ouderen, wat is daar bijzonder aan? Ergotherapie Magazine, 2011;2:1-4.
- Van't Leven N. Ouderen met dementie en hun mantelzorger strategieën aanleren. Hoofdstuk
 7 In: Ergovaardig deel 2 Uitvoeren van interventies. Redactie: Speth-Lemmens I, Stomph M, de
 Vries K. Den Haag: Lemma, 2009.

- Van't Leven N, van Herwaarde H. Plaats en rollen van de paramedicus in de ouderenzorg. Hoofdstuk 9 In: *Paramedische zorgverlening aan ouderen*. Redactie: van Herwaarde H, Van't Leven N. Den Haag: Lemma, 2008.
- Daniëls R, Van't Leven N, van Nes F. Preventieprogramma voor ouderen, Lifestyle Redesign Programma, Gezond actief ouder worden. *Nederlands Tijdschrift voor Ergotherapie*, 2006;6:252-255.
- Van't Leven N, Klaassen D, Kramer M. Behoefte aan het Lifestyle Redesign Programma, De mening van ouderen. *Nederlands Tijdschrift voor Ergotherapie*, 2006;6:257-258.
- Driessen M, Van't Leven N, Mulders R, Kuiper C. Stages vormgeven. Nederlands Tijdschrift voor Ergotherapie, 1996;3:110-116.
- Kamerling A, Van't Leven N, van Tilburg R. Ergotherapie in de Psychogeriatrie. *Nederlands Tijdschrift voor Ergotherapie*, 1992;5:181-186.
- Van't Leven N. "Zo zit dat", Ontwikkelen van een Standaard voor de ergotherapeutische advisering van zitvoorzieningen voor ouderen met dementie in het verpleeghuis. *Nederlands Tijdschrift voor Ergotherapie*, 1989:3:82-88.

Presentations

- Presentatie minisymposium: Activerende interventies voor mensen met dementie en hun mantelzorgers: waarom en wat zijn de mogelijkheden? Van't Leven N, Prick AE, Bielderman A, Graff M. Geriatriedagen Den Bosch, 2020.
- Poster: Activiteiten behouden voor mensen met dementia wie zijn gebaat bij activerende interventies? Van't Leven N, de Lange J, van der Ploeg E, PotAM. Geriatriedagen Den Bosch, 2020.
- Poster: Working mechanisms of dyadic, psychosocial, activating interventions for people with dementia and informal caregivers: a qualitative study. Van't Leven N, de Lange J, van der Ploeg E, Pot AM. Occupational Science in Europe Congress, Amsterdam, 2019.
- Workshop: De mantelzorger hoort er ook bij Wederkerigheid en gezamenlijkheid: systemische benadering van chronisch zieke oudere cliënt en mantelzorger. Van't Leven N, Gerrits. I. Jaarcongres Ergotherapie Nederland, Bunnik, 2018.
- Poster: The fit of activating interventions to individual needs of persons with dementia and caregivers. Van't Leven N, de Lange J, Prick AE, Pot AM. European Nursing Congress: Caring for Older People, Rotterdam, 2016.
- Workshop: Dyadic approach for persons living with dementia and their informal caregivers. Van't Leven N, Gerrits. I. COTEC (Council Occupational Therapy in European Countries), Galway, 2016.
- Workshop: De kleur van de mantelzorger 'ieder vogeltje zingt zoals het gebekt is': zorgstijlen en meersporig communiceren. Van't Leven N, Gerrits. I. Studiedag voor ergotherapeuten in de Ouderenzorg, Bunnik, 2015.

- Presentatie: Psychosocial interventions for community-dwelling people with dementia and their family caregivers: a systematic review. Van't Leven N, Prick AE, Groenewoud JH, Roelofs PDDM, de Lange J, Pot AM. 16th International Congress of the World federation of Occupational Therapists, Yokohama, 2014.
- Presentatie: Community living people with dementia and their informal caregivers: matching a program and needs? Van't Leven N, de Lange J, Prick AE, Pot AM. 16th International Congress of the World federation of Occupational Therapists, Yokohama, 2014.
- Gastcollege: Dyadic Interventions for Community living People with Dementia and their Family Caregivers. Tokyo Metropolitan University, Tokyo, Faculty of Health Sciences, Division of Occupational Therapy, 2014.
- Presentatie: Interventies voor thuiswonende ouderen met dementie en hun mantelzorger. Van't Leven N, Prick AE, Groenewoud JH, Roelofs PDDM, de Lange J, Pot AM. Symposium Ergotherapie in de ouderenzorg: 'n update!, 2012.
- Workshop: Plezierige Activiteiten voor mensen met dementie en hun mantelzorgers. Van't Leven N, Tetteroo E. Landelijk Congres Moderne dementiezorg, Nieuwegein, 2011.
- Presentatie: Quality-time: professional care for informal carers of community-living older persons with dementia. Van't Leven N, de Lange J, Fourth European Nursing Congress Older Persons: the Future of Care, Rotterdam, 2010.
- Presentatie: EBP als basis voor onderzoek in het HBO. Verhoef J, Van't Leven N. HGZO congres, 2008.
- Workshop: Onderzoek doen in het HBO, leuk of lastig? Verhoef J, Van't Leven N. HGZO congres, 2008.
- Workshop: Gezond, Actief Ouder Worden. Daniëls R, Van't Leven N, van Nes F. Jaarcongres Ergotherapie Nederland, 2007.
- Presentatie: Doing and being in the atmosphere of the doing: Environmental influences on occupational performance in a nursing home". Van't Leven N, Jonsson H. World Congress for Occupational Therapy, Stockholm, 2002.

This thesis focuses on the needs of people living with dementia and their informal caregivers for activating interventions. We studied the impact of dyadic, activating interventions for people with dementia and their caregivers. The interventions were aimed at maintaining meaningful activities and using the affected person's remaining capacities to support both people with dementia and their caregivers. Professionals such as physicians and case managers see a lot of people with dementia who may profit from these activating interventions. However, knowledge about matching dyadic, activating interventions to person-specific needs and preferences is needed. We therefore developed criteria for the assessment of activity needs that might help referring professionals estimate the appropriateness of dyadic, activating interventions. Referring professionals can use these criteria to improve the assessment of activity needs of people with dementia and caregivers.

Still, greater awareness of the impact of strength-based and enabling approaches, such as dyadic, activating interventions is needed.

Dit proefschrift gaat over de behoeften van mensen met dementie en hun mantelzorgers aan activerende interventies. We onderzochten de impact van deze activerende interventies voor mensen met dementie en hun mantelzorgers als koppel. De interventies benutten de nog aanwezige capaciteiten van de persoon met dementie om hen beiden ondersteuning te bieden bij het behouden van betekenisvolle activiteiten. Professionals zoals artsen en casemanagers zien veel mensen met dementie, die gebaat kunnen zijn bij de activerende interventies. Maar er is meer kennis nodig over de aansluiting van deze interventies op de individuele behoeften en voorkeuren van de persoon met dementie en mantelzorger. Daarom ontwikkelden wij criteria om behoeften en voorkeuren gericht op betekenisvolle activiteiten te inventariseren. We toetsten deze criteria op herkenbaarheid en bruikbaarheid in de klinische praktijk. Deze criteria kunnen professionals helpen bij het verwijzen naar passende activerende interventies voor mensen met dementie en hun mantelzorgers.

Daarnaast is meer aandacht nodig voor benaderingen die uitgaan van mogelijkheden van mensen met dementie, zoals deze activerende interventies.

